Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFETs

Gopi Krishna Saramekala



Department of Electronics and Communication Engineering National Institute of Technology Rourkela

Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFETs

Dissertation submitted to the National Institute of Technology Rourkela in partial fulfillment of the requirements of the degree of

Doctor of Philosophy in Electronics and Communication Engineering

by

Gopi Krishna Saramekala

(Roll Number: 512EC1015)

based on research carried out under the supervision of

Prof. Pramod Kumar Tiwari



Department of Electronics and Communication Engineering National Institute of Technology Rourkela



May 27, 2017

Centre of Examination

Roll Number: 512EC1015 Name: Gopi Krishna Saramekala Title of Dissertation: Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFETs

We the below signed, after checking the dissertation mentioned above and the official record books of the student, hereby state our approval of the dissertation submitted in partial fulfillment of the requirements for the degree of *Doctor of Philosophy* in *Electronics and Communication Engineering* at *National Institute of Technology Rourkela*. We are satisfied with the volume, quality, correctness, and originality of the work.

Pramod Kumar Tiwari Supervisor

Umesh Chandra Pati Member, DSC

Prasanna Kumar Sahu Member, DSC

Gopala Krishna Srungavarapu Member, DSC

Kamalakanta Mahapatra Chairperson, DSC External Examiner

Kamalakanta Mahapatra Head of the Department



Dr. Pramod Kumar Tiwari Assistant Professor

May 27, 2017

Supervisor's Certificate

This is to certify that the work presented in the dissertation entitled "Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFETs" submitted by "Gopi Krishna Saramekala", Roll Number: 512EC1015, is a record of original research carried out by him under my supervision and guidance in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Electronics and Communication Engineering. Neither this dissertation nor any part of it has been submitted earlier for any degree or diploma to any institute or university in India or abroad.

Pramod Kumar Tiwari Assistant Professor

Dedicated to My teachers, family, and friends

Gopi Krishna Saramekala

Declaration of Originality

I, Gopi Krishna Saramekala, Roll Number 512EC1015 hereby declare that this dissertation entitled "*Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFETs*" presents my original work carried out as a doctoral student of NIT Rourkela and, to the best of my knowledge, contains no material previously published or written by another person, nor any material presented by me for the award of any degree or diploma of NIT Rourkela or any other institution. Any contribution made to this research by others, with whom I have worked at NIT Rourkela or elsewhere, is explicitly acknowledged under the section "Reference". I have also submitted my original research records to the scrutiny committee for evaluation of my dissertation.

I am fully aware that in case of any non-compliance detected in future, the Senate of NIT Rourkela may withdraw the degree awarded to me on the basis of the present dissertation.

May 27, 2017 NIT Rourkela

Gopi Krishna Saramekala

Acknowledgment

First of all, I would like to express my sincere gratitude to my supervisor Prof. Pramod Kumar Tiwari for his guidance and support. His extreme energy, encouragement, and immense knowledge that resolved all my queries have always been a constant source of motivation throughout my Ph.D. study and research. His passion and enthusiasm for teaching, sharing his knowledge and motivating students have not only inspired me, but has also made every individual an ardent academic disciple who has been taught by him. Being student-responsive and an effective mentor, his guidance helped me throughout thick and thin periods of my research, for which I am honestly indebted and thankful throughout my career and lifetime.

Further, I would like to thank my department head Prof. Kamalakanta Mahapatra for his constant support and words of encouragement for which he is extremely popular among the students. I would also like to thank Prof. Umesh Chandra Pati who has motivated me, since my M. Tech days from the next door lab. I am lucky enough to have some outstanding teachers in my life viz., Prof. Sukadev Meher, Prof. Kishore Chandra Pati, Prof. Korra Sathya Babu, Prof. Santos Kumar Das, and Prof. Ayas Kanta Swain who had supported me in every tough situation and held me high with their confidence to rebuild me.

I sincerely thank my dissertation committee: Prof. Prasanna Kumar Sahu and Prof. Gopala Krishna Srungavarapu for their encouragement, insightful comments and interrogations which helped to prove myself.

A special thanks to my fellow lab mates Mr. Visweswara Rao, Mr. Abirmoya Santra, Mr. Shiv Bhushan, Mr. Santanu Sarangi, Mr. Santosh Kumar, Ms. Sri Kanya, Mr. Mukesh Kumar, Ms. Sradhanjali, Mr. Ravi Kumar, Ms. Kiranmayi Prasada, Ms. Shara Mathew, Mr. Anand Kumar and others for the exciting discussions, for the sleepless nights to meet deadlines, and for all the fun in the last four years. I add my heartfelt thanks to all my department colleagues.

Mr. M N Anand Babu, Mr. Kiran Kumar, Mr. Sankata, Ms. Achala, Mr. Guru, Mr. Sujeevan, Ms. Pani, Mr. Sukanth, Mr. Ratnam, Mr. Jaya Rao, Mr. Govind, Mr. VRK, Mr. Katkam & others are those special friends beyond well-wishers for whom, I take this opportunity to thank for reminding me of other equal important things in life to maintain a balanced persona which had kept my days at NIT Rourkela vibrant across academia and social life.

Last but not least, I would like to thank my family and especially my parents Mr. Prasad and Mrs. Padmavathi, and my brother Mr. Vinay for supporting me emotionally throughout my life.

Abstract

Non-conventional metal-oxide-semiconductor (MOS) devices have attracted researchers' attention for future ultra-large-scale-integration (ULSI) applications since the channel length of conventional MOS devices approached the physical limit. Among the nonconventional CMOS devices which are currently being pursued for the future ULSI, the fully-depleted (FD) SOI MOSFET is a serious contender as the SOI MOSFETs possess some unique features such as enhanced short-channel effects immunity, low substrate leakage current, and compatibility with the planar CMOS technology. However, due to the ultra-thin source and drain regions, FD SOI MOSFETs possess large series resistance which leads to the poor current drive capability of the device despite having excellent shortchannel characteristics. To overcome this large series resistance problem, the source/drain area may be increased by extending S/D either upward or downward. Hence, elevatedsource/drain (E-S/D) and recessed-source/drain (Re-S/D) are the two structures which can be used to minimize the series resistance problem. Due to the undesirable issues such as parasitic capacitance, current crowding effects, etc. with E-S/D structure, the Re-S/D structure is a better choice. The FD Re-S/D SOI MOSFET may be an attractive option for sub-45nm regime because of its low parasitic capacitances, reduced series resistance, high drive current, very high switching speed and compatibility with the planar CMOS technology. The present dissertation is to deal with the theoretical modeling and computerbased simulation of the FD SOI MOSFETs in general, and recessed source/drain (Re-S/D) ultra-thin-body (UTB) SOI MOSFETs in particular. The current drive capability of Re-S/D UTB SOI MOSFETs can be further improved by adopting the dual-metal-gate (DMG) structure in place of the conventional single-metal-gate-structure. However, it will be interesting to see how the presence of two metals as gate contact changes the subthreshold characteristics of the device. Hence, the effects of adopting DMG structure on the threshold voltage, subthreshold swing and leakage current of Re-S/D UTB SOI MOSFETs have been studied in this dissertation. Further, high-k dielectric materials are used in ultra-scaled MOS devices in order to cut down the quantum mechanical tunneling of carriers. However, a physically thick gate dielectric causes fringing field induced performance degradation. Therefore, the impact of high-k dielectric materials on subthreshold characteristics of ReS/D SOI MOSFETs needs to be investigated. In this dissertation, various subthreshold characteristics of the device with high-*k* gate dielectric and metal gate electrode have been investigated in detail. Moreover, considering the variability problem of threshold voltage in ultra-scaled devices, the presence of a back-gate bias voltage may be useful for ultimate tuning of the threshold voltage and other characteristics. Hence, the impact of back-gate bias on the important subthreshold characteristics such as threshold voltage, subthreshold swing and leakage currents of Re-S/D UTB SOI MOSFETs has been thoroughly analyzed in this dissertation. The validity of the analytical models are verified by comparing model results with the numerical simulation results obtained from ATLASTM, a device simulator from SILVACO Inc.

Keywords: recessed-source/drain (Re-S/D) SOI MOSFET, drain-induced barrier lowering (DIBL), hot-carrier effects (HCEs), dual-metal-gate (DMG), high-k gate dielectric, and back-gate

CONTENTS

Certificate of Examination	ii
Supervisor's Certificate	iii
Dedication	iv
Declaration of Originality	v
Acknowledgment	vi
Abstract	vii
List of Figures	xiii
List of Tables	xvii
List of Abbreviations and Symbols	xviii

Chapter 1

Introduction

1.1	Introduction	1
1.2	History and Perspective of Transistors	2
1.3	MOSFET Technology Scaling	4
1.4	CMOS Technology Boosters	5
	1.4.1 Strained-Silicon (s-Si) Technology	5
	1.4.2 Hybrid-Orientation-Technology (HOT)	7
	1.4.3 High- <i>k</i> Dielectric Materials in MOS Transistors	8
	1.4.4 Dual-Metal-Gate (DMG) MOSFETs	8
	1.4.5 Non-Conventional MOSFETs	9
	1.4.6 Back-Gated SOI MOSFETs	12
1.5	SOI MOSFET	12
	1.5.1 SOI Manufacturing Technologies	13
	1.5.2 Salient Features of SOI MOSFETs	15
	1.5.3 Challenges and Remedies	17
1.6	Dissertation Motivation	19
1.7	Scope of Dissertation	21

SOI MOSFETs: A General Review

2.1	Introduction	23
2.2	SOI MOSFETs	24
	2.2.1 Source and Drain Silicides	25
	2.2.2 Tungsten Clad	26
	2.2.3 Schottky Source and Drain	27
	2.2.4 Elevated-Source/ Drain(E-S/D)	27
	2.2.5 Recessed-Source/Drain (Re-S/D)	27
2.3	Earlier Works Related to SOI MOSFETs	28
2.4	Subthreshold Characteristic Models of SOI MOSFETs	36
	2.4.1 Threshold Voltage Models of SOI MOSFETs	37
	2.4.2 Subthreshold Current Models of SOI MOSFETs	39
	2.4.3 Subthreshold Swing Models of SOI MOSFETs	40
2.5	Summary and Conclusion	41

Chapter 3

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

3.1	Introduction	44
3.2	Modeling of Two-Dimensional (2D) Surface Potential	45
3.3	Threshold Voltage Formulation	60
	3.3.1 Front Channel Threshold Voltage Formulation	61
	3.3.2 Back Channel Threshold Voltage Formulation	63
	3.3.3 Threshold Voltage Modifications due to Quantum Mechanical Effects	
	(QMEs)	64
3.4	Subthreshold Current Formulation	65
3.5	Subthreshold Swing Formulation	67
3.6	Results and Discussion	69

	3.6.1 Surface Potential	69
	3.6.2 Threshold Voltage	71
	3.6.3 Subthreshold Current	77
	3.6.4 Subthreshold Swing	82
3.7	Summary and Conclusion	86

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFET with High-*k* Gate-Dielectric Material

4.1	Introduction	88
4.2	Modeling of Two-Dimensional (2D) Surface Potential	89
	4.2.1 Derivation of $V(x, \sigma_i)$	98
4.3	Threshold Voltage Formulation	100
4.4	Subthreshold Current Formulation	102
4.5	Subthreshold Swing Formulation	103
4.6	Results and Discussion	104
	4.6.1 Surface Potential	105
	4.6.2 Threshold Voltage	107
	4.6.3 Subthreshold Current	111
	4.6.4 Subthreshold Swing	114
4.7	Summary and Conclusion	119

Chapter 5

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFET with Back-Gate Control

5.1	Introduction	120
5.2	Modeling of Two-Dimensional (2D) Surface Potential	121
5.3	Threshold Voltage Formulation	132

5.5Subthreshold Swing Formulation15.6Results and Discussion15.6.1 Surface Potential15.6.2 Threshold Voltage1
5.6Results and Discussion1.5.6.1 Surface Potential1.5.6.2 Threshold Voltage1.
5.6.1 Surface Potential15.6.2 Threshold Voltage1
5.6.2 Threshold Voltage 1.
5.6.3 Subthreshold Current 14
5.6.4 Subthreshold Swing 14
5.7 Summary and Conclusion 1

Conclusion and Future Scopes of Work

6.1	Introduction	152
6.2	Chapter-wise Summary and Conclusion	152
6.3	Suggestions for further work	157
Ref	erences	158
List	t of Publications	168

List of Figures

Fig. 1.1: A brief timeline of the major milestones in the development of CMOS	5
integration [Taur <i>et al.</i> (1998)]	3
Fig. 1.2: Lattice structure of (a) unstrained Si and SiGe (b) strained Si on relaxed SiGe	5
Fig. 1.3: Schematic images of silicon (100) and (110) surfaces. (Shadowed areas are the	e
channel areas indicating that the channeling effect of implanted ions is larger or	1
the (110) surface) [Yang <i>et al.</i> (2006)]	7
Fig. 1.4: Cross-sectional view of a dual-metal-gate (DMG) FET [Long <i>et al.</i> (1999)])
Fig. 1.5: Schematic structure of (a) partially-depleted (PD) SOI MOSFET (b) fully-	-
depleted (FD) SOI MOSFET 10)
Fig. 1.6: Various SOI device: (a) single-gate SOI transistor, (b) double-gate planar SO	I
transistor, (c) double-gate non-planar FinFET, (d) tri-gate FET, (e) quadruple	-
gate (or gate-all-around) FET, (f) gate-all around (or surrounding gate) FET	ſ
(nanowire FET). [Kim (2010)] 11	1
Fig. 1.7: Cross-sectional view of a fully-depleted SOI MOSFETs with back-gate contro	1
12	2
Fig. 1.8: SOS wafer fabrication flow13	3
Fig. 1.9: SIMOX process 14	4
Fig. 1.10: Smart-cut process 14	4
Fig. 1.11: Junction capacitances	5
Fig. 1.12: Cross-sectional view of (a) bulk-CMOS (b) SOI CMOS	7

Fig. 1.13: Schematic cross-sections of UTB SOI MOSFETs with (a) re	cessed
source/drain (Re-S/D) structure and (b) elevated source/drain (E-S/D) structure	ructure
	19
Fig. 2.1: Silicide Formation	25
Fig. 2.2: Schematic view of an SOI MOSFET with silicide layer above the sour	ce and
drain regions	25
Fig. 2.3: Schematic cross-sectional view of the W-TFT. [Zan et al. (2003)]	26
Fig. 2.4: Various SOI structures: (a) conventional (b) gate-all-around (c) ground pla	ane.
	30
Fig. 2.5: Cross-sectional view of a MOS transistor with various capacitance compo	nents
	34
Fig. 3.1: Cross-sectional view of dual-metal-gate (DMG) Re-S/D UTB SOI MOSFI	ET 46
Fig. 3.2: The DMG Re-S/D UTB SOI MOSFET with dominant intrinsic capac	citance
components	47
Fig. 3.3: Surface potential along the channel length at front-gate and back-gate	70
Fig. 3.4: Back surface potential along the channel at different control to scree	n gate
length ratio	71
Fig. 3.5:Threshold voltage versus channel length for varying oxide thickness.	73
Fig. 3.6: Threshold voltage versus channel length for varying Si film thickness.	74
Fig. 3.7: Threshold voltage versus channel length for varying recessed oxide thic	ckness.
	75
Fig. 3.8: Threshold voltage versus channel length for varying control to scree	n gate
length ratio.	76
Fig. 3.9: Subthreshold current variation with gate-to-source voltage for different c	hannel
lengths	78
Fig. 3.10: Subthreshold current variation with gate-to-source voltage for different	control
to screen gate length ratios	79
Fig. 3.11: Subthreshold current variation with gate-to-source voltage for different	t oxide
thicknesses	80

xiv

Fig. 3.12: Subthreshold current variation with gate-to-source voltage for different Si film
thicknesses 81
Fig. 3.13: Subthreshold current variation with gate-to-source voltage for different
recessed source/drain thicknesses 82
Fig. 3.14: Subthreshold swing variation with device channel length for different control
to screen gate length ratios 83
Fig. 3.15: Subthreshold swing variation with device channel length for different oxide
thicknesses 84
Fig. 3.16: Subthreshold swing variation with device channel length for different Si film
thicknesses 85
Fig. 3.17: Subthreshold swing variation with device channel length for different recessed
source/drain thicknesses 86
Fig. 4.1: The schematic structure of the Re-S/D FD SOI MOSFET showing the internal
parasitic fringe capacitance 90
Fig. 4.2: Potential on source and drain due to uniformly charged plate99
Fig. 4.3: Surface potential along the channel length at front-channel and back-channel 106
Fig. 4.4: Back surface potential along the channel length for various gate-dielectric
constants (ε_k) 107
Fig. 4.5: Threshold voltage versus channel length for different gate oxide thicknesses 109
Fig. 4.6: Threshold voltage versus channel length for different recessed oxide thicknesses
110
Fig. 4.7: Threshold voltage versus gate-dielectric constant (ε_k) 111
Fig. 4.8: Subthreshold current versus gate to source voltage for different channel lengths.
112
Fig. 4.9: Subthreshold current versus gate to source voltage for different oxide
thicknesses. 113
Fig. 4. 10: Subthreshold current versus gate to source voltage for varying gate dielectric
constants. 114
Fig. 4.11: Subthreshold swing variation with device channel length for different Si film
thicknesses 115

Fig. 4.12: Subthreshold swing variation with device channel length for different oxide
thicknesses 116
Fig. 4.13: Subthreshold swing variation with device channel length for different recessed
source/drain thicknesses 117
Fig. 4.14: Subthreshold swing variation with device channel length for different gate-
dielectric constants. 118
Fig. 5.1: The schematic structure of the Re-S/D FD SOI MOSFET with back-gate control
123
Fig. 5.2: Surface potential along the channel length at front-channel and back-channel 138
Fig. 5.3: Threshold voltage along the channel length for various back gate bias voltages
140
Fig. 5.4: Front-channel threshold voltage along the channel length for various silicon
channel thickness 141
Fig. 5.5: Front-channel threshold voltage versus gate length for varying recessed oxide
thickness 142
Fig. 5.6: Front and back threshold voltage versus back gate potential143
Fig. 5.7: Subthreshold current versus gate to source voltage for different buried oxide
thickness. 144
Fig. 5.8: Subthreshold current versus gate to source voltage for different channel lengths
145
Fig. 5.9: Subthreshold current versus gate to source voltage for different recessed
source/drain thicknesses 146
Fig. 5.10: Subthreshold current versus gate to source voltage for different back-gate
voltage 147
Fig. 5.11: Subthreshold swing variation with channel length for different silicon
thicknesses. 148
Fig. 5. 12: Subthreshold swing variation with channel length for different buried oxide
thickness 149
Fig. 5.13: Subthreshold swing variation with channel length for different back-gate
voltage. 150

xvi

List of Tables

Table 1.1: Scaling of device dimensions and circuit parameters [Taur (1998)]	5
Table 2.1: Brief literature work related to SOI MOSFET	38
Table 3.1: Device parameter values used for modeling and simulation of a DMG Re-S/D	UTB
SOI MOSFET	48
Table 4.1: Device parameter values used for modeling and simulation of a Re-S/D FD	SOI
MOSFET with high-k gate-dielectric material	91
Table 5.1: Device parameter values used for modeling and simulation of a Re-S/D FD	SOI
MOSFET with back-gate control	123

List of Abbreviations

WSTS	World semiconductor trade statistics
Si	Silicon
s-Si	Strained silicon
Ge	Germanium
Ga	Gallium
JFET	Junction field effect transistor
MOSFET	Metal-oxide-semiconductor field effect transistor
BJT	Bipolar junction transistor
TFT	Thin-film transistor
nFET	n-channel field effect transistor
pFET	p-channel field effect transistor
Re-S/D	Recessed- source/drain
E-S/D	Elevated- source/drain
IC	Integrated circuit
CMOS	Complementary metal-oxide-semiconductor
RAM	Random-access memory
SOI	Silicon-on-insulator
s-SOI	Strained Silicon-on-insulator
SOIAS	Silicon-on-insulator with active substrate
BOX	Buried oxide
DMG	Dual-metal-gate
ASD	Asymmetric source/drain
SSD	Symmetric source/drain
DSSB	Dopant segregated schottky barrier
TRIMAGS	Tri-material-gate stacks
MuGFETs	Multi-gate field effect transistors
ITRS	International technology roadmap for semiconductors
VLSI	Very large scale integration
ULSI	Ultra-large scale integration

DIBL	Drain-induced barrier-lowering
GIDL	Gate-induced drain leakage
RILC	Radiation-induced leakage current
SISP	Substrate induced surface potential
LOCOS	Local oxidation of silicon
FIPOS	Full isolation by porous oxidized silicon
SWCVD	Selective tungsten chemical vapor deposition
SIMOX	Separation by implantation of oxygen
SOS	Silicon-on-sapphire
HCEs	Hot-carrier effects
НОТ	Hybrid orientation technology
SCEs	Short-channel effects
QMEs	Quantum mechanical effects
UTB	Ultra-thin body
FD	Fully-depleted
PD	Partially-depleted
DD	Drift-diffusion
IR	Internet resource
RF	Radio frequency
FOMs	Figures-of-merits

List of Symbols

Permittivity of Si (F cm ⁻¹)
Permittivity of SiO ₂ (F cm ⁻¹)
Permittivity of High-k dielectric ($F cm^{-1}$)
Silicon channel thickness (nm)
Oxide thickness (nm)
Buried oxide thickness (nm)
Recessed source/drain thickness (nm)
Spacer thickness (nm)
Lengths of source/drain overlap (nm)
Ratio of control gate length to screen gate length
Channel length (nm)
Depletion layer width (nm)
Potential in channel region (V)
Charge of electron $(1.6e^{-19} \text{ C})$
Acceptor doping concentration of channel (cm ⁻³)
Source/drain doping concentration (cm ⁻³)
Substrate doping concentration (cm ⁻³)
Flat-band voltage (V)
Gate material work function (eV)
Electron affinity of Si (eV)
Silicon energy band gap (eV)
Thermal voltage (V)
Intrinsic carrier density (cm ⁻³)
Front channel surface potential (V)

$\phi_b(x)$	Back channel surface potential (V)
V_{GS}	Gate to source voltage (V)
V _{DS}	Drain to source voltage (V)
V_{BG}	Back-gate voltage (V)
V_{bi}	Built in voltage (V)
λ	Characteristic length (nm)
k	Boltzmann's constant (1.38e ⁻²³ JK ⁻¹)
Т	Temperature (300K)
C _{bot}	Bottom fringing capacitance (fF)
C _{ox}	Gate oxide capacitance (fF)
C _{rsd}	Recessed source/drain capacitance (fF)
C _{box}	Buried-oxide capacitance (fF)
C_{Si}	Channel capacitance (fF)
V _{th}	Threshold voltage (V)
ϕ_{Smin}	Minimum surface potential (V)
ϕ_{vc}	Virtual cathode potential (V)
Ε	Electric field (V cm ⁻¹)
I_s	Subthreshold current (A/um)
S	Subthreshold swing (mV/Dec.)
μ	Mobility of the free carrier $(cm^2V^{-1}sec^{-1})$
W	Width of the device along Z-axis (nm)
J	Charge density (C/cm ³)

Introduction

1.1 Introduction

The semiconductor industry has grown enormously since 1960 and today it is one of the world's largest industries. Such a phenomenal growth was never seen before in any sector in the history of mankind. World Semiconductor Trade Statistics (WSTS) forecast its 2016 global semiconductor sales growth to \$355.3 billion, which is 3.1% higher from 2015 sales. Moreover, the growth of the semiconductor industry has not been monotonous but has a great impact on other industries like communication, transportation, healthcare, security and surveillance, smart buildings and homes, and space etc. A semiconductor material is one whose electrical conductivity lies in between those of insulators and good conductors which are classified as intrinsic (pure) and extrinsic (impure) semiconductors. Even though, some pure elements and several compounds exhibit semiconductor properties, silicon (Si), germanium (Ge), and compounds of gallium (Ga) are the most commonly used in electronic devices. Further, depending upon the used impurity, extrinsic semiconductors are sub-divided into two classes which are *n*-type semiconductors and *p*-type semiconductors. Application of the semiconductors has been expanding day by day widely from radio in the 1960s to almost every electronic device that has an on-off switch in 2016. Semiconductor devices have been rigorously pursued targeting performance improvement in terms of higher speed, lower power consumption, higher efficiency with much functionality. The continuous need of performance improvement has been the motivating force behind the invention of new semiconductor devices and makes this field really fascinating as well as challenging.

1.2 History and Perspective of Transistors

A transistor (**tran**sfer + re**sistor**) is a semiconductor device which can amplify or switch electrical signals. It was invented in the mid-1940s by a team of scientists namely William Shockley, John Bardeen, and Walter Brattain [Brinkman *et al.* (1997)]. This solid state option of the vacuum tubes revolutionized the field of electronics and led to smaller and cheaper radios, calculators, and computers, among other things. The idea of 'field-effect transistor (FET)' was presented and patented by Lilienfeld in 1926 and 1930, respectively [Lilienfeld (1930)]. Shockley proposed a 'junction field-effect transistor (JFET)' in 1952 [Shockley (1952)] based on the unipolar concept with three terminals. In 1960, Kahng and Atalla [Kahng *et al.* (1960)] presented one of the most essential unipolar transistors called the metal-oxide-semiconductor field-effect transistor (MOSFET) with four terminals including a substrate for controlling the transport property more effectively. Even after the MOSFETs' invention, the BJT remained the transistor of choice for several analog circuits due to its superior electrical properties and ease of manufacturing.

In 1952, a technique to integrate a variety of electronic components onto a monolithic semiconductor crystal was invented [Internet Resource (IR1)]. In 1957, Jack Kilby [Kilby (1964)] developed an integrated circuit (IC) comprises of transistors, resistors, and capacitors, and received the Noble Prize in physics in the year 2000. Further, in 1959, a monolithic chip made of silicon material was presented by Robert Noyce, where metal vaporization technique was used to create interconnects. The major breakthrough in the improvement of the integration level onto the chip was marked in 1963, when the complementary-metal-oxide-semiconductor (CMOS) technology was invented [Wanlass *et al.* (1963)]. A CMOS circuit is designed with both *n*-channel and *p*-channel MOSFETs on a single substrate [Wanlass *et al.* (1963)]. Using the CMOS technology, researchers have been able to integrate hundreds of millions of transistors on a single chip which can even operate at room temperature [Taur *et al.* (1998)]. Today, the CMOS technology is the leading semiconductor technology and has been used for the fabrication of microcontrollers, microprocessors, static RAM, and other digital logic circuits because of

high packing density. Fig. 1.1 gives a sketch of the major milestone events in the development of CMOS technology.



Fig. 1.1: A brief timeline of the major milestones in the development of CMOS integration [Taur *et al.* (1998)]

Moore's law, proposed by Intel's co-founder Gordon Moore in 1965, states that the number of transistors per square inch on an integrated circuit will get double in every 18 months [Moore (1965), Thompson (2006)]. Moore's law has been an important guideline in developing the technology for the semiconductor industry since 1965. Semiconductor industry witnessed two important milestones in the year 1989 and 2005. The million and billion transistors were fabricated onto a chip for the first time in the year 1989 and 2005, respectively [Clarke (2005)]. Thus, ultra-large-scale integration (ULSI) was possible in

the year 2005 because of the MOSFET scalability and advanced manufacturing technology.

1.3 MOSFET Technology Scaling

Scaling of CMOS transistors below 32nm technology node causes severe short-channel effects (SCEs) such as subthreshold swing roll-up and threshold voltage roll-off [Wong (2002)]. It also increases the standby power dissipation worsening the switching characteristics and offers lower Ion/Ioff ratio. Hence, the suppression of the SCEs up to the tolerable level is of significant importance of the MOSFET scaling to continue the IC technology scaling trend.

The first scaling theory proposed by Dennard *et al.* [Dennard *et al.* (1974)] is known as 'constant-field scaling' theory. The principle of this theory is to scale the supply voltage and device dimensions (both horizontal and vertical) by the same scaling factor S > 1, in order to keep electric field unchanged. As a result, the circuit speeds up by the same factor S, and reduces the power dissipation per circuit by a factor of S^2 . The other important scaling schemes are constant-voltage scaling and generalized scaling. The scaling rules for various device parameters and circuit performance factors of constant-field, constant-voltage, and generalized scaling are shown in Table 1.1

Several attempts have been made to find out the alternative ways to follow Moore's law [Ieong *et al.* (2006), Thompson *et al.* (2006), Colinge (2008)]. As a result of these efforts, several kinds of techniques and alternative materials have been proposed for CMOS performance improvement [Chang *et al.* (2003), Ieong *et al.* (2006)]. Apart from introducing new materials for the manufacturing of transistors to improve the functionality per unit area, new transistor structures are also finding significance in the microelectronics industry to keep the scaling alive [Chang *et al.* (2003), Ieong *et al.* (2003), Ieong *et al.* (2006)]. Further, new materials, structures, and technologies that have been adopted by the semiconductor industry to combat the increasing leakage power and other scaling issues are discussed in section 1.4.

	Multiplication Factor, $S > 1$ Scaling parameter, α				
		Constant -field Rules	Constant -voltage Rules	General	ized Rules
	Device dimensions (t_{ox}, L, W)	1/S	1/S	1 / S	
Scaling assumptions	Doping concentration (N_a , N_d)	S	S^2	αS	
	Voltage (V)	1 / S	1	α / S	
	Electric field (<i>E</i>)	1	S	α	
Derived scaling behavior of device	Depletion-layer width (W_d)	1 / S	1 / S	1 / S	
	Capacitance ($C = \varepsilon A / t$)	1 / S	1 / S	1/S	
	Inversion/ layer charge density (O_1)	1	S	1	
parameters				Long Channel	Velocity Saturation
	Carrier velocity (<i>v</i>)	1	S	α	1
	Drift current (I)	1 / S	S	α^2/S	α / S
	Channel resistance (R_{ch})	1	1 / S	α^2 / S	α / S
Derived	Circuit delay time ($\tau \approx CV/I$)	1 / S	$1/S^2$	$1/\alpha S$	1 / S
scaling behavior of circuit	Power dissipation per circuit $(\tau \approx CV / I)$	$1/S^{2}$	S	α^3/S^2	α^2/S^2
parameters	Power density (P/A)	1	<i>S</i> ³	α^3	α^2
	Power-delay product per circuit ($\tau \approx CV/I$)	$1/S^3$	S	α^2/S^3	

Table 1.1: Scaling of device dimensions and circuit parameters [Taur (1998)]

1.4 CMOS Technology Boosters

As discussed in the previous section, the main challenge faced by the CMOS scaling is the presence of excess short-channel effects, when the device channel length enters in sub-100nm regime. CMOS technology boosters such as strained channel, hybridorientation-technology (HOT), high-*k* dielectric, multi-gate, back-gate and non-conventional MOS structures help to continue CMOS scaling in sub-100nm regime. Few CMOS technology boosters are discussed below.

1.4.1 Strained-Silicon (s-Si) Technology

Strained-Si technology has been adopted by the microelectronics industry as a performance booster for mainstream ULSI since the introduction of 32nm technology

node [Takagi (2007)]. Basically, strained-Si technology modifies the lattice constant of the material by a lattice deformation, which changes its energy band structure to trap carriers through the well formation and enhances mobility.

One of the techniques to generate the strain in the silicon channel is to grow a thin silicon epitaxial layer on the silicon-germanium $(Si_{1-x}Ge_x)$ substrate [Barraud *et al.* (2005)], where, *x* is the mole fraction of germanium in the silicon-germanium compound. The mole fraction *x* can be varied in order to introduce different amount of strain in the silicon film. In strained Si film, the 6-fold degenerate valley in the conduction band splits into a 2-fold non-planar and 4-fold planar degenerate valleys [Chaudhry *et al.* (2010)]. In the same way, the valence band also splits into two bands consisting of light and heavy holes respectively [Chaudhry *et al.* (2010)]. The carriers then prefer the lower energy valley, while occupying them, resulting in the reduction of inter-valley scattering and effective mass of the carrier [Chaudhry *et al.* (2010)]. By increasing the Ge concentration in the relaxed Si_{1-x}Ge_x substrate, the amount of biaxial strain increases and, therefore, a higher magnitude of the mobility enhancement can be achieved. The lattice structure strained-Si which grows on SiGe is shown in Fig. 1.2



Fig. 1.2: Lattice structure of (a) unstrained Si and SiGe (b) strained Si on relaxed SiGe

1.4.2 Hybrid-Orientation-Technology (HOT)

Hybrid-Orientation-Technology (HOT) improves carriers' mobility in Si CMOS devices [Yang *et al.* (2003)]. Properties that are exploited for mobility enhancement are surface orientation and tensile strain. To take full advantage of the carrier mobility dependence on surface orientation, nFETs should be fabricated on the (100) silicon surface and pFETs on the (110) silicon surface [Yang *et al.* (2006)]. The *n*-channel FET is formed in (100) orientation layer due to low oxide-interface charge density and highest electron mobility. The *p* channel FET is formed in (110) orientation layer in which the atom density is higher compared to (100) orientation layer, as can be observed from the Fig. 1.3. Mixing these two orientations on a single substrate is done by **Hybrid-Orientation-Technology** (HOT). It provides a way of reducing the current drive imbalance between *n*-type and *p*-type channels [Sheraw *et al.* (2005)]. The technological processes that fabricate these hybrid surface layers are membrane transfer and overgrowth. HOT has received considerable attention by researchers since its introduction due to its fabrication processes compatibility with the current VLSI technology [Yang *et al.* (2003), Sheraw *et al.* (2005)].



Fig. 1.3: Schematic images of silicon (100) and (110) surfaces. (Shadowed areas are the channel areas indicating that the channeling effect of implanted ions is larger on the (110) surface) [Yang *et al.* (2006)]

1.4.3 High-k Dielectric Materials in MOS Transistors

When the device channel length is scaled down below 32nm, an ultra-thin gate oxide, i.e., less than 1nm~five atomic layers becomes a necessity [Andresa *et al.* (2007)]. However, this very thin layer of gate oxide tends to leak a lot of current which leads to excess power consumption and a build-up of heat. Hence, to cut down the gate-leakage current, a physically thicker gate dielectric without compromising the gate-channel electrostatics coupling is required. In other words, the dielectric material needs to be physically thick but electrically thin [Internet Resource (IR2)]. The physical thickness of a high-*k* dielectric gate oxide is given by $t_k = t_{ox} \left(\frac{\varepsilon_{SiO_2}}{\varepsilon_k} \right)$, where, ε_{SiO_2} is the permittivity of silicon dioxide, ε_k is the permittivity of high-*k* dielectric material and t_{ox} is the effective gate oxide thickness of the device. Researchers have found some suitable high-*k* dielectric materials to control both short-channel effects and gate-leakage current simultaneously. Materials like HfO₂ (relative permittivity, k = 22), ZrO₂ (k = 25) and Ta₂O₅ (k = 26), HfSiO₄ (k = 27), are strong contenders to replace SiO₂ when the channel length is scaled down below 32nm [Park *et al.* 2015, Robertson (2004)].

1.4.4 Dual-Metal-Gate (DMG) MOSFETs

Dual-Metal-Gate (DMG) structure which was proposed by Long *et al.* [Long *et al.* 1999] is one of the prominent CMOS technology boosters today. The cross-sectional view of dual-metal-gate (DMG) FET is shown in Fig. 1.4. The gate electrode of a DMG FET is made of two different materials which are termed as control and screen gates with work function ϕ_{M_1} and ϕ_{M_2} respectively. The work function of the control gate (ϕ_{M_1}) is higher than that of screen gate (ϕ_{M_2}) i.e. $\phi_{M_1} > \phi_{M_2}$ for an *n*-channel MOSFET and vice versa for a *p*-channel MOSFET [Long *et al.* (1999)]. A device with DMG structure is found to be superior than a conventional one as it possesses excellent short-channel effects (SCEs) immunity, lower hot-carrier effects (HCEs) and drain-induced barrier lowering (DIBL) [Long *et al.* (1999), Chaudhry *et al.* (2004), Kumar *et al.* (2004), Jin *et al.* (2010)].



Fig. 1.4: Cross-sectional view of a dual-metal-gate (DMG) FET [Long et al. (1999)]

1.4.5 Non-Conventional MOSFETs

Beyond 32nm technology node, non-conventional MOS structures have been pursued relentlessly in order to continue the scaling of MOS structures. Non-conventional MOSFETs may be classified into following two broad categories (i) planar silicon-oninsulator (SOI) MOSFETs, and (ii) non-planar multi-gate MOSFETs. In SOI MOSFETs, a thin Si channel is grown on a thick buried-oxide (BOX) insulator which is grown on a Si substrate. The thin Si channel on BOX causes the reduced parasitic capacitances, improved electrical isolation, better immunity to radiation-induced leakage current (RILC) and diminished latch-up effect. SOI MOSFETs can be categorized into partially-depleted (PD) SOI MOSFETs and fully-depleted (FD) SOI MOSFETs. In the PD devices, the silicon film is made to be much thicker than the maximum gate depletion width so that a neutral body region exists below the gate depletion boundary as shown in Fig. 1.5 (a). On the other hand, the silicon channel thickness of fully-depleted SOI MOSFETs is made to be thin enough so that the entire film is depleted under the zero-bias condition of the device. The schematic structure of a FD SOI MOSFET is shown in Fig. 1.5 (b).

It may be mentioned that unlike the bulk MOSFETs, the body of PD-SOI devices is not tied to the ground. In other words, the body can be floated to different potentials depending on the applied drain and gate voltages. When the device operates under large drain bias conditions, carriers of the same type as that of the neutral body may be generated by impact-ionization near the drain and get stored in the neutral region of the device. The stored charges may alter the body potential and hence the threshold voltage of the device leading to an adverse phenomenon called the floating body effect of the SOI MOSFETs [Yoshimi *et al.* (1989)].



Fig. 1.5: Schematic structure of (a) partially-depleted (PD) SOI MOSFET (b) fully-depleted (FD) SOI MOSFET

The SOI MOSFETs have a number of advantages over the bulk MOSFETs. The advantages of ultra-thin SOI MOSFETs include reduced junction capacitance, improved electrical isolation, the possibility of optimal operation with relatively light channel doping, better immunity to radiation-induced leakage current and diminished latch-up effect [Colinge (2004), Celler et al. (2003), Markov et al. (2012)]. Further, SOI devices have lower SCEs than the bulk devices. The buried-oxide (BOX) cuts off most of the leakage current path for a MOSFET fabricated on it. Another important merit of SOI technology is that it provides the base for new device structures such as multi-gate fieldeffect transistors (MuGFETs) as shown in Fig. 1.6. The MuGFETs may be classified as double-gate planar SOI transistor, double-gate non-planar FinFET, tri-gate FET, quadruple-gate FET, gate-all-around (or surrounding-gate) FET and nanowire FET [Park et al. (2001), Yang et al. (2002), Colinge (2004), Jiménez et al. (2004), Manoj et al. (2008), Bangsaruntip et al. (2009), Huang et al. (2010), Kuhn (2012)]. The MuG MOS structures provide extra gate control over the channel region. Hence, effective suppression of the off-state leakage current and SCEs can be achieved [Colinge (2008)]. Additionally, multi-gate MOS structures have the flexibility that they may be controlled by a single-gate electrode, wherein the multiple-gate surfaces act electrically a singlegate, or by independent-gate electrodes [Ferain et al. (2011)].

However, several technological problems of MuGFETs must be solved before, to use MuGFETs in VLSI circuits. Some of these problems are discussed below.

- The design and fabrication process of non-planar structures (MuGFETs) are more complicated than the planar structures (SOI MOSFETs). MuGFETs require more advanced fabrication techniques such as improved etching accuracy and reliability [Pacha *et al.* (2006), Masahara *et al.* (2009)].
- MuGFETs exhibits a very undesirable characteristic known as corner-effect which occurs due to the electrostatic coupling between two adjacent gates at the corners. This effect degrades the device performance by increasing the off-state leakage current [Zhao (2006), Yu *et al.* (2007)].
- Another important challenge associated with the performance and reliability of MuGFETs is that in order to retain the desired values for threshold voltage, subthreshold current and subthreshold swing, the width of the fin needs to be decreased as gate length is decreased [Masahara *et al.* (2009)]. This imposes a difficult challenge on the fabrication process.



Fig. 1.6: Various SOI device: (a) single-gate SOI transistor, (b) double-gate planar SOI transistor, (c) double-gate non-planar FinFET, (d) tri-gate FET, (e) quadruple-gate (or gate-all-around) FET, (f) gate-all around (or surrounding gate) FET (nanowire FET). [Kim (2010)]

1.4.6 Back-Gated SOI MOSFETs

SOI MOSFETs are very attractive for low power applications because of many reasons including lower junction capacitance, steeper subthreshold slope, and higher short-channel effect immunity [Kumar *et al.* (2008)]. In addition, back-gate of SOI MOSFETs may be used to control the threshold voltage of the structure. The cross-sectional view of a fully-depleted SOI MOSFETs with back-gate control is shown in Fig. 1.7. The front-gate can be used to switch the device, whereas the back-gate can be used to set the correct threshold voltage.



Fig. 1.7: Cross-sectional view of a fully-depleted SOI MOSFETs with back-gate control

1.5 SOI MOSFET

In SOI-based CMOS technology, the conventional silicon substrate of a bulk MOSFET is replaced by a layered silicon-insulator (i.e. SiO₂ or sapphire)-silicon substrate to reduce parasitic device capacitances. The top layer of the SOI substrate which is used as the channel of the device is usually a very thin layer of crystalline silicon. The thin silicon layer is electrically isolated from the substrate by a thick (typically 100nm or more) buried-oxide layer. Many techniques have been developed for producing a film of single crystal silicon on an insulator [Colinge (2004)]. Silicon-on-insulator structure can be produced from a bulk silicon wafer by isolating a thin silicon layer from the substrate through the Full isolation by porous oxidized silicon (FIPOS) or through the ion beam synthesis of a buried insulator layer separation by implanted oxygen (SIMOX) [Colinge (2004)]. Moreover, there is another layer transfer technique in which the smart-cut has enabled the

"peeling-off" of a thin silicon layer from a wafer and its transfer onto an oxidized wafer. The approaches for making SOI wafers have been discussed in section 1.5.1.

1.5.1 SOI Manufacturing Technologies

• SOS (Silicon-on-Sapphire)

Silicon-on-sapphire (SOS) is obtained by an epitaxial growth of a thin layer of silicon on a sapphire (Al₂O₃) wafer. SOS is used for manufacturing of high-frequency and highpower CMOS integrated circuits in satellite communication systems, cellular telephones, and others [Internet resource (IR3)]. In the early days, SOS faced many difficulties in the commercial manufacturing of small-scale transistors. Later, various techniques including solid-phase epitaxial and regrowth (SPEAR), double solid-phase epitaxial (DSPE), were developed to reduce the problem associated with the fabrication of SOS based transistors [Paul *et al.* (2004)]. The SOS wafer fabrication process is illustrated in Fig. 1.8.



(c) Heat treatment & surface oxidation

Fig. 1.8: SOS wafer fabrication flow

• SIMOX (Separation by Implanted Oxygen)

In SIMOX technique, oxygen atoms are implanted into Si wafer in order to synthesise SiO_2 . The formation of SiO_2 by O_2 -ion implantation into Si wafer was first reported by Watanabe and Tooi in 1966 [Watanabe *et al.* (1966)]. Later, in 1978, K. Izumi *et al.* [Izumi (1978)] invented a new SIMOX technique and fabricated a 19-stage CMOS ring

oscillator using a buried SiO₂ layer formed by O₂-ion ($^{16}O^+$) implantation into silicon. The SIMOX process is shown in Fig. 1.9.



Fig. 1.9: SIMOX process

• Smart-Cut

Using smart-cut technique, a very thin surface layer of crystalline silicon material is transferred onto the insulating substrate. In 1995, Michel Bruel invented the smart-cut process [Bruel *et al.* (1995)]. This process contains three major steps which are: implantation of H^+ ions, bonding to a stiffener, and thermal annealing [Colinge (2004)]. The wafers which are fabricated by the smart-cut process are called as UNIBOND wafers [Colinge (2004)]. The sequence of graphics given in Fig. 1.10 describes the process of SOI wafer fabrication using the smart cut technology.




1.5.2 Salient Features of SOI MOSFETs

SOI MOSFETs offer several advantages over bulk silicon CMOS such as high speed, low power dissipation, low drain/source junction capacitances, low leakage currents, high short-channel effects immunity, high subthreshold voltage swing, and better manufacturing compatibility with the existing bulk silicon CMOS technology [Colinge *et al.* (2004), Kumar *et al.* (2008)]. Some of the advantages of SOI MOSFET are discussed below.

• Low Drain Junction Capacitance:

In bulk MOSFETs, the parasitic capacitance at the drain junction consists of two components: Capacitance between the drain/substrate junction and the capacitance between the drain and the channel stop implant under the field oxide as shown in Fig 1.11(a). As scaling the MOSFETs to nanometer regime, highly doped substrate regions are inevitable which leads to high junction capacitances. On the other hand, SOI MOSFETs mainly contains only one parasitic capacitance which exits between the buried oxide and the silicon substrate. This capacitance is typical much lower when compared to the capacitance of bulk MOSFETs [Colinge (2004)]. Thus, reduced parasitic capacitances in SOI MOSFETs contribute to the high performance. In addition, the buried oxide thickness does not necessarily to be scaled down with device miniaturization.



(a) (b)

Fig. 1.11: Junction capacitances [Colinge (2004)]

• Reduced Short-Channel Effects (SCEs) :

In SOI MOSFETs, SCEs such as threshold voltage roll-off and DIBL are mainly influenced by the channel thickness, buried oxide thickness, substrate doping and channel doping concentration. The threshold voltage roll-off occurs at the smaller gate voltages in SOI MOSFETs than in bulk MOSFETs, by using ultra-thin SOI film the threshold voltage roll off can be reduced. Another SCE, DIBL occurs due to the charge sharing between the gate and the S/D junctions. However, in ultra-thin SOI MOSFETs, the channel region prohibits drain to take control of channel charges. [Colinge (2004)].

• No Latch-Up :

Latch-up is an unwanted condition occurs in bulk CMOS devices as shown in Fig. 12 (a). The triggering of inherent PNPN thyristor structure formed by the Bi-polar transistors in bulk CMOS devices lead to severe problems in device performance. However in SOI CMOS devices, the latch-up problem is ruled out as there is electrically isolated thick buried oxide layer between the devices as shown in Fig 1.12(b) [Matloubian (1989), Kushwaha (2011)].



(a)



(b)

Fig. 1.12: Cross-sectional view of (a) bulk CMOS (b) SOI CMOS

1.5.3 Challenges and Remedies

Although SOI technology is already a mature technology, there are still some serious challenges exist. Primary challenges include fabrication of ultra-thin film, thin box, excellent thickness uniformity and lower defect content [Colinge (2004)]. Further, due to the ultra-thin source and drain regions, FD SOI MOSFETs possess large series resistance which leads to the poor current drive capability of the device despite having excellent short-channel characteristics [Zhang et al. (2004)]. Furthermore, interconnects are essential to be a minimum of sub-10nm in diameter in order to connect extremely small devices [Li et al. (2009)]. As the interconnect wires diameter scale downs to the mean free path of electrons, the surface scattering and boundary scattering would delay the electronic conduction in wires, [Li et al. (2009)]. Therefore, the circuit operation becomes slower and it is not promising to attain performance enhancement from scaling [Li et al. (2009)]. Hence, the FD SOI MOSFET structure needs some structural modifications to overcome these problems. The modification can be done in two possible ways (i) by elevating S/D regions upward (ii) by recessing S/D regions deeper into the buried oxide. Based on these, two different kinds of FD SOI MOSFET structures have been proposed as follows:

• Elevated Source/Drain (E-S/D) SOI MOSFETs

The elevated source/drain (E-S/D) MOSFET was proposed and fabricated by Wong *et al.* [Wong *et al.* (1984)]. In Fig. 1.13 (a), the elevated source/drain (E-S/D) regions are created by growing selective epitaxial silicon at the top of source/drain regions. In E-S/D SOI MOSFET, the elevated source/drain regions are separated from the gate electrode by gate spacers. As a result, a parasitic capacitance (Miller capacitance) coming from the coupling between the gate and the elevated source/drain regions [Zhang *et al.* (2004)]. Further, E-S/D regions are affected with current crowding effects which form a non-homogenous distribution of current density through a semiconductor, especially at the surrounding area of the contacts and over the PN junctions and leads to localized overheating and formation of thermal hot spots [Thean *et al.* (2006)]. Moreover, due to the elevated source/drain regions in E-S/D SOI MOSFET, the device structure loses its planar nature. Hence, E-S/D may not be compatible with planar CMOS technology.

• Recessed-Source/Drain (Re-S/D) SOI MOSFETs

In Re-S/D SOI MOSFET, the large series resistance is cut down by extending the source and drain regions deeper into the buried-oxide (BOX) [Hanafi *et al.* (2005), Ahn *et al.* (2004), Ahn *et al.* (2006), Wu *et al.* (2015)]. The Re-S/D ultra-thin body (UTB) SOI MOSFET was designed and fabricated by Zhang *et al.* [Zhang *et al.* (2004)]. The crosssectional view of Re-S/D SOI MOSFET is shown in Fig. 1.13 (b). By using the wet etching process two trenches have been created at both edges of thin SOI film to remove the thin Si film and BOX for the formation of source/drain and Re-S/D regions. The formed trenches will be filled with the amorphous silicon to act as the S/D and Re-S/D regions. The filled amorphous silicon is planarised up to the gate oxide region by using the chemical mechanical planarization (CMP) technique. Further, to form the p^+ S/D and p^+ Re-S/D regions, the amorphous silicon regions have been heavily doped with the arsenic material. Finally, the back-end steps like contact hole etching, metal deposition, and metal patterning steps are performed to complete the process.

In case of Re-S/D SOI MOSFET, the capacitance coupling between the gate and source/drain regions are almost eliminated. Further, Re-S/D SOI MOSFETs are free from

current crowding effect [Thean *et al.* (2006)]. Moreover, unlike an E-S/D SOI MOSFET, the Re-S/D SOI MOSFET is a planar structure. Thus, the Re-S/D SOI MOSFET seems to be an excellent device structure which is compatible with planar CMOS process technology.



Fig. 1.13: Schematic cross-sections of UTB SOI MOSFETs with (a) recessed source/drain (Re-S/D) structure and (b) elevated source/drain (E-S/D) structure

1.6 Dissertation Motivation

In the era of sub-100nm device dimensions, continuing conventional MOSFET scaling and keeping up with the ITRS roadmap are proved to be more difficult than ever. This is attributed to the severe SCEs in the shrinking nanometer order. In the conventional MOSFET, as the channel length diminishes, the gate control over the channel gets reduced due to the increased source/drain capacitances. Various alternative MOS structures have been pursued in order to further continue the scaling. Among all of them, SOI MOSFET is one of the alternative CMOS devices that can be scaled more aggressively than the bulk silicon MOSFET. The device offers diminished SCEs, smaller subthreshold swing, smaller mobility degradation and higher on-to-off current ratio compared to the conventional MOS devices [Colinge (2004), Kumar *et al.* (2008)]. Being three-dimensional structures, multi-gate FETs (MuGFETs) have several technological problems such as fabrication complexity, corner effect and fabrication of thin silicon (fin) etc. Hence, a planar fully-depleted (FD)

SOI MOSFETs could be the best option as the device offers almost equivalent on-off current ratio.

An ultra-thin-body (UTB) is used in FD SOI MOSFETs in order to improve the shortchannel effects immunity. However, the UTB gives rise to a large series resistance and contact formation problems which lead to the poor current drive capability of the device despite having excellent short-channel characteristics. Since, an UTB is required to diminish the SCEs, the source/drain area may be increased by extending S/D either upward or downward. Thus, two schemes discussed in the last section can be employed in FD SOI MOSFETs to counter the large series resistance and contact problem by elevating or recessing source/drain regions. By keeping in mind the issues associated with E-S/D SOI MOSFETs, the Re-S/D SOI MOSFETs have been pursued rigorously to address the large series resistance problem.

Recessing source and drain deeper into buried oxide ensures high drive current and avoid the contact formation problem. However, the recessed source and drain may deteriorate the subthreshold characteristics, including threshold voltage, subthreshold current and subthreshold swing of the transistor owing to a strong coupling between the recessed source/drain and the back surface of the channel. It may happen that the strong coupling between recessed source/drain and back of the channel activates the channel at the back surface before the gate voltage does the same on the front-surface. If it happens, the device will show an anomalous subthreshold behavior.

In light of above-mentioned facts, an attempt has been made to analyze the subthreshold behavior of Re-S/D SOI MOSFETs. Theoretical models of the threshold voltage, subthreshold current and subthreshold swing have been developed for Re-S/D SOI MOSFETs. The impact of various device parameters on the subthreshold characteristics of Re-S/D SOI MOSFETs has been analyzed through the developed theoretical models.

Further, CMOS performance boosters such as high-k dielectric, multi-gate and back-gate etc. help to improve the on-current of FD Re-S/D SOI MOSFETs. Therefore, these performance boosters have been employed in the Re-S/D SOI MOSFETs in order to investigate their impact on the subthreshold characteristics of the device.

20

1.7 Scope of Dissertation

The main objective of this dissertation is to present a detailed modeling & simulation - based study of the subthreshold characteristics of short-channel fully-depleted (FD) recessed-source/drain (Re-S/D) SOI MOSFETs. The dissertation consists of six chapters, including the present one as Chapter 1. The contents of the remaining chapters of the dissertation are outlined as follows:

Chapter 2 presents a detailed review of the modeling and simulation of SOI MOSFETs in detail and Re-S/D SOI MOSFET in particular. The state-of-the-art works reported on the threshold voltage, subthreshold swing and subthreshold current of Re-S/D SOI MOSFETs have been discussed in this chapter. Finally, some major observations from the literature have been summarized at the end of the chapter to justify the motivations behind the research works carried out in the following three chapters of the present dissertation.

Chapter 3 includes the analytical modeling and simulation of the threshold voltage, subthreshold current and subthreshold swing of short-channel dual-metal-gate (DMG) recessed-source/drain (Re-S/D) SOI MOSFETs. The formulated models are derived considering the appropriate physics of the device operation. The model results are analyzed with wide variations in the device parameters. The validity of the present models is confirmed by comparing the model results with the numerical simulation results.

Chapter 4 presents the analytical modeling and simulation of the threshold voltage, subthreshold current and subthreshold swing of Re-S/D SOI MOSFETs with the high-*k* dielectric material. The two-dimensional (2D) Poisson's equation is solved in the channel region in order to obtain the surface potential under the assumption of the parabolic potential profile in the transverse direction of the channel with appropriate boundary conditions. Subsequently, threshold voltage, subthreshold current and subthreshold swing models are presented by considering the influence of the fringing field in terms of bottom plate capacitance C_{bot} which is formed between bottom edge of the gate electrode and the source/drain regions. Following which, all the developed models are tested by

varying the different device parameters. Model results are also compared with the 2D numerical simulation results in order to show their validity.

Chapter 5 presents the analytical modeling of subthreshold characteristics of Re-S/D SOI MOSFETs with back-gate control. The impact of back-gate on the subthreshold characteristics of Re-S/D SOI MOSFETs has been analyzed in detail. Again, to confirm the validity of presented models, 2D numerical simulations are carried out.

Chapter 6 includes the summary and conclusions of the present dissertation. The major findings of the entire dissertation have been summarized in this chapter. In addition, suggestions to future the work in this thesis is presented at the end of this chapter.

Chapter 2

SOI MOSFETs: A General Review

2.1 Introduction

MOSFET miniaturization has enabled amazing improvement in the switching speed [Critchlow (2007)], density [Taur et al. (1998)], functionality [Iwai (2009)] and cost of microprocessors [Cavin et al. (2012)]. However, the problems associated with miniaturization of the conventional MOS transistors have increased with the increasing transistor density in integrated circuits (ICs) [Ieong et al. (2006)]. The CMOS performance boosters, which were discussed in Chapter 1, are very useful in order to counter the problems associated with miniaturization [Takagi (2007), Yang et al. (2006), Chaudhry et al. (2004), Kumar et al. (2004), Bangsaruntip et al. (2009), Kuhn (2012), Amlan et al. (2009)]. In Chapter 1, it was also discussed that the Fully depleted (FD) silicon-on-insulator (SOI) MOSFET is one of the potential CMOS devices that can be scaled down to a greater extent compared to the bulk silicon MOSFET because of their outstanding short-channel effects (SCEs) controlling capability [Taur et al. (1997), Colinge (2008)]. However, owing to the ultra-thin source and drain regions, FD SOI MOSFETs exhibit large series resistance problem which results in the diminished current drive capability of the device despite possessing excellent short-channel characteristics [Chaudhry et al. (2004)]. In order to reduce the effect of this large series resistance problem, Zhang et al. [Zhang et al. (2004)] proposed and fabricated a recessed-source/drain (Re-S/D) ultra-thin body (UTB) SOI MOSFET. In the Re-S/D UTB SOI MOSFETs, source and drain regions are extended deeper into the buried oxide (BOX) to mitigate the effects of large source and drain resistances.

The objective of this dissertation is to perform the modeling and simulation of the subthreshold characteristics of some SOI MOSFET structures. Since, the future trends in research, in any area can only be predicted by having a thorough knowledge of the state-of-the-art research in that particular field of interest, the present chapter is dedicated to discuss a detailed review of the state-of-the-art-work on various aspects of SOI MOSFETs in general, and recessed-source/drain SOI MOSFETs in particular, in order to validate the scope of the dissertation outlined in the previous chapter.

The layout of the chapter is as follows. Sections 2.2 and 2.3 include the review of FD SOI MOSFETs. Some state-of-the-art works on the modeling and simulation of the subthreshold characteristics of SOI MOSFETs have been briefly discussed in section 2.4. Finally, some major findings of the literature review have been summarized in section 2.5.

2.2 SOI MOSFETs

Fully depleted (FD) SOI MOSFETs beat the conventional MOSFETs in various aspects like ability to operate in critical environment such as high temperature [Francis *et al.* (1992)], SCEs immunity [Colinge *et al.* (2004)], high drive current capability [Chaudhry *et al.* (2004)], as well as low parasitic capacitances and leakage currents [Markov *et al.* (2012)]. In addition, the SCEs in a FD SOI MOSFET can be suppressed by using a thinner silicon body and BOX [Ghanatian *et al.* (2009)]. Further, the FD SOI MOSFET with a BOX thickness of less than 5nm may be treated as a double-gate device structure on SOI substrate [Ghanatian *et al.* (2009)]. In contrast to BOX thickness scaling, the silicon film thickness scaling is more feasible from a technology point of view [Trivedi *et al.* (2005)], which generates a great deal of interest in ultra-thin body FD SOI MOSFETs.

However, as the film thickness decreases, the source/drain resistance increases and can attain significantly high value to degrade circuit speed performance [Chaudhry *et al.* (2004)]. The source/drain resistance of the device can be pulled down by various techniques such as silicidation of source/drain, tungsten clad, schottky source/drain,

elevated-source/drain (E-S/D), and recessed-source/drain (Re-S/D) etc. [Chen *et al.* (2007), Zan *et al.* (2003), Ieong *et al.* (1998), Wong *et al.* (1984), Zhang *et al.* (2004)].

2.2.1 Source and Drain Silicides

Formation of a low-resistance silicide layer on the source/drain is one of the common techniques to reduce source and drain resistance of the device [Chen *et al.* (2007)]. For this low-barrier silicide formation, firstly, metal is deposited on a silicon substrate and then it is energized by thermal heating or laser irradiation or ion beam mixing. The metal reacts with silicon at the interface, leaving a silicide there. This process is depicted in Fig. 2.1 [Chen *et al.* (2007)]. The commonly used silicides are titanium silicide (TiSi₂), cobalt silicide (CoSi₂), platinum silicide (PtSi) and nickel silicide (NiSi). Schematic view of an SOI MOSFET with low-barrier silicide layer above the source and drain regions is shown in Fig. 2.2 [Colinge *et al.* (2004)].



Thermal heating or laser irradiation or Ion beam mixing



Fig. 2.1: Silicide formation





Deng *et al.* [Deng *et al.* (1997)] calculated the source/drain sheet resistance (R_s) against the thickness of the deposited titanium. They observed that the sheet resistance decreased from 300 Ω /square to 2 Ω /square with the increase in titanium thickness from 0nm to 45nm respectively. But, silicon possesses a tendency to diffuse into a titanium silicide. Hence, the silicon in the source/drain and channel region starts diffusing into silicides forming Kirkendall voids [Tu *et al.* (2005)] under the gate edges. This may lead to failure of the device. Therefore, the thinner silicide layer, which gives rise to comparatively lower resistance than the conventional SOI MOSFET, is used in practice.

2.2.2 Tungsten Clad

For the first time, the ultra-thin-channel poly-silicon thin-film transistor (TFT) with tungsten-clad source/drain (W-TFT) was proposed by Zan *et al.* [Zan *et al.* (2003)] to reduce S/D resistance by selective tungsten chemical vapor deposition (SWCVD) technology. SWCVD is one of the best ways to reduce source/drain resistance because of its small silicon consumption. The schematic cross-sectional view of W-TFT is shown in Fig. 2.3 [Zan *et al.* (2003)]. The W-deposition on silicon film can be done by CVD (chemical vapor deposition) process, where SiH₄ chemically reduces WF₆ to W at 250^oC followed by a rapid thermal annealing at 550^oC temperature. Zan *et al.* [Zan *et al.* (2003)] also compared the output characteristics of conventional TFT and W-TFT and they reported that the output characteristics of W-TFTs are better than those of the conventional TFTs.



Fig. 2.3: Schematic cross-sectional view of the W-TFT. [Zan et al. (2003)]

2.2.3 Schottky Source and Drain

The source/drain resistance can be decreased as well by using Schottky source and drain [Ieong *et al.* (1998)]. The Schottky source/drain gives rise to lower sheet resistance, but leads to a higher contact resistance because of its finite Schottky barrier height. Hence, a material with lower barrier height must be used as a contact in order to reduce contact resistance [Ieong *et al.* (1998)]. It had been reported that in short-channel devices, if a very small or negative Schottky barriers developed, then both the source/drain resistance and contact resistance were significantly reduced [Larson *et al.* (2006)]. For a p-channel and n-channel devices, platinum silicide (PtSi) and **e**rbium silicide (ErSi_{1.7}) have been used for Schottky junction respectively [Ieong *et al.* (1998)]. Ieong *et al.* [Ieong *et al.* (1998)] analyzed I-V characteristics of a Schottky source/drain MOSFET for various Schottky barrier heights and observed that the drive current severely degrades with higher Schottky barrier height, but increases substantially with a decrease in barrier height due to decreased contact resistance.

2.2.4 Elevated source and drain (E-S/D) regions

An elevated source and drain regions can also cut down the series resistance of a MOS device. In this, an ultra-thin Si film is used in the channel region along with a thicker Si film for source and drain regions as shown in Fig. 1.13 (a) of Chapter 1 [Wong *et al.* (1984)]. This can be done in either of the two ways, i.e. (1) by adopting elevated source and drain technique [Colinge (2004)], or (2) by adopting the local oxidation of silicon (LOCOS) technique [Colinge (2004)]. In the former approach, firstly an ultra-thin Si film is used and then, a selective epitaxial growth (SEG) is carried out to increase the thickness of source/drain regions [Colinge (2004)]. In the latter approach, a thicker Si film is used first and then, the channel region is thinned using a LOCOS technique [Colinge (2004)].

2.2.5 Recessed-Source/Drain (Re-S/D)

Recessing of source and drain regions diminishes the large series resistance of a MOS device [Zhang *et al.* (2004)]. The cross-sectional view and fabrication process of the

Re-S/D SOI MOSFET have already been presented in section 1.5.4 of Chapter 1. Zhang *et al.* [Zhang *et al.* (2004)] designed and fabricated a self-aligned Re-S/D UTB SOI MOSFET. The comparison of series resistances/parasitic capacitances of E-S/D and Re-S/D structures were presented by Zhang *et al.* (2004). It was found that the E-S/D structure offers an extra parasitic capacitance which is called as a gate to source/drain Miller capacitance. However, increasing the spacer thickness between the gate and the E-S/D can reduce the gate to S/D Miller capacitance, but it also increases the series resistance. Hence, in case of an E-S/D structure, a compromise among series resistance or Miller capacitance is inevitable [Zhang *et al.* (2004)]. Further, in case of Re-S/D structure, there is no spacer adjacent to the gate as in the E-S/D MOSFET, therefore, a fall of 25% miller capacitance is attained in Re-S/D structure [Zhang *et al.* (2004)]. Furthermore, E-S/D regions are also affected by current crowding effects and planarity which were discussed in section 1.5.3 of Chapter 1.

2.3 Earlier Works Related to SOI MOSFETs

The first SOI transistor was fabricated by Mueller et al. in the year 1964 [Mueller et al. (1964)]. Most of the early SOI MOSFETs were fabricated with SOS (silicon-onsapphire) wafer technology. SOI technology was advocated enormously by J. P. Colinge in 1980's [Colinge (1991)]. Introducing a BOX in conventional MOS structure delivers a lot of improved characteristic such as excellent isolation [Colinge (1996)], less leakage current [Markov et al. (2012)], enhanced SCEs immunity [Colinge (2004)], improved latch-up free condition [Narayanan et al. (2013)], enhanced switching speed due to less drain-body capacitance [Markov et al. (2012)] and radiation hardness [Chan et al. (1994)]. The SOI CMOS technology emerged as an alternative to planar CMOS technology for deep sub-micron CMOS. It appeared to be one of the best options for low-power electronics [Colinge (2004)]. SOI MOSFETs attracted a number of researchers [Colinge (1986), Young et al. (1989), Yan et al. (1991), Joachim et al. (1993), Yan et al. (1992), Wong et al. (1998), Kumar et al. (2005), Reddy et al. (2005), Pal et al. (2014), Mohapatra et al. (2003), Kumar et al. (2007), Lu et al. (2011), Zhang et al. (2004), Ke et al. (2007), Svilicic et al. (2009)] due to their several advantages over conventional MOSFETs.

In 1986, J. P. Colinge [Colinge (1986)] presented a subthreshold slope model of thinfilm SOI MOSFETs. One-dimensional (1D) Poisson's equation was solved in the channel region using the suitable boundary conditions to model the electrostatic behavior of SOI MOSFETs. A subthreshold slope of 62 mV/decade, which is almost equivalent to the ideal subthreshold slope of 60 mV/decade, was observed in a longchannel undoped FD SOI device [Colinge (1986)]. However, the near-ideal subthreshold slope cannot be obtained in short-channel FD SOI MOSFETs with thick BOX due to the lateral field coupling into the channel through the BOX [Colinge (1986)].

Young [Young (1989)] derived a two-dimensional (2D) potential distribution equation of FD SOI MOSFETs in the silicon thin film by assuming a simple parabolic function perpendicular to the channel region. It was found that the vertical field through the depleted film strongly influences the lateral field across the source and drain regions. It was observed that the SCEs were found to be decreased with decreasing silicon film thickness. [Young (1989)]. In the early 90's, R. H. Yan [Yan et al. (1991)] also derived a 2D channel potential for SOI MOSFETs following K.K. Young's approach [Young et al. (1989)]. However, both Young and Yan assumed the electric field inside the thick BOX to be zero in order to simplify the problem. Therefore, these models did not show BOX thickness dependency [Joachim et al. (1993)]. Joachim et al. [Joachim et al. (1993)], however, specified that 2D effects in the BOX region could not be neglected for short-channel devices [Joachim et al. (1993)]. Joachim et al. [Joachim et al. (1993)] modified the boundary condition at the bottom of the SOI layer to account for the nonlinear potential distribution inside the BOX. They [Joachim et al. (1993)] found that in a FD SOI MOSFET, the subthreshold slope is improved by increasing the channel doping concentration.

Yan *et al.* [Yan *et al.* (1992)], in year 1992, introduced natural length scale (λ) as a scaling parameter for several SOI devices such as (a) conventional, (b) gate-all-around, and (c) ground plane, which are shown in Fig. 2.4. It was mentioned that in a deep-submicrometer regime, to achieve better subthreshold leakage control, heavily doped devices were essential, but high channel doping leads to undesirable large junction

capacitance and degraded mobility. Hence, Yan *et al.* [Yan *et al.* (1992)] rigorously studied about the scaling of FD SOI devices and found that the horizontal leakage due to short-channel effects could be very less in non-planar structures. It was observed that the gate length of the device needed to be lengthier than λ for obtaining better subthreshold characteristics [Yan *et al.* (1992)]. The minimum scalable channel length of a FD SOI device could be simply projected by its scale length [Yan *et al.* (1992)]. Yan *et al.* (1992)] reported that the minimum channel length of gate-all-around could be scaled by 30% shorter than the conventional SOI MOSFET with good subthreshold characteristics. It was also reported that the ground-plane structure shows an improvement in λ about 75% over the conventional SOI and a 20% over the gate-all-around structure respectively.

Structure	Conventional	Gate-All-Around	Ground Plane
Schematic	gate source drain Silicon oxide (insulator) Silicon substrate	V Source Deniar esk Source Deniar esk Source	Source Gate Drain Gate Oxide L Buried Oxide Ground Plane Back Gate
Scaling	$\lambda = \frac{\varepsilon_{Si}}{\varepsilon_{ox}} t_{Si} t_{ox}$	$\lambda = \frac{\varepsilon_{Si}}{2\varepsilon_{ox}} t_{Si} t_{ox}$	$\lambda = \frac{\varepsilon_{Si}}{2\varepsilon_{ox}} \frac{t_{Si}t_{ox}}{\left[1 + \frac{\varepsilon_{Si}t_{ox}}{\varepsilon_{ox}t_{Si}}\right]}$

Fig. 2.4: Various SOI structures: (a) conventional (b) gate-all-around (c) ground plane. [Yan *et al.* (1992), Sasaki *et al.* (2015), Chena *et al.* (2016)]

Wong *et al.* [Wong *et al.* (1998)] investigated the severity of SCEs in 25nm FD SOI, ground-plane and double-gate (DG) MOSFETs based on 2D simulations. It was observed that the FD SOI MOSFET does not provide adequate short-channel immunity for 25 nm channel length devices even though the channel potential control was increased. Further, in ground-plane MOSFETs, bottom gate insulator must be kept thin to allow effective tuning of the threshold voltage by a bottom gate voltage not exceeding the power supply voltage. However, the thickness of back-gate insulator was considered to be 1.5 times than that of top gate insulator thickness. The design space for the oxide thickness (t_{ox}) and the channel thickness (t_{Si}) were also studied [Wong *et al.*]

(1998)]. The obtained models were validated with the simulation results which were obtained from the device simulators FIELDAY program from IBM [Wong *et al.* (1998)].

Kranti *et al.* [Kranti *et al.* (2010)] investigated the analog/radio-frequency (RF) performance of sub-100nm SOI MOSFETs with gate-source/drain underlap channel for low voltage applications. It was reported that the use of underlap source/drain architecture over conventional one enhances the analog/RF performance. [Kranti *et al.* (2010)]. It was, further, claimed that the underlap channel design possesses high voltage gain up to 30 dB and cutoff frequency up to 100 GHz.

Patil *et al.* [Patil *et al.* (2013)] proposed the asymmetric source/drain (ASD) underlap channel dopant-segregated schottky barrier (DSSB) SOI MOSFET and compared the performance with the same of the symmetric S/D (SSD) overlap and underlap structures. The fabrication steps of the ASD underlap structure were also specified in detail [Patil *et al.* (2013)]. The RF figures-of-merits (FoMs) of the considered structures were extracted using H and Y parameters found by carrying out the small-signal AC analysis [Patil *et al.* (2013)]. It was reported that ASD underlap structure (where the source side and drain side were considered to be overlapped and underlapped, respectively) reduced the gate-induced-drain-leakage (GIDL) and improved the analog figures of merit such as transconductance (g_m) and cutoff frequency (f_T) [Patil *et al.* (2013)]. Further, the optimization study revealed that the mixed-signal circuit performance of the ASD underlap device, however, degraded significantly beyond certain underlap length at drain side [Patil *et al.* (2013)].

In 1999, Long *et al.* [Long *et al.* (1999)] proposed a novel structure named as dualmaterial gate (DMG) FET, where the two gate materials were cascaded such that the source side gate metal (M_1) had a relatively higher work function than the drain side metal (M_2) . The DMG structure provides the benefits of high electron velocity and enhanced source side electric field resulting in increased carrier transport efficiency in the channel region [Long *et al.* (1999)]. Long *et al.* [Long *et al.* (1999)] observed a step-like channel potential profile which ensured screening of the minimum potential point from drain voltage fluctuations. Hence, the metal gate M_2 behaves as the screen gate and the metal M_1 behaved as the control gate.

In 2005, for the first time, Reddy *et al.* [Reddy *et al.* (2005)] developed an analytical threshold voltage model for a double-material double-gate (DMDG) SOI MOSFET. The front-gate and back-gate surface potentials of the DG and DMDG SOI MOSFETs were compared along the channel region [Reddy *et al.* (2005)]. It was observed that the DMDG SOI MOSFET illustrated a step-like profile in the surface potential at both the surfaces [Reddy *et al.* (2005)]. It was reported that in case of an asymmetrical DG SOI MOSFET the surface potential at the back-gate dominates in determining the threshold voltage (V_{th}) .

Pal *et al.* [Pal *et al.* (2014)] recently developed an analytical threshold voltage model for the dual-material surrounding-gate MOSFET (DMSG). The threshold voltage (V_{th}) variations were compared for DMSG and SG MOSFETs [Pal *et al.* (2014)]. It was observed that the DMSG MOSFETs offer higher efficiency to V_{th} roll-off compared to SG MOSFETs [Pal *et al.* (2014)]. It was reported that in gate-engineered MOSFETs, due to the electric field discontinuities in the channel, the increased drain to source potential is absorbed by the screen gate.

In 2010, the concept of DMG was invoked in strained-silicon-on-insulator (s-SOI) MOSFETs by Jin *et al.* [Jin *et al.* (2010)]. It was analytically demonstrated that the significant improvements in the SCEs such as DIBL, HCEs, and carrier transport efficiency could be obtained in strained-SOI MOSFETs [Jin *et al.* (2010a)](2010)].

By utilizing the concepts of the straddle-gate structure of the MOSFETs [Tiwari *et al.* (1998)] and the electrically induced shallow junction MOSFETs [Han *et al.* (2001)], Goel *et al.* [Goel *et al.* (2006)] presented a new structure called tri-material gate-stack (TRIMAGS) MOSFET in 2006. The gate dielectric of their proposed structure consisted of a gate oxide layer on the substrate, followed by a high-*k* dielectric layer. The polysilicon at the gate-electrode was replaced by laterally connected three different materials with three different work functions placed in such a way that the materials

with the highest and lowest work functions were at the source side and drain side respectively.

Chiang [Chiang *et al.* (2009)] reported a subthreshold current model for a tri-material gate-stack (TRIMAGS) SOI MOSFET using evanescent mode analysis. The twodimension (2D) Poisson's equation was solved in the channel region using suitable boundary conditions to obtain channel potential which was finally used to derive the subthreshold current model of the device. It was observed that a thin effective stackgate oxide, a thin silicon body and large $L_1 : L_2$ ratio were required to deliver better subthreshold behavior, where, L_1 and L_2 be the lengths of control and screen gates respectively [Chiang *et al.* (2009)].

As per the device scaling rule [Dennard *et al.* (1985)], an ultra-short-channel length device should have a very thin gate oxide in order to diminish the unwanted shortchannel effects. But, through the ultra-thin gate oxide, quantum mechanical tunneling of the charge carrier is inevitable [Basak *et al.* (2015)]. Therefore, the gate dielectric material (SiO₂) should be replaced by a high-*k* gate dielectric material (HfO₂, Ta₂O₅ and ZrO₂) which offers large physical thickness and small electrical thickness [Tripathi *et al.* (2012)]. However, the high-*k* dielectric material induces fringing field lines, which influence the channel electrostatic potential; hence, it is very important to examine the subthreshold characteristics of the device in the presence of the high-*k* dielectric material. Several attempts have been made in the past, to model the capacitance associated with the bottom edge of the gate electrode and the source and drain regions for different CMOS devices [Kumar *et al.* (2007), Mohapatra *et al.* (2003), Liu *et al.* (2002)].

Mohapatra *et al.* [Mohapatra *et al.* (2003)] presented parasitic capacitance models of MOS transistors with high-k dielectric material, as shown in Fig. 2.5. It was reported that three types of parasitic capacitances might affect the device: the first one is C_{side} , the sidewall fringing capacitance between gate and source/drain regions (through high-k); the second one is C_{pp} , the parallel plate capacitance between gate and source/drain electrodes and the last one is C_{top} , the top fringing capacitance between top surface of

the gate and source/drain electrodes through the first layer of passivation dielectrics [Mohapatra *et al.* (2003)]. Mohapatra *et al.* [Mohapatra *et al.* (2003)] used the developed model to study the dependence of parasitic capacitance on different parameters such as channel length, gate dielectric constant, gate electrode thickness, gate dielectric thickness, and spacer thickness.



Fig. 2.5: Cross-sectional view of a MOS transistor with various capacitance components [Mohapatra *et al.* (2003)]

Further, Kumar *et al.* [Kumar *et al.* (2006)] presented a parasitic capacitance model for high-k gate dielectric SOI MOSFETs by considering the effect of the parasitic internal fringe capacitance allied with the bottom edge of the gate electrode and the source and drain regions. A threshold voltage model including the effect of fringing capacitance was also presented by Kumar *et al.* [Kumar *et al.* (2006)]. It was observed that there is a significant threshold voltage drop because of fringing field lines from the bottom edge of the gate electrode to the source/drain region for high-k dielectric. In addition, it was also mentioned that this significant threshold voltage drop might degrade the device characteristics and performance considerably.

Considering the variability problem of threshold voltage owing to the random dopant fluctuation in the doped channel, the back-gate biasing is one of the unique features of MOSFET devices for ultimate tuning of threshold voltage even after the device has been fabricated [Yang *et al.* (2013)]. Hence, one approach to control the threshold voltage

fluctuation is to apply an appropriate back-gate bias voltage with low doped substarte [Yang *et al.* (2013)]. Thus, the threshold voltage of FD SOI MOSFETs can be controlled by three parameters, the gate work function (ϕ_M), the channel doping (N_a), and the back-gate voltage, (V_{BG}) [Numata *et al.* (2004)].

Yang *et al.* [Yang *et al.* (1995)] presented a novel SOI device called FD silicon-oninsulator-with-active substrate (SOIAS) MOSFET structure in which an oxide-isolated polysilicon back-gate electrode was formed beneath the channel and back-gate was used to control the threshold voltage of the device. The fabrication of SOIAS MOSFET using bonded SIMOX and BE (bond and etch-back) -SOI processes was presented in detail [Yang *et al.* (1995)]. It was reported that a dynamic threshold voltage control scheme was needed for high-performance and low-power requirements [Yang *et al.*].

Lu *et al.* [Lu *et al.* (2011)] presented an analytical threshold voltage model for FD SOI MOSFETs in which the front-gate and back-gate were independently biased. Lu *et al.* [Lu *et al.* (2011)] considered the non-ideal effects such as short-channel effects (SCEs), quantum effects, leakage currents, field dependent mobility, and parasitic effects to model the characteristics of a real device. The threshold voltage (V_{th}) versus back-gate voltage (V_{BG}) for both symmetric and asymmetric double-gate (DG) MOSFETs were compared. The derived model was validated with the simulation results which were obtained from the device simulator TCAD.

Ultra-thin-body (UTB) SOI MOSFETs exhibit excellent short-channel characteristics such as low parasitic capacitances and leakage currents; but suffer from poor contacts and source/drain resistances because of ultra-thin source/drain regions [Chaudhry *et al.* (2004)]. As discussed in section 2.2, Re-S/D SOI MOSFETs not only facilitates better contacts but also possesses smaller source/drain resistances. Further, the Re-S/D SOI MOSFET fabrication process steps are almost similar to those of CMOS. Hence, due to the above advantages of Re-S/D SOI MOSFETs, researchers have developed an interest towards it.

Ke *et al.* [Ke *et al.* (2007)] performed a comprehensive study on the Re-S/D UTB SOI MOSFET in detail using a commercial simulator ISE 7.0. The Re-S/D structure offer

significantly reduced source/drain series resistance than the E-S/D structure [Ke *et al.* (2007)]. Further, it was also observed that the E-S/D SOI MOSFETs show degraded SCEs immunity than that of Re-S/D SOI MOSFETs. The design guidelines and design window were also proposed for the Re-S/D SOI MOSFETs by Ke *et al.* [Ke *et al.* (2007)]. It was claimed that the Re-S/D SOI MOSFETs were capable of scaling down to 10nm node [Ke *et al.* (2007)].

Zhang *et al.* [Zhang *et al.* (2004)] presented a drain current characteristic of the Re-S/D SOI MOSFET with a reasonably good on and off current ratio of 3.8×10^7 . The step by step procedure for the fabrication of UTB Re-S/D SOI MOSFET was given by Zhang *et al.* [Zhang *et al.* (2004)]. Zhang *et al.* [Zhang *et al.* (2004)] compared the performance of E-S/D and Re-S/D SOI structures. It was reported that the UTB Re-S/D SOI MOSFET had many advantages over UTB E-S/D SOI MOSFET, such as lesser source/drain series resistance and reduced Miller capacitance [Zhang *et al.* (2004)].

Svilicic *et al.* [Svilicic *et al.* (2009)] developed an analytical threshold voltage model for Re-S/D SOI MOSFETs. It was pointed out that in case of Re-S/D SOI MOSFET, the front-channel got inverted just like the conventional SOI MOSFET for lower recessed S/D length, but as the recessed source/drain thickness was increased, the source/drain coupling with the back-channel dominated over the gate coupling with the front-channel, which lead to an early inversion in the back-channel [Svilicic *et al.* (2009)]. The back-channel barrier height found to be smaller compared to the front-channel [Svilicic *et al.* (2009)]. Since the back inverted channel remained too far from the front-gate, and also under the control of source and drain, poor subthreshold characteristics of the device were inevitable.

2.4 Subthreshold Characteristic Models of SOI MOSFETs

Moreover, many attempts have also been made to model and investigate the subthreshold characteristics of SOI MOSFETs [Lee *et al.* (1989), Woo *et al.* (1990), Banna *et al.* (1995), Suziki *et al.* (1995), Svilicic *et al.* (2009), Kumar *et al.* (2014), Yeh *et al.* (1995), Yan *et al.* (1992), Kumar *et al.* (2013), Ding *et al.* (2011), Chen *et al.*

(2011), Dey et al. (2008)]. Few of these attempts are summarized in forthcoming sections.

2.4.1 Threshold Voltage Models of SOI MOSFETs

The threshold voltage (V_{th}) of a MOSFET is defined as the gate-to-source voltage at which the conduction of current begins. The role of V_{th} has become increasingly important in designing of VLSI circuits and systems targeting low-voltage, low-power, and high-speed applications [Schrom *et al.* (1996), Giustolisi *et al.* (2003)]. A number of attempts have been made to model the threshold voltage of SOI MOSFETs [Lee *et al.* (1989), Woo *et al.* (1990), Banna *et al.* (1995), Suziki *et al.* (1995), Svilicic *et al.* (2009), Kumar *et al.* (2014)].

Lee *et al.* [Lee *et al.* (1989)] presented a numerical study of the threshold voltage of a long-channel lightly doped ultra-thin SOI MOSFET, where Si-film was assumed to be thin enough so that complete depletion of the film could take place at the threshold condition. The results obtained by Lee *et al.* [Lee *et al.* (1989)] were limited only to the conditions at which the back channel was neither accumulated nor inverted enough to cause significant floating body effects or leakage current. It was found that the classical definition of the threshold voltage condition (i.e. $\phi_{sf} = 2\phi_B$, where ϕ_{sf} was the front-surface potential and $\phi_{f_{st}}$ was the Fermi potential of the film) of the conventional bulk CMOS device could not be considered for SOI MOSFETs [Lee *et al.* (1989)]. Instead, Lee *et al.* [Lee *et al.* (1989)] assumed that the electron concentration of the front-surface must reach a critical value η_T at the threshold condition and accordingly, front-surface potential was approximated as $\phi_{sf} = \phi_{f_{st}} + \phi_o$ [Lee *et al.* (1989), Chen (2003)]

where
$$\phi_o = \phi_{f_{S_i}}$$
 if $N_a > \eta_T$ or $\phi_o = \frac{kT}{q} \ln\left(\frac{\eta_T}{n_i}\right)$ if $N_a < \eta_T$, $-q$ was the electron charge,

 N_a was the channel doping concentration, n_i was the intrinsic carrier concentration, k was the Boltzmann's constant, and T was the temperature. Based on the simulation results, Lee *et al.* [Lee *et al.* (1989)] concluded that high-performance sub-micrometer complementary MOSFETs could be achieved by using lightly doped or near-intrinsic

ultra-thin SOI films, gate materials with an appropriate work function, and back-gate biasing of the SOI MOSFET. Further, Balestra *et al.* [Balestra *et al.* (1987)] discovered an important physical phenomenon 'volume-inversion' in DG SOI MOSFETs in the year 1987. Volume inversion is a phenomenon which appears in thin-film multi-gate SOI MOSFETs. As per volume inversion concept, the inversion carriers are not confined near the Si-channel/SiO2 interface, but somewhat at the center of the film [Balestra *et al.* (1987)].

In year 1995, Banna *et al.* [Banna *et al.* (1995)] developed a threshold voltage model for FD SOI MOSFET using a quasi-2D approach in the sub- μ m regime. It was observed that the threshold voltage roll-off in FD SOI MOSFET was lesser than that of conventional MOSFET for the same effective channel length. Banna *et al.* [Banna *et al.* (1995)] used experiential fitting parameter ' η ' while finding the characteristic length of FD SOI MOSFET. The obtained model was validated with the simulation results which were obtained from MINIMOS5 and found to be in excellent agreement.

Suzuki *et al.* [Suziki *et al.* (2003)] presented an analytical threshold voltage model for FD single-gate SOI MOSFETs by assuming the 2D effects in both silicon film and BOX. This model is valid for both long and short-channel SOI MOSFETs. However, Suzuki *et al.* [Suziki *et al.* (2003)] limited their investigation to the highly doped base substrates only. It was stated that the lightly doped substrates gives negative V_{th} in most cases.

Zhang *et al.* [Zhang *et al.* (2008)] derived an analytical 2D model of the threshold voltage of a FD SOI MOSFET with vertical Gaussian doping profile by considering the 2D effects in both SOI and BOX [Zhang *et al.* (2008)]. Zhang *et al.* [Zhang *et al.* (2008)] limited their analysis with an approximation that the lateral channel doping concentration was assumed uniform and the vertical doping concentration be non-uniform distribution (Guassian profile). The resulted characteristic lengths of the model was in agreement with Suzuki *et al.* [Suziki *et al.* (2003)] uniformly doped FD SOI MOSFETs model, when $\sigma_p \ll R_p$; where, R_p and σ_p were the projected range and straggle of the Gaussian doping profile.

In the year 2014, our group [Kumar *et al.* (2014)] presented a threshold voltage model of short-channel Re-S/D UTB SOI MOSFETs considering the substrate induced surface potential (SISP) effects. Kumar *et al.* [Kumar *et al.* (2014)] derived the surface potential expression of the front-surface and the back- surface of the channel region using the evanescent mode analysis method. It was observed that the recessed thickness (t_{rsd}) is an important parameter to decide the threshold voltage of the device and the threshold voltage is switched from back surface dominance to the front-surface dominance of the channel as channel doping (N_a) is increased [Kumar *et al.* (2014)].

2.4.2 Subthreshold Current Models of SOI MOSFETs

Since the subthreshold current plays key role in determining the static power dissipation of any MOS device, an accurate subthreshold current model of an SOI MOSFET could be an extreme concern for researchers for the optimization of the power dissipation in short-channel FD SOI MOSFET based VLSI/ULSI circuits [Yeh *et al.* (1995)]. Various attempts have been made to model the subthreshold current of FD SOI MOS devices [Yeh *et al.* (1995), Yan *et al.* (1992), Kumar *et al.* (2013)].

Yeh *et al.* [Yeh *et al.* (1995)] presented a physical subthreshold current model of FD SOI MOSFETs in a sub-micrometer regime considering that the diffusion current is dominant in the subthreshold regime of device operation. The effect of substrate charge (under source/drain regions) was included in the model [Yeh *et al.* (1995)].

Analytical models of the subthreshold current and subthreshold slope of asymmetric three-terminal (3-T) and four-terminal (4-T) DG MOSFETs were presented by Dey *et al.* [Dey *et al.* (2008)] using the evanescent mode analysis method to solve the 2D Poisson's equation in the channel region. Dey *et al.* [Dey *et al.* (2008)] used the models to study the subthreshold characteristics of a DG MOSFET with asymmetry in gate oxide thickness, gate voltage, and gate material work function.

Luan *et al.* [Luan *et al.* (2009)] presented a subthreshold current model of DMG SOI nMOSFETs with a single halo (DMGH) near the source region. The parabolic potential approximation was considered in the channel region and the current of the device was

obtained by utilizing the drift-diffusion theory [Luan *et al.* (2009)]. To obtain a fully analytical model for subthreshold current, a piecewise approximation was employed to the surface potential [Luan *et al.* (2009)].

2.4.3 Subthreshold Swing Models of SOI MOSFETs

Subthreshold swing (S) is one of the important parameters of a device as it determines the switching characteristics. It is defined as the amount of gate voltage required to change the output current by one decade. Since the subthreshold swing has vital implications in device scaling, it is important to develop analytical models to study the subthreshold swing characteristics of the SOI MOSFETs. Various theoretical models for the subthreshold swing of SOI MOSFETs have been reported in the literature [Chen *et al.* (2011), Svilicic *et al.* (2010), Kumar *et al.* (2013)].

Chen *et al.* [Chen *et al.* (2011)] developed an analytical subthreshold swing model for FD SOI MOSFETs with vertical Gaussian-like doping profile in the channel region. It was reported that the subthreshold swing decreases with increasing peak doping concentration [Chen *et al.* (2011)].

In 2010, an analytical subthreshold swing model for a Re-S/D UTB SOI MOSFET was developed by Svilicic *et al.* [Svilicic *et al.* (2010)]. Svilicic *et al.* [Svilicic *et al.* (2010)] used the developed model to analyze the dependence of subthreshold swing on various device parameters, such as channel length, channel thickness, gate material, gate oxide thickness, BOX material, and Re-S/D thickness.

Kumar *et al.* [Kumar *et al.* (2013)] presented a subthreshold swing model of the strained-Si on silicon-germanium-on-insulator (SGOI) MOSFETs for different gate lengths. It was reported that the subthreshold swing increases with the increase in strain (Ge mole fraction (x)). For an increase in the Ge mole fraction from 0 to 30%, the subthreshold swing is found to be increased by 14.73% for 30nm gate length, 2.94% for 50nm gate length and 0.8% for 70nm gate length [Kumar *et al.* (2013)]. The switching characteristics were observed to be degraded with the increased strain and decreased gate length of the device.

2.5 Summary and Conclusion

Based on the literature survey presented in above sections, Chapter 2 can be summarized under some important observations as follows:

- The FD UTB SOI MOSFET can beat the conventional MOSFETs in various aspects like ability to operate in critical environment such as high temperature [Francis *et al.* (1992)], enhanced SCEs immunity [Colinge *et al.* (2004)], high drive current capability [Chaudhry *et al.* (2004)], enhanced effective carrier mobility [Chu *et al.* (2009)], enhanced transconductance [Lim *et al.* (2009)], low parasitic capacitances and leakage currents [Markov *et al.* (2012)], and high scalability [Colinge *et al.* (2004)]. Above mentioned advantages of FD UTB SOI MOSFETs make it suitable for a number of applications like subthreshold logic operation, low-power analog circuits, RF applications, memory applications, systems-on-a-chip (SOC) applications [Trivedi *et al.* (2005)].
- In FD UTB SOI MOSFETs, the ultra-thin source and drain regions offer large series resistance and contact formation problems which lead to the poor current drive capability of the device despite having excellent short-channel characteristics [Chaudhry *et al.* (2004)]. Hence, it was observed from the literature that the large series resistance and contact formation problems could be reduced by implementing various techniques such as silicide source/drain, tungsten clad, schottky source/drain, elevated-source/drain (E-S/D), recessed-source/drain (Re-S/D) etc. [Chen *et al.* (2007), Zan *et al.* (2003), leong *et al.* (1998), Wong *et al.* (1984), Zhang *et al.* (2004)]. Among these aforementioned techniques, Re-S/D is the viable option [Zhang *et al.* (2004), Svilicic *et al.* (2009)]. Further, the Re-S/D SOI MOSFETs offer several advantages over the conventional SOI MOSFETs such as (i) higher drive current capability (ii) reduced parasitic resistances (iii) enhanced transconductance (iv) higher scalability [Zhang *et al.* (2004), Cheng *et al.* (2003), Kuchipudi and Mahmoodi (2007)].
- It was observed from the literature that CMOS performance boosters such as dualmetal-gate, high-*k* dielectric, and back-gate etc. help to improve the 'on' current

drive capability, reduce the subthreshold leakage current, and reduce the threshold voltage roll-off of FD Re-S/D SOI MOSFETs. Therefore, an investigation is required for subthreshold characteristics of Re-S/D SOI MOSFETs with CMOS performance boosters [Takagi (2007), Yang *et al.* (2006), Chaudhry *et al.* (2004), Kumar *et al.* (2004), Bangsaruntip *et al.* (2009), Kuhn (2012), Amlan *et al.* (2009)].

- Threshold voltage modeling of short-channel Re-S/D SOI MOSFETs has drawn the significant attention of the researchers because of its better scalability features over the conventional CMOS devices [Kumar *et al.* (2014)]. Various models [Kumar *et al.* (2014), Svilicic *et al.* (2009),] have been developed to estimate the SCEs by inspecting the sensitivity of threshold voltage (V_{th}) towards channel length (*L*), channel thickness (t_{Si}) , and oxide thickness (t_{ox}) . The impact of performance boosters on threshold voltage characteristics have yet to be investigated on Re-S/D FD SOI MOSFETs.
- Subthreshold current is another important characteristic of a Re-S/D SOI MOSFET as it decides the standby power dissipations and on-off current ratio which is an important parameter for characterizing the switching performance of a device. A number of subthreshold current models for SOI MOSFETs were presented based on surface potential expression [Yeh *et al.* (1995), Yan *et al.* (1992), Kumar *et al.* (2013)]. However, to the best of our knowledge, no subthreshold current models for Re-S/D SOI MOSFETs as well as Re-S/D FD SOI MOSFETs with CMOS performance boosters have been reported yet. Thus, it is very important to analytically analyze the subthreshold current for both the Re-S/D SOI MOSFETs as well as Re-S/D FD SOI MOSFETs with CMOS performance boosters.
- A long-channel SOI MOSFET is an ideal candidate for subthreshold logic circuits targeting ultra-low-power applications, since it provides an ideal value of subthreshold swing, i.e 60 mV/Decade [Yeh *et al.* (1995)]. But with reduced channel lengths, the subthreshold swing of short-channel SOI MOSFET increases

continuously, thereby degrading the switching performance of the device. Thus, it is very important to search for the new techniques for improving the switching performance of the device. Further, various subthreshold swing models for SOI MOSFETs were presented [Chen *et al.* (2011), Dey *et al.* (2008)]. However, only one model is available for Re-S/D SOI MOSFETs [Svilicic *et al.* (2010)]. Hence, there is an ample scope of work in modeling of subthreshold swing for Re-S/D FD SOI MOSFETs with CMOS performance boosters.

• Further, it was observed from the literature that recessing source and drain may deteriorate the subthreshold characteristics including threshold voltage, subthreshold current and subthreshold swing of the structure due to a strong coupling between recessed source/drain and the back surface of the channel. It may happen that the strong coupling between Re-S/D and back of the channel activates the channel at back surface before the gate voltage does the same on the front-surface. If it happens the device will show an anomalous subthreshold behavior. However, CMOS boosters such as multi-gate, high-*k* dielectric, and back-gate etc. are expected to improve the subthreshold behavior of the Re-S/D SOI MOSFETs. Therefore, modeling of subthreshold characteristics of Re-S/D SOI MOSFETs with CMOS boosters becomes requisite.

In brief, there is a lot of opportunity in the modeling and simulation of the subthreshold characteristics of Re-S/D SOI MOSFETs. The scope of the dissertation outlined in Chapter 1 is based on the observations discussed above.

Chapter 3

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

3.1 Introduction

As discussed in Chapters 1 and 2, a dual-metal-gate (DMG) MOS structure can be used for improving both the hot-carrier effects (HCEs) and drain-induced barrier lowering (DIBL) simultaneously [Kumar *et al.* (2004)]. The structure provides the benefits of high electron velocity and enhanced source side electric field resulting in increased carrier transport efficiency in the channel region [Reddy *et al.* (2005)]. The DMG structure creates a step-like channel potential profile which ensures the screening of the minimum potential point from drain voltage fluctuations. It may be noted that the DMG concept has already been implemented in the conventional silicon-on-insulator (SOI) MOSFETs and significant improvements have been observed in terms of enhanced short-channel effects (SCEs) and hot-carrier effects (HCEs) [Kumar *et al.* (2004)]. However, it can be observed from Chapter 2 that no significant work has been reported on the modeling of the subthreshold characteristics of the short-channel DMG Re-S/D SOI MOSFET. Hence, in this chapter, an attempt has been made to evaluate the impact of DMG on subthreshold characteristics of the Re-S/D UTB SOI MOSFET.

The layout of the present chapter is as follows. Section 3.2 describes the modeling of the two-dimensional (2D) surface potential function. Using the results of section 3.2,

the short-channel threshold voltage model of the device is presented in section 3.3. Analytical models for the subthreshold current and subthreshold swing are presented in sections 3.4 and 3.5 respectively. Results and discussion related to the developed models are presented in section 3.6. Finally, the present chapter is concluded in section 3.7.

3.2 Modeling of Two-Dimensional (2D) Surface Potential

A cross-sectional view of a dual-metal-gate (DMG) Re-S/D UTB SOI MOSFET which is used in our modeling and simulation is shown in Fig. 3.1. The entire Si channel is considered to be fully depleted to avoid the floating body effect. The position along the channel length is represented by the x- axis, whereas the channel depth is represented by the y- axis as shown in the Fig. 3.1. Two different metals for control and screen gates with work functions ϕ_{M1} and ϕ_{M2} , respectively, divide the entire channel into two virtual regions which are named as region I and region II. The region I is defined within $0 < x < L_1$ and $0 < y < t_{Si}$ while the region II is within $L_1 < x < L_1 + L_2$ and $0 < y < t_{Si}$; where L_1 and L_2 are lengths of the control gate and screen gate respectively, t_{Si} is silicon channel thickness. Thus, the total channel length is $L = L_1 + L_2$. The potential distributions of regions I and II are taken as $\phi_1(x, y)$ and $\phi_2(x, y)$ respectively. The symbols t_{ox} and t_{box} represent the thicknesses of the gate oxide and the BOX. The source and the drain regions penetrate into the buried oxide by a thickness t_{rsd} , which is also known as the recessed thickness. The symbol d_{box} is the length of the source/drain overlap over the buried-oxide. The channel region is assumed to be lightly doped with a doping concentration of N_a while source and drain regions are considered to be heavily doped with a doping concentration of N_d each. The source/drain to channel junction is assumed to be abrupt. The substrate doping concentration is N_{sub} . The MOSFET is biased by the gate voltage of V_{GS} , drain voltage of V_{DS} and substrate voltage as V_{sub} keeping the source voltage at $V_S = 0$. The notations and parameter values used in modeling and simulation are given in Table 3.1.



Fig. 3.1: Cross-sectional view of dual-metal-gate (DMG) Re-S/D UTB SOI MOSFET

 Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET



Fig. 3.2: The DMG Re-S/D UTB SOI MOSFET with dominant intrinsic capacitance components

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

Device Parameters (Notation)	Value and Units
Control gate work function, ϕ_{M_1}	4.8eV
Work function of screen gate, ϕ_{M_2}	4.4eV - 4.6eV
Channel doping, N_a	10^{15}cm^{-3}
Source/ drain doping, N_d	10^{20}cm^{-3}
Substrate doping, N_{sub}	10^{18}cm^{-3}
Front-channel oxide thickness, t_{ox}	1.5nm - 3nm
Buried oxide thickness, t_{box}	100nm - 300nm
Silicon thickness, t_{Si}	6nm-10nm
Substrate thickness, t_{sub}	300nm
Length of the source/drain overlap	3nm
over the buried-oxide, d_{box}	
Control gate to screen gate length ratio,	1:2, 1:1, 2:1
$L_1:L_2$	
Channel length, L	30nm - 300nm

Chapter 3

Table 3.2: Device parameter values used for modeling and simulation of a DMG Re-S/D UTB SOI MOSFET

The two-dimensional (2D) potential distribution in the channel region of a DMG Re-S/D UTB SOI MOSFET before the onset of strong inversion can be obtained by solving following Poisson's equation [Young *et al.* (1989)]

$$\frac{\partial^2 \phi_i(x, y)}{\partial x^2} + \frac{\partial^2 \phi_i(x, y)}{\partial y^2} = \frac{qN_a}{\varepsilon_{Si}} , \text{ where } i = 1, 2 \text{ for region I and II}$$
(3.1)

where, N_a , q and ε_{Si} are the doping concentration of channel region, electronic charge and the permittivity of Si film respectively. The 2D electrostatic potential distribution in regions I and II are approximated by following parabolic polynomials [Young *et al.* (1989)]

$$\phi_i(x, y) = \phi_{fi}(x) + C_{i1}(x)y + C_{i2}(x)y^2$$
(3.2)

where, i = 1 stands for region I, and i = 2 for region II, $\phi_{fi}(x)$ be the front-surface potential at SiO₂/Si interface under both metals M_1 and M_2 . The coefficients $C_{i1}(x)$ and $C_{i2}(x)$ are the functions of x only and can be determined by using following boundary conditions: The electric flux at SiO₂/Si interface is assumed to be continuous in the device. Therefore, it can be written as [Svilicic *et al.* (2009)]:

$$\left[\frac{\partial\phi_1(x,y)}{\partial y}\right]_{y=0} = \frac{C_{ox1}}{\varepsilon_{Si}} \frac{\phi_{f1}(x) - (V_{GS} - V_{FB1})}{L_1}$$
(3.3)

$$\left[\frac{\partial\phi_2(x,y)}{\partial y}\right]_{y=0} = \frac{C_{ox2}}{\varepsilon_{Si}} \frac{\phi_{f2}(x) - (V_{GS} - V_{FB2})}{L_2}$$
(3.4)

where, V_{GS} is the applied gate to source voltage. C_{ox1} , C_{ox2} are the front-gate oxide capacitances per unit length under metals M_1 and M_2 , respectively and are given by

$$C_{ox1} = \frac{\varepsilon_{ox} L_1}{t_{ox}},\tag{3.5}$$

$$C_{ox2} = \frac{\varepsilon_{ox} L_2}{t_{ox}}$$
(3.6)

where, ε_{ox} is the permittivity of gate oxide material.

Chapter 3

 V_{FB1} and V_{FB2} are the flat-band voltages for the control and screen gates and are given by

$$V_{FB1} = \phi_{M1} - \phi_{Si} \tag{3.7}$$

$$V_{FB2} = \phi_{M2} - \phi_{Si} \tag{3.8}$$

where, ϕ_{M_1} and ϕ_{M_2} are the metal work functions of the control gate and the screen gate respectively, ϕ_{Si} is the silicon work function given by

$$\phi_{Si} = \frac{\chi_{Si}}{q} + \frac{E_g}{2q} + \phi_{f,Si}$$
(3.9)

where, χ_{Si} and E_g are the electron affinity and energy band gap of the silicon respectively, $\phi_{f,Si}$ is the Fermi level potential

$$\phi_{f,Si} = V_T \ln\left(\frac{N_a}{n_i}\right) \tag{3.10}$$

where, n_i is the intrinsic carrier concentration of the silicon.

2) The electric flux at the Si / BOX interface should also be continuous in the device. Therefore, it can be written as [Svilicic *et al.* (2009)]:

Chapter 3

$$\begin{bmatrix} \frac{\partial \phi_{1}(x, y)}{\partial y} \end{bmatrix}_{y=t_{Si}} = \frac{C_{rsd1}}{\varepsilon_{Si}} \frac{V_{S} - V_{FB3} - \phi_{b1}(x)}{L_{1}} + \frac{C_{rsd2}}{\varepsilon_{Si}} \frac{V_{D} - V_{FB3} - \phi_{b1}(x)}{L_{1}} + \frac{C_{box1}}{L_{1}} \frac{V_{sub} - V_{FB4} - \phi_{b1}(x)}{L_{1}} \\ + \frac{C_{box1}}{\varepsilon_{Si}} \frac{V_{sub} - V_{FB4} - \phi_{b1}(x)}{L_{1}} \\ \begin{bmatrix} \frac{\partial \phi_{2}(x, y)}{\partial y} \end{bmatrix}_{y=t_{Si}} = \frac{C_{rsd3}}{\varepsilon_{Si}} \frac{V_{S} - V_{FB3} - \phi_{b2}(x)}{L_{2}} + \frac{C_{rsd4}}{\varepsilon_{Si}} \frac{V_{D} - V_{FB3} - \phi_{b2}(x)}{L_{2}} \\ + \frac{C_{box2}}{\varepsilon_{Si}} \frac{V_{sub} - V_{FB4} - \phi_{b2}(x)}{L_{2}} \end{aligned}$$
(3.12)

where, V_S , V_{sub} , and V_D are source voltage, substrate voltage and drain voltage respectively. $\phi_{bi}(x) = \phi_i(x, y = t_{Si})$ is the channel potential at the channel/BOX interface and named as back-surface potential. C_{rsd1} , C_{rsd2} , C_{rsd3} and C_{rsd4} are recessed source/drain- back channel capacitances per unit length as shown in Fig. 3.2 and are given by [Svilicic *et al.* (2009)]

$$C_{rsd\,1} = \begin{cases} \frac{\varepsilon_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{d_{box}}\right)} \ln\left(1 + \frac{L_{1}}{d_{box}}\right), \text{ for } L_{1} < t_{rsd} \\ \frac{\varepsilon_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{d_{box}}\right)} \ln\left(1 + \frac{t_{rsd}}{d_{box}}\right), \text{ for } L_{1} > t_{rsd} \text{ or } t_{rsd} = 0 \end{cases}$$
(3.13)

$$C_{rsd\,2} = \begin{cases} \frac{\mathcal{E}_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{(d_{box} + L_2)}\right)} \ln\left(1 + \frac{L_1}{(d_{box} + L_2)}\right), \text{ for } L_1 < t_{rsd} \\ \frac{\mathcal{E}_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{(d_{box} + L_2)}\right)} \ln\left(1 + \frac{t_{rsd}}{(d_{box} + L_2)}\right), \text{ for } L_1 > t_{rsd} \text{ or } t_{rsd} = 0 \end{cases}$$
(3.14)
Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

$$C_{rsd3} = \begin{cases} \frac{\varepsilon_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{(d_{box} + L_{1})}\right)} \ln\left(1 + \frac{L_{2}}{(d_{box} + L_{1})}\right), \text{ for } L_{2} < t_{rsd} \\ \frac{\varepsilon_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{(d_{box} + L_{1})}\right)} \ln\left(1 + \frac{t_{rsd}}{(d_{box} + L_{1})}\right), \text{ for } L_{2} > t_{rsd} \text{ or } t_{rsd} = 0 \end{cases}$$

$$C_{rsd4} = \begin{cases} \frac{\varepsilon_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{(d_{box})}\right)} \ln\left(1 + \frac{L_{2}}{d_{box}}\right), \text{ for } L_{2} < t_{rsd} \\ \frac{\varepsilon_{ox}}{\frac{\pi}{2} + \frac{\pi}{2} \sec h\left(\frac{t_{rsd}}{(d_{box})}\right)} \ln\left(1 + \frac{t_{rsd}}{d_{box}}\right), \text{ for } L_{2} < t_{rsd} \end{cases}$$

$$(3.16)$$

Chapter 3

 C_{box1} and C_{box2} are the buried oxide capacitances per unit length in the regions which are under the influence of metals M_1 and M_2 , respectively and are given by

$$C_{boxl} = \frac{\varepsilon_{ox} L_l}{t_{box}} \tag{3.17}$$

$$C_{box2} = \frac{\varepsilon_{ox} L_2}{t_{box}}$$
(3.18)

 C_{si1} and C_{si2} are the channel capacitances per unit length under metals M_1 and M_2 , respectively given as

$$C_{Si1} = \frac{\varepsilon_{Si}L_1}{t_{Si}}$$
(3.19)

$$C_{Si2} = \frac{\varepsilon_{Si}L_2}{t_{Si}}$$
(3.20)

 C_{ox1} and C_{ox2} are the front-gate oxide capacitances per unit length in the regions under the metals M_1 and M_2 , respectively given by

$$C_{boxl} = \frac{\varepsilon_{ox} L_1}{t_{box}}$$
(3.21)

$$C_{box2} = \frac{\varepsilon_{ox} L_2}{t_{box}}$$
(3.22)

 V_{FB3} and V_{FB4} are the source/drain-back channel flat-band voltage and substrate back channel flat-band voltage, respectively and are given by

$$V_{FB3} = \frac{kT}{q} ln \left(\frac{N_a N_d}{n_i^2} \right)$$
(3.23)

$$V_{FB4} = \frac{kT}{q} ln \left(\frac{N_{sub}}{N_a} \right)$$
(3.24)

where, N_a , N_d and N_{sub} are the channel, source/drain and substrate doping concentration respectively.

3) The potential at the source end is

$$\phi_1(0, y) = V_{bi} \tag{3.25}$$

where, $V_{bi} = \frac{kT}{q} ln \left(\frac{N_a N_d}{n_i^2} \right)$ is the built in potential across the source/drain and Si

junction.

4) The potential at the drain end is

$$\phi_2(L, y) = V_{bi} + V_{DS}$$
(3.26)

where, V_{DS} is the applied source/drain bias voltage.

The coefficients $C_{11}(x)$, $C_{12}(x)$, $C_{21}(x)$ and $C_{22}(x)$ of Eq. (3.2) are obtained by utilizing boundary conditions of Eqs. (3.3), (3.4), (3.11) and (3.12) in Eq. (3.2) and can be written as

$$C_{11}(x) = \frac{C_{ox1}}{C_{Si1}t_{Si}} \left(\phi_{f1}(x) - (V_{GS} - V_{FB1}) \right)$$
(3.27)

$$C_{12}(x) = \frac{C_{rsd\,1} + C_{rsd\,2}}{2C_{Si\,1}t_{Si}^{2}} (V_{DS} - 2V_{FB3} - 2\phi_{b1}(x)) + \frac{C_{box1}}{C_{Si\,1}t_{Si}^{2}} (V_{sub} - 2V_{FB4} - \phi_{b1}(x)) - \frac{C_{ox1}}{C_{Si\,1}t_{Si}^{2}} (\phi_{f1}(x) - (V_{GS} - V_{FB1}))$$
(3.28)

$$C_{21}(x) = \frac{C_{ox2}}{C_{Si2}t_{Si}} \left(\phi_{f2}(x) - (V_{GS} - V_{FB2}) \right)$$
(3.29)

Chapter 3

$$C_{22}(x) = \frac{C_{rsd 3} + C_{rsd 4}}{2C_{si 2}t_{si}^{2}} (V_{DS} - 2V_{FB3} - 2\phi_{b2}(x)) + \frac{C_{box 2}}{C_{si 2}t_{si}^{2}} (V_{sub} - 2V_{FB4} - \phi_{b2}(x)) - \frac{C_{ox 2}}{C_{si 2}t_{si}^{2}} (\phi_{f2}(x) - (V_{GS} - V_{FB2}))$$
(3.30)

Thus, the expressions for $\phi_1(x, y)$ and $\phi_2(x, y)$ are obtained by substituting the values of $C_{11}(x)$, $C_{12}(x)$, $C_{21}(x)$ and $C_{22}(x)$ into Eq. (3.2). Now, by utilizing Eq. (3.2) in the expressions of $\phi_1(x, y)$ and $\phi_2(x, y)$, potential $\phi_i(x, y)$ can be written in terms of front- and back- surface potential separately as follows

$$\phi_{1}(x, y) = \phi_{f1}(x) \left[1 + \frac{C_{ox1}}{C_{Si1}t_{Si}} y - \frac{C_{ox1}}{C_{Si1}t_{Si}^{2}} y^{2} - (2C_{Si1} + C_{ox1})z_{1}y^{2} \right] + (V_{GS} - V_{FB1}) \left[-\frac{C_{ox1}}{C_{Si1}t_{Si}} y + \frac{C_{ox1}}{2C_{Si1}t_{Si}^{2}} y^{2} - (C_{ox1})z_{1}y^{2} \right] + (V_{DS} - 2V_{FB3}) \left[\frac{C_{rsd1} + C_{rsd2}}{2C_{Si1}t_{Si}^{2}} y^{2} + (C_{rsd1} + C_{rsd2})z_{1}y^{2} \right] + (V_{sub} - V_{FB4}) \left[\frac{C_{box1}}{2C_{Si1}t_{Si}^{2}} y^{2} + (C_{box1})z_{1}y^{2} \right]$$
(3.31)

where,
$$z_1 = \left(\frac{2(C_{rsd1} + C_{rsd2}) + C_{box1}}{2C_{si1}t_{si}^2(2C_{si1} + 2(C_{rsd1} + C_{rsd2}) + C_{box1})}\right)$$
 (3.32)

$$\phi_{2}(x, y) = \phi_{f2}(x) \left[1 + \frac{C_{ox2}}{C_{Si2}t_{Si}} y - \frac{C_{ox2}}{C_{Si2}t_{Si}^{2}} y^{2} - (2C_{Si2} + C_{ox2})z_{2}y^{2} \right] + (V_{GS} - V_{FB2}) \left[-\frac{C_{ox2}}{C_{Si2}t_{Si}} y + \frac{C_{ox2}}{2C_{Si2}t_{Si}^{2}} y^{2} - (C_{ox2})z_{2}y^{2} \right] + (V_{DS} - 2V_{FB3}) \left[\frac{C_{rsd3} + C_{rsd4}}{2C_{Si2}t_{Si}^{2}} y^{2} + (C_{rsd3} + C_{rsd4})z_{2}y^{2} \right] + (V_{sub} - V_{FB4}) \left[\frac{C_{box2}}{2C_{Si2}t_{Si}^{2}} y^{2} + (C_{box2})z_{2}y^{2} \right]$$
(3.33)

where
$$z_2 = \left(\frac{2(C_{rsd3} + C_{rsd4}) + C_{box2}}{2C_{si2}t_{si}^2(2C_{si2} + 2(C_{rsd3} + C_{rsd4}) + C_{box2})}\right)$$
 (3.34)

$$\phi_{1}(x, y) = \phi_{b1}(x) \Biggl[\Biggl(\frac{2C_{si1} + 2(C_{rsd1} + C_{rsd2}) + C_{box1}}{2C_{si1} + C_{ox1}} \Biggr) z_{3} - \Biggl(\frac{C_{rsd1} + C_{rsd2}}{C_{si1} t_{si}^{2}} + \frac{C_{box1}}{2C_{si1} t_{si}^{2}} \Biggr) y^{2} \Biggr]$$

$$+ (V_{GS} - V_{FB1}) \Biggl[-\frac{C_{ox1}}{C_{si1} t_{si}} y + \frac{C_{ox1}}{C_{si1} t_{si}^{2}} y^{2} + \Biggl(\frac{C_{ox1}}{2C_{si1} + C_{ox1}} \Biggr) z_{3} \Biggr]$$

$$+ (V_{DS} - 2V_{FB3}) \Biggl[\frac{C_{rsd1} + C_{rsd2}}{C_{si1} t_{si}^{2}} y^{2} - \Biggl(\frac{C_{rsd1} + C_{rsd2}}{2C_{si1} + C_{ox1}} \Biggr) z_{3} \Biggr]$$

$$+ (V_{sub} - V_{FB4}) \Biggl[\frac{C_{box1}}{C_{si1} t_{si}^{2}} y^{2} - \Biggl(\frac{C_{box1}}{2C_{si1} + C_{ox1}} \Biggr) z_{3} \Biggr]$$

$$+ (V_{sub} - V_{FB4}) \Biggl[\frac{C_{box1}}{C_{si1} t_{si}^{2}} y^{2} - \Biggl(\frac{C_{box1}}{2C_{si1} + C_{ox1}} \Biggr) z_{3} \Biggr]$$

where,
$$z_3 = \left(1 + \frac{C_{\alpha x1}}{C_{si1}t_{si}}y - \frac{C_{\alpha x1}}{2C_{si1}t_{si}^2}y^2\right)$$
 (3.36)

$$\begin{split} \phi_{2}(x,y) &= \phi_{b2}(x) \Biggl[\Biggl(\frac{2C_{Si2} + 2(C_{rsd3} + C_{rsd4}) + C_{box2}}{2C_{Si2} + C_{ox2}} \Biggr) z_{4} - \\ & \left(\frac{C_{rsd3} + C_{rsd4}}{C_{Si2} t_{Si}^{2}} + \frac{C_{box2}}{2C_{Si2} t_{Si}^{2}} \Biggr) y^{2} \Biggr] \\ &+ (V_{GS} - V_{FB2}) \Biggl[- \frac{C_{ox2}}{C_{Si2} t_{Si}} y + \frac{C_{ox2}}{C_{Si2} t_{Si}^{2}} y^{2} + \Biggl(\frac{C_{ox2}}{2C_{Si2} + C_{ox2}} \Biggr) z_{4} \Biggr] \end{split}$$
(3.37)
$$&+ (V_{DS} - 2V_{FB3}) \Biggl[\frac{C_{rsd3} + C_{rsd4}}{C_{Si2} t_{Si}^{2}} y^{2} - \Biggl(\frac{C_{rsd3} + C_{rsd4}}{2C_{Si2} + C_{ox2}} \Biggr) z_{4} \Biggr] \\ &+ (V_{sub} - V_{FB4}) \Biggl[\frac{C_{box2}}{C_{Si2} t_{Si}^{2}} y^{2} - \Biggl(\frac{C_{box2}}{2C_{Si2} + C_{ox2}} \Biggr) z_{4} \Biggr] \end{split}$$

where,
$$z_4 = \left(1 + \frac{C_{ox2}}{C_{Si2}t_{Si}}y - \frac{C_{ox2}}{2C_{Si2}t_{Si}^2}y^2\right)$$
 (3.38)

Further, by substituting $\phi_1(x, y)$ and $\phi_2(x, y)$ in Eq. (3.1) and putting y = 0 and $y = t_{Si}$ respectively, the following differential equations of front- and back-surface potential are obtained

$$\frac{d^2 \phi_{fi}(y)}{dx^2} - \alpha_{fi} \phi_{fi}(y) = \beta_{fi}, \ i = 1,2$$
(3.39)

$$\frac{d^2 \phi_{bi}(y)}{dx^2} - \alpha_{bi} \phi_{bi}(y) = \beta_{bi} , i = 1,2$$
(3.40)

where,

$$\alpha_{f1} = 2 \frac{1 + \frac{C_{ox1} (C_{Si1} + C_{rsd1} + C_{rsd2} + C_{box1})}{C_{Si1} (C_{rsd1} + C_{rsd2} + C_{box1})}}{t_{Si}^{2} \left(1 + \frac{2C_{Si1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right)}$$
(3.41)

$$\alpha_{f2} = 2 \frac{1 + \frac{C_{ox2} (C_{si2} + C_{rsd3} + C_{rsd4} + C_{box2})}{C_{si2} (C_{rsd3} + C_{rsd4} + C_{box2})}}{t_{si}^{2} \left(1 + \frac{2C_{si2}}{C_{rsd3} + C_{rsd4} + C_{box2}}\right)}$$
(3.42)

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

$$\beta_{f1} = \left[\frac{qN_a}{\varepsilon_{Si}} - (V_{GS} - V_{FB1}) \frac{2\frac{C_{ox1}(C_{Si1} + C_{rsd1} + C_{rsd2} + C_{box1})}{C_{Si1}(C_{rsd1} + C_{rsd2} + C_{box1})}}{t_{Si}^2 \left(1 + \frac{2C_{Si1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right)} - (V_{DS} - 2V_{FB3}) \frac{\frac{C_{rsd1} + C_{rsd2} + C_{box1}}{C_{rsd1} + C_{rsd2} + C_{box1}}}{t_{Si}^2 \left(1 + \frac{2C_{Si1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right)}\right]$$

$$-(V_{sub} - V_{FB4}) \frac{\frac{2C_{box1}}{C_{rsd1} + C_{rsd2} + C_{box1}}}{t_{Si}^2 \left(1 + \frac{2S_{Si1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right)}\right]$$

$$-(V_{sub} - V_{FB4}) \frac{\frac{2C_{ox2}(C_{Si2} + C_{rsd3} + C_{rsd4} + C_{box2})}{c_{rsd1} + C_{rsd2} + C_{box1}}}$$

$$\beta_{f2} = \left[\frac{qN_a}{\varepsilon_{Si}} - (V_{GS} - V_{FB2}) \frac{2 \frac{\varepsilon_{ox2}(\varepsilon_{Si2} + \varepsilon_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2})}{C_{Si2}(C_{rsd3} + C_{rsd4} + \varepsilon_{box2})} - (V_{DS} - 2V_{FB3}) \frac{2C_{si2}(C_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2})}{C_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2}} - (V_{Sib} - 2V_{FB3}) \frac{\frac{C_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2}}{C_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2}}}{t_{Si}^2 \left(1 + \frac{2C_{Si2}}{C_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2}}\right)} - (V_{sub} - V_{FB4}) \frac{\frac{2C_{box2}}{\varepsilon_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2}}}{t_{Si}^2 \left(1 + \frac{2C_{Si2}}{\varepsilon_{rsd3} + \varepsilon_{rsd4} + \varepsilon_{box2}}\right)}\right)}$$

$$(3.44)$$

$$\alpha_{b1} = 2 \frac{1 + \frac{C_{ox1} (C_{Si1} + C_{rsd_1} + C_{rsd_2} + C_{box1})}{C_{Si1} (C_{rsd_1} + C_{rsd_2} + C_{box1})}}{t_{Si}^{2} \left(1 + \frac{2C_{Si1}}{C_{ox1}}\right)}$$
(3.45)

$$\alpha_{b2} = 2 \frac{1 + \frac{C_{ox2} (C_{Si2} + C_{rsd3} + C_{rsd4} + C_{box2})}{C_{Si2} (C_{rsd3} + C_{rsd4} + C_{box2})}}{t_{Si}^{2} \left(1 + \frac{2C_{Si2}}{C_{ox2}}\right)}$$
(3.46)

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

$$\beta_{b1} = \left[\frac{qN_a}{2\varepsilon_{Si}} - \frac{2(V_{GS} - V_{FB1})}{t_{Si}^2 \left(1 + 2\frac{C_{Si1}}{C_{ox1}}\right)} - 2(V_{DS} - 2V_{FB3})\frac{\left(\frac{C_{rsd1}}{C_{ox1}} + \frac{C_{rsd2}}{C_{Si1}}\right)}{t_{Si}^2 \left(1 + 2\frac{C_{Si1}}{C_{ox1}}\right)} - 2(V_{sub} - V_{FB4})\frac{\left(\frac{C_{box1}}{C_{ox1}} + \frac{C_{box1}}{C_{Si1}}\right)}{t_{Si}^2 \left(1 + 2\frac{C_{Si1}}{C_{ox1}}\right)} \right]$$
(3.47)

$$\beta_{b2} = \left[\frac{qN_a}{2\varepsilon_{Si}} - \frac{2(V_{GS} - V_{FB2})}{t_{Si}^2 \left(1 + 2\frac{C_{Si2}}{C_{ox2}}\right)} - 2(V_{DS} - 2V_{FB3})\frac{\left(\frac{C_{rsd3}}{C_{ox2}} + \frac{C_{rsd4}}{C_{Si2}}\right)}{t_{Si}^2 \left(1 + 2\frac{C_{Si2}}{C_{ox2}}\right)} - 2(V_{sub} - V_{FB4})\frac{\left(\frac{C_{box2}}{C_{ox2}} + \frac{C_{box2}}{C_{Si2}}\right)}{t_{Si}^2 \left(1 + 2\frac{C_{Si2}}{C_{ox2}}\right)}$$
(3.48)

Now, the following equations of front- and back-surface potentials, $\phi_{fi,bi}(x)$, are obtained by solving the differential Eqs. (3.39) and (3.40) with the help of boundary conditions of Eqs. (3.25)-(3.26)

$$\phi_{f1,b1}(x) = \frac{\psi_{fd1,bd1} \sinh(\lambda_{f1,b1} x) - \psi_{fs1,bs1} \sinh(\lambda_{f1,b1} (x - L_1))}{\sinh(\lambda_{f1,b1} L_1)} - \sigma_{f1,b1}$$
(3.49)

$$\phi_{f2,b2}(x) = \frac{\psi_{fd2,bd2} \sinh(\lambda_{f2,b2}(x-L_1)) - \psi_{fs2,bs2} \sinh(\lambda_{f2,b2}(x-L))}{\sinh(\lambda_{f_{2,b2}}L_2)} - \sigma_{f2,b2}$$
(3.50)

where,

$$\lambda_{f_i} = \sqrt{\alpha_{f_i}} \tag{3.51}$$

$$\lambda_{b_i} = \sqrt{\alpha_{bi}} \tag{3.52}$$

$$\sigma_{fi} = \frac{\beta_{fi}}{\alpha_{fi}} \tag{3.53}$$

$\sigma_{bi} = \frac{\beta_{bi}}{\beta_{bi}},$	(3.54)
α_{bi}	

$$\psi_{fs1} = V_{bi} + \sigma_{f1} \tag{3.55}$$

$$\psi_{fd1} = V_{fp} + \sigma_{f1} \tag{3.56}$$

$$\psi_{fs2} = V_{fp} + \sigma_{f2} \tag{3.57}$$

$$\psi_{fd2} = V_{bi} + V_{DS} + \sigma_{f2} \tag{3.58}$$

$$\psi_{bs1} = V_{bi} + \sigma_{b1} \tag{3.59}$$

$$\psi_{bd1} = V_{bp} + \sigma_{b1} \tag{3.60}$$

$$\psi_{bs2} = V_{bp} + \sigma_{b2} \tag{3.61}$$

$$\psi_{bd2} = V_{bi} + V_{DS} + \sigma_{b2} \tag{3.62}$$

$$V_{fp} = \frac{\psi_{f,d2} \operatorname{cos} \operatorname{ech}(\lambda_f L_2) + \psi_{f,s1} \operatorname{cos} \operatorname{ech}(\lambda_f L_1) - \sigma_{f1} \operatorname{coth}(\lambda_f L_2) - \sigma_{f2} \operatorname{coth}(\lambda_f L_2)}{\operatorname{coth}(\lambda_f L_1) + \operatorname{coth}(\lambda_f L_2)}$$
(3.63)

$$V_{bp} = \frac{\psi_{b,d2} \cos ech(\lambda_b L_2) + \psi_{b,s1} \cos ech(\lambda_b L_1) - \sigma_{b1} \coth(\lambda_b L_1) - \sigma_{b2} \coth(\lambda_b L_2)}{\coth(\lambda_b L_1) + \coth(\lambda_b L_2)}$$
(3.64)

where, $\lambda_{f_i} = \lambda_f$ and $\lambda_{b_i} = \lambda_b$ are the characteristic lengths associated with the frontsurface potential and back-surface potential, respectively.

The position (x_{min}) of the front- and back-surface minimum potential value (virtual

cathode) lies under the control gate and is estimated by solving $\frac{d\phi_{f1,b1}(x)}{dx}\Big|_{(x_{min})} = 0$

which yields

$$(x_{\min})_{f_{1,b_{1}}} = \frac{1}{2\lambda_{f,b}} ln \left(\frac{b_{f_{1,b_{1}}}}{a_{f_{1,b_{1}}}}\right)$$
(3.65)

Now the front- and back- channel minimum surface potential, $\phi_{f_{1min,b_{1min}}}$, under the control gate region can be obtained by putting Eq. (3.65) into Eq. (3.49) and its value is found to be

$$\phi_{f1min,b1min} = 2\sqrt{a_{f1,b1}b_{f1,b1}} - \sigma_{f1,b1}, \tag{3.66}$$

where,

Chapter 3

$$a_{f1,b1} = \frac{-1}{2\sinh(\lambda_{f,b}L_{1})} \left[\psi_{fs1,bs1} e^{-\lambda_{f1,b1}L_{1}} - \psi_{fd1,bd1} \right]$$
(3.67)

$$b_{f1,b1} = \frac{-1}{2\sinh(\lambda_{f,b}L_{1})} \left[-e^{\lambda_{f1,b1}L_{1}} \psi_{fs1,bs1} + \psi_{fd1,bd1} \right]$$
(3.68)

Here, it is worth mentioning that $\phi_{f_{1min}}$ and $\phi_{b_{1min}}$ are two important parameters which determine the source-channel barrier heights at front and back surfaces of the channel respectively. The condition $\phi_{f_{1min}} > \phi_{b_{1min}}$ indicates that the source-channel barrier height at the front-surface of the channel is lower compared to the same at the back channel. And, as a result the inversion layer is formed at the front-surface of the channel when gate voltage is raised up to the threshold voltage level. Similarly, the condition $\phi_{f_{1min}} < \phi_{b_{1min}}$ facilitates the inversion layer to be formed at the back surface of the channel under suitable gate bias condition. In a conventional SOI MOSFET, the condition $\phi_{f_{1min}} > \phi_{b_{1min}}$ holds generally true because the back surface of the channel remains far away from the front-gate. Further, the weak source/drain-back channel coupling also does not help much. However, in case of Re-S/D SOI MOSFETs, both conditions (i.e $\phi_{f_{1min}} > \phi_{b_{1min}}$ or $\phi_{f_{1min}} < \phi_{b_{1min}}$) may hold true depending upon the device parameters [Kumar et al. (2014)]. For example, if the channel region is highly doped or the device channel length is considerably large, the Re-S/D SOI MOSFET behaves as a conventional SOI MOSFETs [Kumar et al. (2014)]. However, in case of a short-channel undoped/lightly doped Re-S/D SOI MOSFET, a strong electrostatics coupling between source/drain and back surface of the channel builds up, and the condition $\phi_{f_{1min}} < \phi_{b_{1min}}$ is found to be exist. This strong coupling may lead the back surface of the channel to the inversion.

Now, we derive the following equation of the so-called virtual cathode potential $\phi_{vc}(y)$ [Chen *et al.* (2003)] ($\phi_{vc}(y) = \phi_1(x, y)|_{x=x_{min}}$, where $x = x_{min}$ is the position where minima of front-/back-surface potential exists). The obtained virtual cathode potential $\phi_{vc}(y)$ which is a function of the minimum value of the front-surface potential is

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

$$\phi_{vc}(y) = \phi_{f1min} \left[1 + \frac{C_{ox1}}{C_{Si1}t_{Si}} y - \frac{C_{ox1}}{C_{Si1}t_{Si}^{2}} y^{2} - (2C_{Si1} + C_{ox1})z_{1}y^{2} \right] + (V_{GS} - V_{FB1}) \left[-\frac{C_{ox1}}{C_{Si1}t_{Si}} y + \frac{C_{ox1}}{2C_{Si1}t_{Si}^{2}} y^{2} - (C_{ox1})z_{1}y^{2} \right] + (V_{DS} - 2V_{FB3}) \left[\frac{C_{rsd1} + C_{rsd2}}{2C_{Si1}t_{Si}^{2}} y^{2} + (C_{rsd1} + C_{rsd2})z_{1}y^{2} \right] + (V_{sub} - V_{FB4}) \left[\frac{C_{box1}}{2C_{Si1}t_{Si}^{2}} y^{2} + (C_{box1})z_{1}y^{2} \right]$$
(3.69)

Similarly, another equation of the so-called virtual cathode potential which is a function of the minimum value of the back surface potential can also be obtained

$$\begin{split} \phi_{vc}(y) &= \phi_{b1min} \left[\left(\frac{2C_{si1} + 2(C_{rsd1} + C_{rsd2}) + C_{box1}}{2C_{si1} + C_{ox1}} \right) z_{3} - \left(\frac{C_{rsd1} + C_{rsd2}}{C_{si1} t_{si}^{2}} + \frac{C_{box1}}{2C_{si1} t_{si}^{2}} \right) y^{2} \right] \\ &+ (V_{GS} - V_{FB1}) \left[- \frac{C_{ox1}}{C_{si1} t_{si}} y + \frac{C_{ox1}}{C_{si1} t_{si}^{2}} y^{2} + \left(\frac{C_{ox1}}{2C_{si1} + C_{ox1}} \right) z_{3} \right] \\ &+ (V_{DS} - 2V_{FB3}) \left[\frac{C_{rsd1} + C_{rsd2}}{C_{si1} t_{si}^{2}} y^{2} - \left(\frac{C_{rsd1} + C_{rsd2}}{2C_{si1} + C_{ox1}} \right) z_{3} \right] \\ &+ (V_{sub} - V_{FB4}) \left[\frac{C_{box1}}{C_{si1} t_{si}^{2}} y^{2} - \left(\frac{C_{box1}}{2C_{si1} + C_{ox1}} \right) z_{3} \right] \end{split}$$
(3.70)

Here, it should be noted that either of Eqs. (3.69) or (3.70) can be utilized for the formulation of the subthreshold charge carrier density at virtual cathode position.

3.3 Threshold Voltage Formulation

Chapter 3

The threshold voltage V_{th} of a MOSFET is defined as that value of the gate voltage V_{GS} at which a number of minority carriers at the channel surface equals the majority carrier concentration in the bulk. In other words, in a conventional MOSFET, the threshold voltage is taken to be that value of the gate-source voltage at which the virtual cathode potential becomes equal to the twice of the Fermi potential (i.e. $\phi_{f1min,b1min} = 2\phi_{f,Si}$

where, $\phi_{f_{1min,b_{1min}}}$ is minimum surface potential) [Svilicic *et al.* (2009)]. However, this definition of the threshold voltage does not hold completely true for a lightly doped or intrinsic SOI MOSFET owing to the volume inversion in the whole channel region, as discussed in Chapter 2. Because of the early inversion of the charge carriers in an intrinsic/lightly doped SOI film, the threshold voltage definition is modified as discussed in section 2.4.1 of Chapter 2. In this chapter, the modified threshold voltage definition of [Chen *et al.* (2003)] is used for threshold voltage formulation. Further, any of the front-channel or back channel may be inverted at first and form a conducting channel between source and drain, threshold voltage associated with front-surface and back-surface of the channel regions have been formulated separately.

3.3.1 Front Channel Threshold Voltage Formulation

The front-channel threshold voltage (V_{thf}) is considered to be the value of the gate voltage at which the minimum of front-channel surface potential $\phi_{f_{1min}}$ is found to be $2\phi^*_{f,Si}$,

i.e.
$$\phi_{f1min}\Big|_{V_G = V_{thf}} = 2\phi^*_{f,Si} = \begin{cases} 2\phi_{f,Si} & \text{for}, N_a > n_T \\ \phi_{f,Si} + \frac{kT}{q} ln\left(\frac{n_T}{n_i}\right) & \text{for}, N_a < n_T \end{cases}$$

$$(3.71)$$

where, q is the electron charge, N_a is the channel doping concentration, n_i is the intrinsic carrier concentration, k is the Boltzmann's constant, and T is the temperature, $\phi_{f,Si}$ is the Fermi potential, n_T is a critical concentration of electron in the channel to turn on the device [Lee *et al.* (1989)]. It should be noted that n_T depends on device parameters like channel thickness and channel length; it has been determined following the method of Chen [Chen (2003), Svilicic *et al.* (2010)].

Thus, solving $\phi_{f_{1min}} = 2\sqrt{a_{f_1}b_{f_1}} - \sigma_{f_1}\Big|_{V_{cs}=V_{l_{hf}}} = 2\phi^*_{f,Si}$ gives the following expression for front-channel threshold voltage, V_{thf}

$$V_{thf} = \frac{-q_f + \sqrt{q_f^2 - 4p_f r_f}}{2p_f}$$
(3.72)

Chapter 3

where, $p_f = v_{f1}v_{f2} - n_f^2$ (3.73)

$$q_f = u_{f1}v_{f2} + u_{f2}v_{f1} - 2n_f l_f \tag{3.74}$$

$$r_f = u_{f1}u_{f2} - l_f^{\ 2} \tag{3.75}$$

$$u_{f1} = \frac{\left(V_{fp11} + m_{f1}\right) - \left(V_{bi} + m_{f1}\right)exp\left(-\lambda_{f}L_{1}\right)}{sinh(\lambda_{f}L_{1})}$$
(3.76)

$$v_{f1} = \frac{V_{fp12} + (1 - exp(-\lambda_f L_1))n_f}{\sinh(\lambda_f L_1)}$$
(3.77)

$$u_{f2} = \frac{(V_{bi} + m_{f1})exp(\lambda_f L_1) - (V_{fp11} + m_{f1})}{\sinh(\lambda_f L_1)}$$
(3.78)

$$v_{f2} = \frac{(exp(\lambda_f L_1) - 1)n_f - V_{fp12}}{sinh(\lambda_f L_1)}$$
(3.79)

$$l_f = 2\phi_f + m_{1f}$$
(3.80)

$$V_{fp_{11}} = \frac{\left[(V_{bi} + V_{DS}) cos ech(\lambda_f L_2) + (V_{bi}) cos ech(\lambda_f L_1) + m_{b1}(cos ech(\lambda_f L_2) - coth(\lambda_f L_2)) + m_{f1}(cos ech(\lambda_f L_1) - coth(\lambda_f L_1)) \right]}{coth(\lambda_f L_1) + coth(\lambda_f L_2)}$$
(3.81)

$$V_{fp_{12}} = \frac{n_f \left[\cos ech(\lambda_f L_2) - \coth(\lambda_f L_2) + \cos ech(\lambda_f L_1) - \coth(\lambda_f L_1) \right]}{\coth(\lambda_f L_1) + \coth(\lambda_f L_2)}$$
(3.82)

$$m_{1f} = \frac{\left[\frac{qN_{a}}{\varepsilon_{Si}} \left(\alpha_{f2} - \beta_{f1} \frac{C_{ox1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right) - k \left(\alpha_{f2} + \beta_{f1} \frac{C_{ox1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right)\right]}{\alpha_{f1} \alpha_{f2} - \beta_{f1} \beta_{f2}}$$
(3.83)

$$n_{f} = \frac{\frac{C_{ox1}C_{Si1}}{t_{Si}^{2}} \left(\alpha_{f2} - \beta_{f1} \frac{C_{ox1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right)}{\alpha_{f1}\alpha_{f2} - \beta_{f1}\beta_{f2}}$$
(3.84)

3.3.2 Back Channel Threshold Voltage Formulation

Similarly, solving $\phi_{b1min} = 2\sqrt{a_b b_b} - \sigma_{b1}\Big|_{V_{GS} = Vt_{hb}} = 2\phi^*_{f,St}$ gives the following expression for

back-channel threshold voltage, V_{thb}

$$V_{thb} = \frac{-q_b + \sqrt{q_b^2 - 4p_b r_b}}{2p_b}$$
(3.85)

where, $p_b = v_{b1}v_{b2} - n_b^2$ (3.86)

$$q_b = u_{b1}v_{b2} + u_{b2}v_{b1} - 2n_b l_b \tag{3.87}$$

$$r_b = u_{b1}u_{b2} - l_b^2 \tag{3.88}$$

$$u_{b1} = \frac{\left(V_{bp11} + m_{b1}\right) - \left(V_{bi} + m_{b1}\right)exp\left(-\lambda_{b}L_{1}\right)}{sinh(\lambda_{b}L_{1})}$$
(3.89)

$$v_{b1} = \frac{V_{bp12} + (1 - exp(-\lambda_c L_1))n_b}{sinh(\lambda_b L_1)}$$
(3.90)

$$u_{b2} = \frac{(V_{bi} + m_{b1})exp(\lambda_b L_1) - (V_{bp11} + m_{b1})}{sinh(\lambda_b L_1)}$$
(3.91)

$$v_{b2} = \frac{(exp(\lambda_b L_1) - 1)n_b - V_{bp12}}{sinh(\lambda_b L_1)}$$
(3.92)

$$l_b = 2\phi_f + m_{b1} \tag{3.93}$$

$$\begin{bmatrix} (V_{bi} + V_{DS}) cos ech(\lambda_b L_2) + (V_{bi}) cos ech(\lambda_b L_1) \\ + m_{b1} (cos ech(\lambda_b L_2) - coth(\lambda_b L_2)) m_{f1} (cos ech(\lambda_b L_1) - coth(\lambda_b L_1)) \end{bmatrix}$$

$$V_{bp_{11}} = \frac{coth(\lambda_f L_1) + coth(\lambda_f L_2)}{coth(\lambda_f L_1) + coth(\lambda_f L_2)}$$
(3.94)

$$V_{bp_{12}} = \frac{n_b \left[\cos ech(\lambda_b L_2) - \coth(\lambda_b L_2) + \cos ech(\lambda_b L_1) - \coth(\lambda_b L_1) \right]}{\coth(\lambda_b L_1) + \coth(\lambda_b L_2)}$$
(3.95)

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

$$m_{b1} = \frac{\left[\frac{qN_{a}}{\varepsilon_{Si}}\left(\alpha_{b1}\frac{C_{ox1}}{C_{rsd1} + C_{rsd2} + C_{box1}} - \beta_{b2}\right) + k\left(\alpha_{b1}\frac{C_{ox1}}{C_{rsd1} + C_{rsd2} + C_{box1}} + \beta_{b2}\right)\right]}{\alpha_{b1}\alpha_{b2} - \beta_{b1}\beta_{b2}}$$
(3.96)

Chapter 3

$$n_{b} = \frac{\frac{C_{ox1}C_{Si1}}{t_{Si}^{2}} \left(\alpha_{b1} \frac{C_{ox1}}{C_{rsd1} + C_{rsd2} + C_{box1}} - \beta_{b2} \right)}{\alpha_{b1}\alpha_{b2} - \beta_{b1}\beta_{b2}}$$
(3.97)

As mentioned above, DMG FD Re-S/D SOI MOSFET may have two different threshold voltages associated with the front-surface and back-surface of the channel region, respectively. Hence, the threshold voltage V_{th} of a DMG Re-S/D SOI MOSFET could be determined either by front-surface threshold voltage (V_{thf}) or by the back surface threshold voltage (V_{thf}) or by the back surface threshold voltage (V_{thb}) depending on the front- and back- minimum surface potential. In this way, the threshold voltage of DMG UTB Re-S/D SOI MOSFET is defined as

$$V_{th} = \begin{cases} V_{thf}, & for \quad \phi_{f,1min} > \phi_{b,1min} \\ V_{thb}, & for \quad \phi_{f,1min} < \phi_{b1,min} \end{cases}$$
(3.98)

3.3.3 Threshold Voltage Modifications due to Quantum Mechanical Effects (QMEs)

The model's accuracy could be improved for ultra-thin body structure by adding following quantum effects induced correction term ΔV_{th} in the threshold voltage model of Eq. (3.98) [Dort *et al.* (1992)].

$$\Delta V_{th} = \Delta E_0 \left(1 + \frac{\varepsilon_{Si} t_{ox}}{\varepsilon_k t_{Si} + \varepsilon_{Si} t_{box}} \right)$$
(3.99)

where, ΔE_0 is the difference between the first discrete energy level and the lowest conduction band energy level which is given as [Dort *et al.* (1992)].

$$\Delta E_0 = \left(\frac{h^2}{2m^*}\right)^{1/3} \left[\frac{9}{8}\pi qE\right]^{2/3}$$
(3.100)

where, h is the planks constant, m^* is effective mass of the electron, E is the normal electric field due to the charge in the depletion layer.

3.4 Subthreshold Current Formulation

The diffusion dominant subthreshold current in weak inversion is assumed to be proportional to the carrier concentration at the virtual cathode, $n_{\min}(y)$, and hence, can be defined as [Kumar *et al.* (2013), Dey at al. (2008)]

$$I_{sub} = \int_{0}^{t_{si}} J_n(y) dy$$
(3.101)

where,
$$J_n(y) = \frac{qD_n n_{min}(y)}{L_e} \left(1 - exp\left(-\frac{V_{DS}}{V_T}\right)\right)$$
 (3.102)

 D_n is the coefficient of diffusion, L_e is the effective channel length, V_T is the thermal voltage. The electron density at the virtual cathode, $n_{min}(y)$ can be expressed as [Dey at al. (2008)]

$$n_{min}(y) = \frac{n_i^2}{N_a} exp\left(\frac{\phi_{vc}(y)}{V_T}\right)$$
(3.103)

where n_i is the intrinsic carrier concentration.

The effective channel length can be expressed as [Dey at al. (2008)]

$$L_e = L - (L_s + L_d) + 2L_D \tag{3.104}$$

where, L_s and L_d represent the source and drain depletion width, L_D be the Debye length, all of which can be formulated as[Yeh *et al.* (1995)]

$$L_{s} = \frac{2(V_{bi} - \phi_{vc}(y_{m}))}{\frac{d\phi(x, y)}{dx}\Big|_{x=0}}$$
(3.105)

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET

$$L_{d} = \frac{2(V_{bi} + V_{ds} - \phi_{vc}(y_{m}))}{\frac{d\phi(x, y)}{dx}}$$
(3.106)

Chapter 3

$$L_D = \sqrt{\frac{\varepsilon_{si} V_T}{q N_a}} \tag{3.107}$$

where, $\phi_{vc}(y_m) = \phi_{vc}(y)$ is the value of the minimum surface potential function along the channel thickness at $y = y_m$, where y_m can be obtained by solving

$$\frac{\partial \phi_{vc}(y)}{\partial y} = 0 \text{ and } \phi_{vc}(y_m) = \phi_{vc}(y)\Big|_{y=y_m}$$
(3.108)

Equation (3.101) is a complex equation and cannot be solved analytically. However, an approximate solution of Eq. (3.101) can be obtained by invoking a piecewise linear approximation of the virtual cathode potential equation. For this purpose the channel region may be divided into two parts; region 1 ($0 \le y \le y_m$) contributing current I_F , and region 2 ($y_m \le y \le t_{Si}$) contributing another current I_B . Further, the virtual cathode potential variation is approximated by two straight lines in regions1 and 2. [Yeh *et al.* (1995)]. Finally, the subthreshold current can now be written as

$$I_{sub} = I_F + I_B$$

$$= K \left[\int_{0}^{y_m} \exp\left(\frac{\phi_{vc}(y)}{V_T}\right) dy + \int_{y_m}^{t_{SI}} \exp\left(\frac{\phi_{vc}(y)}{V_T}\right) dy \right]$$

$$\text{where } K = \frac{q D_n V_T n_i^2}{L_e N_a} \left(1 - \exp\left(-\frac{V_{DS}}{V_T}\right) \right)$$

$$(3.110)$$

Assuming the linear variation of $\phi_{vc}(y)$ in y direction, Eq. (3.109) can be solved as [Dey at al. (2008)]

$$I_{sub} = KV_T \left(\frac{I_f}{E_f} + \frac{I_b}{E_b}\right)$$
(3.111)

where,

$$I_f = exp\left(\frac{\phi_{vc}(y=0)}{V_T}\right) - exp\left(\frac{\phi_{vc}(y=y_m)}{V_T}\right)$$
(3.112)

$$I_{b} = exp\left(\frac{\phi_{vc}(y=y_{m})}{V_{T}}\right) - exp\left(\frac{\phi_{vc}(y=t_{Si})}{V_{T}}\right)$$
(3.113)

$$E_{f} = (\phi_{vc}(y=0) - \phi_{vc}(y=y_{m})) / y_{m}$$
(3.114)

$$E_{b} = \left(\phi_{vc}(y = y_{m}) - \phi_{vc}(y = t_{Si})\right) / (t_{Si} - y_{m})$$
(3.115)

Eqs. (3.114 & 3.115) are the electric fields associated with the front- and back- surfaces of the channel. Eq. (3.111) is the desired expression for subthreshold current derived for DMG Re-S/D FD UTB SOI MOSFET.

3.5 Subthreshold Swing Formulation

The switching property of a MOS device is governed by the subthreshold swing characteristics in the weak inversion region. It is defined as the inverse slope of the $log(I_D) vsV_{GS}$ characteristics in the subthreshold region. The typical value of subthreshold swing is 60mV/dec (at room temperature).

The subthreshold swing of a Re-S/D SOI MOSFETs can be expressed as [Svilicic *et al.* (2010)]

$$S = \begin{cases} V_T(ln10) \times \left(\frac{d\phi_{f1min}}{dV_{GS}}\right)^{-1} for \phi_{f1min} > \phi_{b1min} \\ V_T(ln10) \times \left(\frac{d\phi_{b1min}}{dV_{GS}}\right)^{-1} for \phi_{b1min} > \phi_{f1min} \end{cases}$$
(3.116)

Utilising Eq. (3.65) in Eq. (3.116) and after some mathematical calculations, we obtain the following closed form expression of the subthreshold swing of the DMG Re-S/D UTB SOI MOSFET

$$S = \begin{cases} V_{T}(\ln 10) \times \left(\frac{-1}{2\sqrt{a_{f1}b_{f1}} \sinh(\lambda_{f}L_{1})} \left| \frac{\frac{K_{1}}{\alpha_{f1}} \left(e^{-\lambda_{f}L_{1}} - 1\right) \left(b_{f} + a_{f}e^{\lambda_{f}L_{1}}\right)}{\sinh(\lambda_{f}L_{1}} \left(1 - \cosh(\lambda_{f}L_{1})\right) \left(1 - \cosh(\lambda_{f}L_{2})\right)} \right| - \frac{K_{1}}{\alpha_{f1}} \right) \\ + \frac{K_{2}}{\alpha_{f2}} \frac{\psi_{f,s1} \left(1 - \cosh(\lambda_{f}L_{1})\right) \left(1 - \cosh(\lambda_{f}L_{2})\right)}{\sinh(\lambda_{f}L_{1}} \right)}{\sinh(\lambda_{f}L_{1}} \right) \\ + \frac{K_{1}}{\alpha_{f1}} \frac{\psi_{f,s1} \sinh(\lambda_{f}L_{1}) \left(1 - \cosh(\lambda_{f}L_{1})\right)^{2}}{\sinh(\lambda_{f}L_{1}) \sinh(\lambda_{f}L_{1})} \right] \\ + \frac{K_{3}}{\alpha_{b1}} \frac{\psi_{b,s1} \left(1 - \cosh(\lambda_{b}L_{1})\right) \left(1 - \cosh(\lambda_{b}L_{2})\right)}{\sinh(\lambda_{b}L_{1}} \right)}{\int dt + \frac{K_{3}}{\alpha_{b1}} \frac{\psi_{b,s1} \left(1 - \cosh(\lambda_{b}L_{1})\right) \left(1 - \cosh(\lambda_{b}L_{2})\right)}{\sinh(\lambda_{b}L_{1}} \right)}{\int dt + \frac{K_{3}}{\alpha_{b1}} \frac{\psi_{b,s1} \left(1 - \cosh(\lambda_{b}L_{1})\right) \left(1 - \cosh(\lambda_{b}L_{2})\right)}{\sinh(\lambda_{b}L_{1}} \right)}{\int dt + \frac{K_{3}}{\alpha_{b1}} \frac{\psi_{b,s1} \sinh(\lambda_{b}L_{1}}{\sinh(\lambda_{b}L_{1})\sinh(\lambda_{b}L_{1})} \right)}{\int dt + \frac{K_{3}}{\alpha_{b1}} \frac{\psi_{b,s1} \sinh(\lambda_{b}L_{1}) \left(1 - \cosh(\lambda_{b}L_{1})\right)^{2}}{\sinh(\lambda_{b}L_{1})\sinh(\lambda_{b}L_{1}} \right)} \right]$$

where,

$$K_{1} = -\frac{\frac{2C_{ox1}(C_{Si1} + C_{rsd1} + C_{rsd2} + C_{box1})}{C_{Si1}(C_{rsd1} + C_{rsd2} + C_{box1})}}{t_{Si}^{2}\left(1 + \frac{2C_{Si1}}{C_{rsd1} + C_{rsd2} + C_{box1}}\right)}$$
(3.118)

$$K_{2} = -\frac{\frac{2C_{ox2}(C_{Si2} + C_{rsd3} + C_{rsd4} + C_{box2})}{C_{Si2}(C_{rsd3} + C_{rsd4} + C_{box2})}}{t_{Si}^{2}\left(1 + \frac{2C_{Si2}}{C_{rsd3} + C_{rsd4} + C_{box2}}\right)}$$
(3.119)

$$K_{3} = -\frac{2}{t_{Si}^{2} \left(1 + \frac{2C_{Si1}}{C_{ox1}}\right)}$$
(3.120)

$$K_{4} = -\frac{2}{t_{Si}^{2} \left(1 + \frac{2C_{Si2}}{C_{ox2}}\right)}$$
(3.121)

3.6 Results and Discussion

Chapter 3

In this section, we have analyzed the derived theoretical models of subthreshold characteristics of the DMG Re-S/D UTB SOI MOSFETs. Further, the models have also been validated against the numerical simulation data obtained by a numerical device simulator ATLASTM. The drift-diffusion model has been used while simulating the carrier transportation in the device. The CVT mobility model has also been used as it is a complete mobility model in which mobility depends on doping density, temperature, lateral and transverse electric field. Fermi-Dirac carrier statistics has been employed to minimize the carrier concentration in the heavily doped regions. Along with this, a quantum transport equation (NEGF) is used to include the quantum effects The numerical threshold voltage has been extracted from the drain current-gate voltage curve by considering the value of that gate voltage at which drain current magnitude is given by $I_d = (W/L) \times 10^{-7} A / \mu m$, where W and L are width and length of the channel, respectively [Conde *et al.* (2002)].

3.6.1 Surface Potential

Figure 3.3 shows both front-channel and back-channel surface potential profiles along the channel length direction for a device channel length of L = 60nm and $t_{rsd} = 30$ nm. It may be noticed that the back-channel minimum surface potential is significantly higher than the front-channel minimum surface potential. This implies that just like Re-S/D SOI MOSFET as discussed in section 3.2, the source-channel barrier height is found lower at back surface of the channel in case of DMG Re-S/D SOI MOSFETs. The estimated value of the source channel barrier height associated with the back-channel is lower than that of the front-channel by ~76mV. This leads to a significant inversion at the back channel when gate-source voltage is raised, and the threshold voltage of the device should be determined by the threshold voltage of the back-surface of the channel. Since, we have considered only lightly doped short-channel length devices in this chapter, the status of back surface of the channel will determine the subthreshold characteristics of the device.

Figure 3.4 demonstrates the back-channel surface potential along the channel length direction for different control-gate to screen-gate length ratios $L_1 : L_2$. It should be noted that the control-to-screen gate ratio is varied whereas the total channel length remains fixed to *L*. As the source side metal gate has higher work function (control gate, ϕ_{M1} =4.8eV) than the drain side metal gate (screen gate, ϕ_{M2} = 4.6eV), the 'virtual cathode' position is found below the control gate and shifted towards the source end when the contribution of screen gate is increased in total channel length. Increasing the contribution of screen gate length from 1/3 to 2/3 in the total gate length, the minimum surface potential increases from 0.19 V to 0.27 V. Thus, varying the gate length ratio ($L_1 : L_2$) could modulate the position and magnitude of the minimum surface potential, which in turn could modulate the subthreshold characteristics of the device.



Fig. 3.3: Surface potential along the channel length at front-gate and back-gate

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET



Fig. 3.4: Back surface potential along the channel at different control to screen gate length ratio

3.6.2 Threshold Voltage

Chapter 3

Figure 3.5 exhibits the threshold voltage variation against the gate length at different oxide thickness. It is observed from the figure that the threshold voltage remains almost constant irrespective of the oxide thickness variations if the device channel length is selected above 150nm, however, devices with channel length less than 150nm become sensitive to the gate oxide thickness. Further, the threshold voltage of device with channel length less than 150nm is found to be decreased severely when a thicker gate oxide is considered in the device.

Figure 3.6 demonstrates the threshold voltage versus device channel length for different Si film thicknesses. The threshold voltage is found to be increased with the decrease in the thickness of the silicon channel. As the thickness of the channel is increased, the controllability of the gate over the entire channel region is reduced due to the reduction in the gate electric field at the deeper section of the channel. Therefore, devices with thinner silicon films have higher channel barrier heights and higher threshold voltage. Thus, a thin Si film is preferred as a channel region as it causes a strong coupling between gate and channel region.

Figure 3.7 shows the threshold voltage versus device channel length at different recessed source/drain thicknesses, t_{rsd} . It should be noted that at $t_{rsd} = 0$, the device structure is similar to a conventional SOI MOSFET where the front-gate controls over the channel. As observed, the threshold voltage is found to be decreased with more and more penetration of source/drain in the BOX (larger t_{rsd}). The decrease in the threshold voltage may be attributed to the coupling of the back-surface of the channel region to the extended source and drain through the buried-oxide. Further, Fig. 3.7 proves the accuracy of our model and its applicability, where Re-S/D SOI MOSFET with two different recessed-source/drain thickness as well as conventional SOI MOSFET ($t_{rsd} = 0$) have been compared.

Figure 3.8 plots the threshold voltage versus channel length at different $L_1 : L_2$ ratio keeping the other device parameters constant. The threshold voltage roll-off is found to be decreased gradually with channel length if a higher ratio of $L_1 : L_2(2:1)$ is chosen. It may be attributed to better short-channel effects (SCEs) immunity in such devices as the control gate length is larger than the screen gate length which in turn shifts the virtual cathode point away from the source end. As a result, it could be said that varying the ratio $L_1 : L_2$ may be an additional option to control the threshold voltage of the device.



Fig. 3.5: Threshold voltage versus channel length for varying oxide thickness.



Fig. 3.6: Threshold voltage versus channel length for varying Si film thickness.



Fig. 3.7: Threshold voltage versus channel length for varying recessed oxide thickness.



Fig. 3.8: Threshold voltage versus channel length for varying control to screen gate length ratio.

3.6.3 Subthreshold Current

Figure 3.9 plots the subthreshold current against the gate-to-source voltage for different channel lengths keeping other device parameters constant. The higher value of subthreshold current at a shorter channel length for a fixed gate voltage may be attributed to the reduced threshold voltage due to the short-channel effects. The influence of control to screen gate length ratio on subthreshold current characteristics of Re-S/D UTB SOI MOSFETs is investigated in Fig. 3.10. Keeping $L_1: L_2 = 1:1$ as reference, the subthreshold current can be observed increasing with smaller control gate length (L_1) for $L_1: L_2 = 1:2$; whereas in case of $L_1: L_2 = 2:1$, the longer L_1 helps to boost the electrostatics in the channel region rendering reduced leakage current.

In Fig. 3.11, the subthreshold current characteristics are plotted with gate-to-source voltages for two different gate oxide thicknesses. Increasing the gate oxide thickness weakens the penetration power of electric field in the channel, which decontrols the channel. Such relaxation in gate control on channel causes enhanced leakage current in the channel. The effect of Si film thickness on the subthreshold current is examined in Fig. 3.12. It is observed from the figure that the subthreshold current is increased with the silicon channel thickness which is attributed to the fact that in case of a thinner silicon body, both the front-channel and the back channel are under the instant control of the gate and thus, the improved short-channel immunity results in reduced leakage current.

The influence of recessed source/drain thickness on subthreshold current is shown in Fig. 3.13 by keeping all other device parameters constant. It is found that at $t_{rsd} = 0$ nm (conventional SOI with back-gate), subthreshold current is ~ 2*10⁻¹²A/µm and as t_{rsd} increases to 30nm and 100nm, subthreshold current is ~ 9*10⁻¹²A/µm and 1*10⁻¹¹A/µm, respectively. It is due to the fact that the thicker recessed source/drain thickness offers higher short-channel effects due to the coupling of recessed source/drain and channel region and thus increasing the subthreshold current. However, a slight increase in subthreshold current is noticed with increasing t_{rsd} above 30nm.

All the theoretical results are well matched with the simulation results obtained from the 2D numerical simulator ATLASTM [SILVCO Int. (2012)] for a gate voltage below the threshold voltage of the device.



Fig. 3.9: Subthreshold current variation with gate-to-source voltage for different channel lengths



Fig. 3.10: Subthreshold current variation with gate-to-source voltage for different control to screen gate length ratios



Fig. 3.11: Subthreshold current variation with gate-to-source voltage for different oxide thicknesses



Fig. 3.12: Subthreshold current variation with gate-to-source voltage for different Si film thicknesses

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-Source/Drain SOI MOSFET



Fig. 3.13: Subthreshold current variation with gate-to-source voltage for different recessed source/drain thicknesses

3.6.4 Subthreshold Swing (*S*)

Chapter 3

The results of Fig. 3.9 and Fig. 3.10 are reflected in Fig. 3.14 that demonstrates the subthreshold swing (S) variation against the device channel length (L) for different control to screen gate length ratios ($L_1 : L_2$). The subthreshold swing is found to be decreased with the increase in the control gate length for a fixed total channel length. The length of the control gate plays important role in determining the device characteristics as devices with longer control gate suffer less with short-channel effects.

The influences of gate oxide thickness and channel thickness on switching characteristics are displayed in Fig. 3.15 and Fig. 3.16 respectively. The subthreshold swing is found to be decreased with the decrease in front-gate oxide thickness and channel thickness for a fixed value channel length. This is due to the fact that a thin

layer of front-gate oxide and silicon channel improves the gate control over the channel by strengthening the vertical electric field across the channel.

Figure 3.17 shows the subthreshold swing variation with the device channel length for different recessed-source/drain thicknesses. It is observed from the figure that the subthreshold swing is found to be increased with increase in recessed-source/drain thickness, which is well expected from the results of Fig. 3.13. In case of $t_{rsd} = 0$ the subthreshold swing is ~ 148mV/Dec., and in case of $t_{rsd} = 30$ nm and $t_{rsd} = 100$ nm the subthreshold is ~ 160 mV/Dec. and ~ 167 mV/Dec., respectively. This increase is due to back coupling between extended source/drain and back surface of the channel region.



Fig. 3.14: Subthreshold swing variation with device channel length for different control to screen gate length ratios



Fig. 3.15: Subthreshold swing variation with device channel length for different oxide thicknesses



Fig. 3.16: Subthreshold swing variation with device channel length for different Si film thicknesses



Fig. 3.17: Subthreshold swing variation with device channel length for different recessed source/drain thicknesses

3.7 Summary and Conclusion

Chapter 3

Analytical models of threshold voltage, subthreshold current and swing of short-channel dual-metal-gate (DMG) fully-depleted (FD) recessed-source/drain (Re-S/D) SOI MOSFET have been presented in this chapter. An extensive analysis has been carried out to find the impact of the numerous device parameters like gate oxide thickness, Si film thickness, and control gate to screen gate length ratio, and recessed source/drain thicknesses on the threshold voltage, subthreshold current and subthreshold swing characteristics of the device. It is found that in case of a lightly doped short-channel length DMG Re-S/D device, due to a strong electrostatics coupling between source drain and back surface of the channel, inversion occurs at the back channel when gate voltage is increased. Thus, threshold voltage of a Re-S/D SOI MOSFET is that value of the gate voltage at which a conducting channel is formed between source and drain near
Analytical Modeling and Simulation of Subthreshold Characteristics
of Short-Channel Dual-Metal-Gate Fully-Depleted Recessed-
Source/Drain SOI MOSFET

the back surface of the channel. Further, it is also concluded that reducing the gate oxide and channel thicknesses effectively curbs down the threshold voltage roll-off. Subthreshold leakage current and swing can also be cut down by selecting thinner gate oxide and silicon film thickness. Thus, the increased threshold voltage roll-off, subthreshold swing and subthreshold current because of recessing source and drain deeper into the BOX can be suppressed by choosing the device parameters judiciously. The control to screen gate length ratio ($L_1 : L_2$) may be an additional controlling parameter which can modulate the subthreshold characteristics of DMG Re-S/D SOI MOSFETs. Further, the ratio $L_1 : L_2$ may be used to optimize the subthreshold behavior of Re-S/D SOI MOSFETs. The proposed models are very general in nature and can predict the subthreshold characteristics of a number of devices including conventional SOI MOSFET (with $t_{rsd}=0$ and $L_2=0$), DMG SOI MOSFETs. The proposed model results are in good agreement with the simulation data obtained by using the commercially available ATLASTM device simulation software from SILVACO.

Chapter 4

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFET with High-*k* Gate-Dielectric Material

4.1 Introduction

The SOI MOSFET is a highly scalable structure. However, an ultra-thin gate oxide (SiO_2) with thickness $t_{ox} < 1$ nm is needed to curb the short-channel effects if the device channel length is scaled down below 32nm [Takeda *et al.* (2010), Ranuarez *et al.* (2006), Chnag *et al.* (2002)]. An ultra-thin gate oxide of thickness less than 1nm causes huge gate-leakage current because of quantum-mechanical direct tunneling of carriers through the gate oxide [Keyes *et al.* (2005), Koh *et al.* (2001)]. Therefore, it has been tried to find out some suitable high-*k* gate-dielectrics with higher physical and lower electrical thickness in order to control both short-channel effects and gate-leakage current simultaneously. HfO₂ (relative permittivity $\varepsilon_k = 22$), ZrO₂ ($\varepsilon_k = 25$), Ta₂O₅ ($\varepsilon_k = 26$) and La₂O₃ ($\varepsilon_k = 27$) are found to be showing excellent performance with high permittivity and energy gap [Tripathi *et al.* (2012), Wilk *et al.* (2001)].

It is observed from Chapter 2 that no significant theoretical work has been reported so far in the literature regarding the subthreshold characteristics of the Re-S/D SOI MOSFET with high-k gate-dielectric material. Keeping the above facts in view, an attempt has been made in this chapter to present a theoretical and simulation based

study of the potential distribution, threshold voltage, subthreshold swing and subthreshold current of the Re-S/D SOI MOSFETs with high-k gate-dielectric material considering the impact of fringing fields. The layout of the present chapter is outlined as follows.

Chapter 4

In section 4.2, the 2D surface potential function of the short-channel FD Re-S/D SOI MOSFET device with high-*k* gate-dielectric material is derived by solving the 2D Poisson's equation using the parabolic approximation method. Section 4.3 presents the analytical threshold voltage modeling of the device. The analytical subthreshold current and subthreshold swing models are developed in sections 4.4 and 4.5 respectively. Results and discussion related to the developed models are presented in section 4.6. Finally, the summary and conclusion of the present chapter are presented in section 4.7.

4.2 Modeling of Two-Dimensional (2D) Surface Potential

This section presents the modeling of subthreshold characteristics of Re-S/D UTB SOI MOSFET with high-*k* gate-dielectric material. The schematic structure of the Re-S/D UTB SOI MOSFET device with high-*k* gate-dielectric material used in our modeling and simulation is shown in Fig.4.1. Here, $L, W, t_{Si}, t_{gt}, t_{sp}, t_k, t_{box}, t_{rsd}$ and d_{box} are the gate length, gate width along *z* axis , channel thickness, thickness of gate, thickness of spacer, gate-dielectric thickness, buried-oxide (BOX) thickness, source/drain penetration in the BOX and length of source/drain overlap over the BOX, respectively. The other parameters used in the analysis of the device of Fig. 4.1 are considered same as those in Fig. 3.1. of Chapter 3. The symbols and values of all the parameters used in modeling and simulation are detailed in Table 4.1.



Fig. 4.1: The schematic structure of the Re-S/D FD SOI MOSFET showing the internal parasitic fringe capacitance

Parameters	Symbol	Values
Gate work-function	ϕ_M	4.6, 4.8eV
Channel Doping	N _a	10^{15}cm^{-3}
Source/Drain Doping	N_d	10^{20}cm^{-3}
Substrate Doping	N _{sub}	10^{18}cm^{-3}
Silicon Thickness	t _{si}	5-8nm
Physical Gate-Dielectric Thickness	t_k	1.5-15nm
Effective Oxide Thickness	t_{ox}	1.5-3nm
Buried-Oxide Thickness	t _{box}	100-200nm
The depth of S/D in the Buried-Oxide	t _{rsd}	30, 100nm
Gate-Thickness	t _{gt}	25nm
Gate-Dielectric Constant	ε	3.9, 12, 22, 27
Thickness of the Spacer	t _{sp}	25nm
Gate Width	W	1µm
Length of S/D Overlap	d_{box}	3nm
Channel Length	L	20-300nm
Gate Voltage	V_{GS}	0.1V
Drain Voltage	V_{DS}	0.1V
Substrate Voltage	V _{sub}	-3.0 to 1V

Chapter 4

 Table 4.1: Device parameter values used for modeling and simulation of Re-S/D FD SOI

 MOSFET with high-k gate-dielectric material

In the subthrehold regime of the device operation, the 2D Poisson's equation in the channel region of the device, shown in Fig. 4.1, can be written as follows [Young *et al.* (1989)]

$$\frac{\partial^2 \phi(x, y)}{\partial x^2} + \frac{\partial^2 \phi(x, y)}{\partial y^2} = \frac{qN_a}{\varepsilon_{Si}}$$
(4.1)

where, $\phi(x, y)$ is the channel potential and ε_{s_i} is permittivity of silicon.

The potential profile in the vertical direction in the channel can be approximated by the following parabolic function [Young *et al.* (1989)]

$$\phi(x, y) = \phi_f(x) + C_1(x)y + C_2(x)y^2$$
(4.2)

The coefficients $C_1(x)$ and $C_2(x)$ are the functions of x only, and $\phi_f(x) = \phi(x,0)$ is the surface potential at gate-dielectric/Si interface. These coefficients are to be determined using the following boundary conditions:

The continuity of electric flux at gate-dielectric/Si interface is as [Svilicic et al. (2009)]:

$$\left[\frac{\partial\phi(x,y)}{\partial y}\right]_{y=0} = \frac{C_k}{\varepsilon_{Si}} \frac{\phi_f(x) \cdot (V_{GS} - V_{FB1})}{L}$$
(4.3)

The electric flux at Si/BOX interface is also continuous and is given as [Svilicic *et al.* (2009)]:

$$\begin{bmatrix} \frac{\partial \phi(x, y)}{\partial y} \end{bmatrix}_{y=t_{Si}} = \frac{C_{rsd}}{\varepsilon_{Si}} \frac{V_S - V_{FB2} - \phi_b(x)}{L} + \frac{C_{rsd}}{\varepsilon_{Si}} \frac{V_D - V_{FB2} - \phi_b(x)}{L} + \frac{C_{box}}{\varepsilon_{Si}} \frac{V_{sub} - V_{FB3} - \phi_b(x)}{L}$$

$$(4.4)$$

 N_d is the source/drain doping concentration, n_i is the intrinsic carrier concentration in the silicon, V_{FB1} , V_{FB2} and V_{FB3} are the gate-front channel flat-band voltage, source/drain-back channel flat-band voltage and substrate-back channel flat-band voltage, respectively and are given by

$$V_{FB1} = \phi_M - \phi_{Si} \tag{4.5}$$

$$V_{FB2} = \frac{kT}{q} \ln \left(\frac{N_a N_d}{n_i^2} \right)$$
(4.6)

$$V_{FB3} = \frac{kT}{q} \ln\left(\frac{N_{sub}}{N_a}\right)$$
(4.7)

where, ϕ_M is the gate work function, $\phi_{Si} = \chi_{Si} + E_g/2 + V_T \ln\left(\frac{N_a}{n_i}\right)$, χ_{Si} and E_g are the electron affinity and energy band gap of the silicon, respectively, and V_T is the thermal

voltage; V_D and V_{GS} are the drain and gate to source bias-voltages, respectively; The potential at channel-BOX interface is $\phi_b(x) = \phi_b(x, y = t_{Si})$, which is named as back surface potential. The symbol V_{sub} refers to the substrate bias voltage.

 C_k is the front-channel gate-dielectric capacitance per unit length and it is given by

$$C_k = \frac{\varepsilon_k L}{t_k} \tag{4.8}$$

 C_{si} is the silicon channel capacitance per unit length and it is given by

$$C_{Si} = \frac{\varepsilon_{Si}L}{t_{Si}}$$
(4.9)

 C_{box} is the buried-oxide capacitance per unit length and it is given by

$$C_{box} = \frac{\varepsilon_{ox}L}{t_{box}}$$
(4.10)

 C_{rsd} is the recessed-source/drain buried-oxide capacitance per unit length and it is given by [Svilicic et al. (2009)]

$$C_{rsd} = \begin{cases} \frac{\varepsilon_{box}}{\theta} \ln\left(1 + \frac{L}{2d_{box}}\right) for \frac{L}{2} < t_{rsd} \\ \frac{\varepsilon_{box}}{\theta} \ln\left(1 + \frac{t_{rsd}}{2d_{box}}\right) for \frac{L}{2} > t_{rsd} , t_{rsd} = 0 \end{cases}$$
(4.11)

where, d_{box} is the length of source/drain overlap over buried-oxide which should be greater than zero, θ is the effective angle between the two slanted electrodes: channel back-side and source/drain extension; $\theta = \left(\frac{\pi}{2} + \frac{\pi}{2} \cdot \sec h \frac{t_{rsd}}{d_{box}}\right)$, From this equation it can

be noticed that θ varies from $\frac{\pi}{2}$ for the Re-S/D SOI MOSFETs to π for the conventional SOI MOSFETs ($t_{rsd} = 0$).

The surface potential at the source end is

$$\phi(0, y) = V_{bi} \tag{4.12}$$

whereas, the surface potential at the drain end is

$$\phi(L, y) = V_{bi} + V_{DS} \tag{4.13}$$

where, V_{bi} is the built-in voltage at the source/drain and Si interface given by

$$V_{bi} = \frac{kT}{q} ln \left(\frac{N_a N_d}{n_i^2} \right),$$

Now, the coefficients $C_1(x)$ and $C_2(x)$ of Eq. (4.2) are determined using the boundary conditions Eq. (4.3) and (4.4) as

$$C_{1}(x) = \frac{C_{k}}{t_{Si} \cdot C_{Si}} \left[\phi_{f}(x) - (V_{G} - V_{FB1}) \right]$$
(4.14)

$$C_{2}(x) = \frac{1}{2t_{Si}^{2}} \left\{ \frac{C_{rsd}}{C_{Si}} \left[V_{DS} - 2V_{FB2} - \phi_{b}(x) \right] + \frac{C_{box}}{C_{Si}} \left[V_{sub} - V_{FB3} - \phi_{b}(x) \right] + \frac{C_{k}}{C_{Si}} \left[\phi_{f}(x) - (V_{G} - V_{FB1}) \right] \right\}$$
(4.15)

Now, using $C_1(x)$ and $C_2(x)$ of Eqs. (4.14) and (4.15) in Eq. (4.2), gives the following equation of 2D channel potential

$$\phi(x, y) = \phi_{f}(x) + \frac{C_{k}}{t_{Si} \cdot C_{Si}} \left[\phi_{f}(x) - (V_{G} - V_{FB1}) \right] y + \frac{1}{2t_{Si}^{2}} \left\{ \frac{C_{rsd}}{C_{Si}} \left[V_{DS} - 2V_{FB2} - \phi_{b}(x) \right] + \frac{C_{box}}{C_{Si}} \left[V_{sub} - V_{FB3} - \phi_{b}(x) \right] + \frac{C_{k}}{C_{Si}} \left[\phi_{f}(x) - (V_{G} - V_{FB1}) \right] \right\} y^{2} (4.16)$$

Further, the following two differential equations for $\phi_{f,b}$ have been obtained using $\phi(x, y)$ of Eq. (4.16) in Eq. (4.1) at y = 0 and $y = t_{Si}$ respectively,

$$\frac{d^2 \phi_{f,b}(x)}{dx^2} - \alpha_{f,b} \phi_{f,b}(x) = \beta_{f,b}$$
(4.17)

where,

Chapter 4

$$\alpha_{f} = 2 \frac{1 + \left(\frac{C_{k} (C_{Si} + 2C_{rsd} + C_{box})}{C_{Si} (2C_{rsd} + C_{k})}\right)}{t_{Si}^{2} \left(1 + 2 \frac{C_{Si}}{2C_{rsd} + C_{k}}\right)}$$
(4.18)

$$\int \left[\left(V_{GS} - V_{FB1} \right) \left(\frac{C_k \left(C_{Si} + 2C_{rsd} + C_{box} \right)}{C_{Si} \left(2C_{rsd} + C_k \right)} \right) + \left(V_{DS} - 2V_{FB2} \right) \left(\frac{2C_{rsd}}{2C_{rsd} + C_{box}} \right) \right)$$

$$\beta_f = \frac{qN_a}{\varepsilon_{Si}} - 2 \frac{\left(+ \left(V_{sub} - V_{FB3} \right) \left(\frac{2C_{box}}{2C_{rsd} + C_{box}} \right) \right)}{t_{Si}^2 \left(1 + 2 \frac{C_{Si}}{2C_{rsd} + C_k} \right)}$$

$$(4.19)$$

$$\alpha_{b} = 2 \frac{1 + \left(\frac{(C_{k} + C_{Si})(2C_{rsd} + C_{box})}{C_{k}C_{Si}}\right)}{t_{Si}^{2}\left(1 + 2\frac{C_{Si}}{C_{k}}\right)}$$
(4.20)

$$\beta_{b} = \frac{qN_{a}}{\varepsilon_{Si}} - 2 \frac{\left(V_{GS} - V_{FB1}\right) + \left(V_{DS} - 2V_{FB2}\right) \left(\frac{C_{rsd}}{C_{k}} + \frac{C_{rsd}}{C_{Si}}\right) + \left(V_{sub} - V_{FB3}\right)}{t_{Si}^{2} \left(1 + 2\frac{C_{Si}}{C_{k}}\right)}$$
(4.21)

By solving the second-order non-homogenous differential equation (4.17), the following expression which is front- and back- surface potential distribution along the channel is obtained

$$\phi_{f,b} = A_{f,b} \cdot e^{\sqrt{\alpha_{f,b}}x} + B_{f,b} \cdot e^{-\sqrt{\alpha_{f,b}}x} - \frac{\beta_{f,b}}{\alpha_{f,b}}$$
(4.22)

Now, the boundary conditions of Eqs. (4.12) and (4.13) can be used to determine the coefficients $A_{f,b}$ and $B_{f,b}$ of Eq. (4.22).

where,

Chapter 4

$$A_{f,b} = \frac{\beta_{f,b} \left(e^{\sqrt{\alpha_{f,b}}L} - 1 \right) + \alpha_{f,b} \left[V_{bi} \left(e^{\sqrt{\alpha_{f,b}}L} - 1 \right) + V_{DS} e^{\sqrt{\alpha_{f,b}}L} \right]}{\alpha_{f,b} \left(e^{2\sqrt{\alpha_{f,b}}L} - 1 \right)}$$
(4.23)

$$B_{f,b} = \frac{e^{\sqrt{\alpha_{f,b}L}} \left\{ \beta_{f,b} \left(e^{\sqrt{\alpha_{f,b}L}} - 1 \right) - \alpha_{f,b} \left[V_{bi} \left(1 - e^{\sqrt{\alpha_{f,b}L}} \right) + V_{DS} \right] \right\}}{\alpha_{f,b} \left(e^{2\sqrt{\alpha_{f,b}L}} - 1 \right)}$$
(4.24)

The positions $(x_{\min})_{f,b}$ of the front- and back- channel minimum surface potentials (virtual cathode potential, $\phi_{f\min,b\min}$ [Svilicic *et al.* (2009)] can be obtained by solving

$$\frac{d\phi_{f,b}(x)}{dx}\bigg|_{(x_{\min})_{f,b}} = 0 \text{ and are determined as}$$

$$(x_{\min})_{f,b} = \frac{1}{2} \frac{\ln\left(\frac{B_{f,b}}{A_{f,b}}\right)}{\sqrt{\alpha_{f,b}}}$$
(4.25)

The minimum of front- and back- surface potential $\phi_{f\min,b\min}$ could be obtained by substituting $(x_{\min})_{f,b}$ of Eq. (4.25) into Eq. (4.22) as

$$\phi_{f\min,b\min} = 2\sqrt{A_{f,b}B_{f,b}} - \frac{\beta_{f,b}}{\alpha_{f,b}}$$
(4.26)

Further, in order to obtain the surface potential equations including the effect of the fringe capacitance C_{bot} , the potentials $V(x,\sigma_1)|_{source}$ and $V(L-x,\sigma_2)|_{drain}$ due to the induced charges on the source and drain regions, respectively, are added to surface potential $\phi_{f,b}$ and the expression for surface potential of Eq. (4.22) is modified as

$$\phi_{f,b} = A_{f,b} \cdot e^{\sqrt{\alpha_{f,b}}x} + B_{f,b} \cdot e^{-\sqrt{\alpha_{f,b}}x} - \frac{\beta_{f,b}}{\alpha_{f,b}} + \left(V(x,\sigma_1) \Big|_{source} + V(L-x,\sigma_2) \Big|_{drain} \right)$$
(4.27)

The origin and derivation of potential $V(x,\sigma_{1,2})$ and fringe capacitance C_{bot} are discussed in section 4.2.1

Now, the modified minimum of front- and back- surface potential could be obtained using Eq. (4.27). However, the direct derivative of Eq. (4.27) may lead us to a very complicated equation of $(x_{min})_{f,b}$ and which in turn gives an implicit equation for threshold voltage. Therefore, assuming that first two terms are varying more rapidly compared to the last two, and the last two terms only modify the minimum values of $\phi_{f\min,b\min}$, following equations for the modified minimum front- and back- surface potential can be obtained as

$$\phi'_{f\min,b\min} = 2\sqrt{A_{f,b}B_{f,b}} - \frac{\beta_{f,b}}{\alpha_{f,b}} + \left(V(x,\sigma_1)\right)_{source} + V(L - x,\sigma_2)_{drain}\right)_{x = (x_{\min})_{f,b}}$$
(4.28)

Now, we derive the following equation of the so-called virtual cathode potential $\phi_{vc}(y)$ [Chen *et al.* (2003)] $(\phi_{vc}(y) = \phi_1(x, y)|_{x=(x_{\min})_{f,b}}$, where $x = (x_{\min})_{f,b}$ is the position where minima of front-/back-surface potential exists). The obtained virtual cathode potential $\phi_{vc}(y)$ which is a function of the minimum value of the front- surface potential is

$$\phi_{vc}(y) = \phi_{f \min}' \left[1 + \frac{C_{k}}{C_{Si} t_{Si}} y - \frac{C_{k}}{C_{Si} t_{Si}^{2}} y^{2} - (2C_{Si} + C_{k}) z_{1} y^{2} \right] + (V_{GS} - V_{FB1}) \left[-\frac{C_{k}}{C_{Si} t_{Si}} y + \frac{C_{k}}{2C_{Si} t_{Si}^{2}} y^{2} - (C_{k}) z_{1} y^{2} \right] + (V_{DS} - 2V_{FB2}) \left[\frac{C_{rsd}}{2C_{Si} t_{Si}^{2}} y^{2} + (C_{rsd}) z_{1} y^{2} \right] + (V_{sub} - V_{FB4}) \left[\frac{C_{box}}{2C_{Si} t_{Si}^{2}} y^{2} + (C_{box}) z_{1} y^{2} \right]$$
(4.29)
where, $z_{1} = \left(\frac{2C_{rsd} + C_{box1}}{2C_{Si} t_{Si}^{2} (2C_{Si} + 2C_{rsd} + C_{box})} \right)$
(4.30)

Similarly, another equation of the so-called virtual cathode potential which is a function of the minimum value of the back surface potential can also be obtained

$$\phi_{vc}(y) = \phi'_{b\min} \left[\left(\frac{2C_{Si} + 2C_{rsd} + C_{box}}{2C_{Si} + C_{k}} \right) z_{2} - \left(\frac{C_{rsd}}{C_{Si1} t_{Si}^{2}} + \frac{C_{box}}{2C_{Si1} t_{Si}^{2}} \right) y^{2} \right]$$

$$+ (V_{GS} - V_{FB1}) \left[-\frac{C_{k}}{C_{Si} t_{Si}} y + \frac{C_{k}}{C_{Si} t_{Si}^{2}} y^{2} + \left(\frac{C_{k}}{2C_{Si} + C_{k}} \right) z_{2} \right]$$

$$+ (V_{DS} - 2V_{FB2}) \left[\frac{C_{rsd}}{C_{Si1} t_{Si}^{2}} y^{2} - \left(\frac{C_{rsd}}{2C_{Si} + C_{k}} \right) z_{2} \right]$$

$$+ (V_{sub} - V_{FB3}) \left[\frac{C_{box}}{C_{Si} t_{Si}^{2}} y^{2} - \left(\frac{C_{box}}{2C_{Si} + C_{k}} \right) z_{2} \right]$$

$$(4.31)$$

where,
$$z_2 = \left(1 + \frac{C_k}{C_{s_i} t_{s_i}} y - \frac{C_k}{2C_{s_i} t_{s_i}^2} y^2\right)$$
 (4.32)

Here, it should be noted that either of Eqs. (4.29) or (4.31) can be utilized for the formulation of the subthreshold charge carrier density at virtual cathode position, as done in Chapter 3.

4.2.1 Derivation of $V(x, \sigma_i)$

Chapter 4

The electric field lines that originate from the bottom of the gate and enter into the source and drain through the spacer induce electrostatic charges in source and drain region. The surface charge density, σ_i (i=1 and 2 for source and drain, respectively) due to fringing electric field lines is given as (Fig. 4.2) [Kumar *et al.* (2006)]

$$\sigma_{i} = \left(\frac{(C_{bot})V_{pi}}{W.t_{sp}}\right), \text{ where } i = 1, 2 \text{ for source and drain, respectively}$$

$$4.33)$$

where, $V_{p1} = V_{bi} - V_{GS} + V_{FB2}$ and $V_{p2} = V_{bi} + V_{DS} - V_{GS} + V_{FB2}$ are the source and drain junction potentials respectively, W is width of the gate along z axis and t_{sp} is the spacer thickness between gate sidewall and source/drain contacts.

Parasitic internal fringe capacitances associated with bottom edge of gate electrode can be calculated as [Kumar *et al.* (2006)]

$$C_{bot} = \frac{(0.3)\varepsilon_{eff}W}{\pi}$$
(4.34)

where, $\varepsilon_{eff} = \frac{\varepsilon_k \varepsilon_{sp}}{\varepsilon_k - \varepsilon_{sp}} \ln \left(\frac{\varepsilon_k}{\varepsilon_{sp}} \right)$, $\varepsilon_{sp} = \left(1 + \frac{t_k}{L} \right) \varepsilon_{sp}$ is the effective permittivity of spacer

material, ε_k and ε_{sp} are the permittivity of gate-dielectric and spacer materials respectively, $t_k = t_{ox} \left(\frac{\varepsilon_k}{\varepsilon_{Sio_2}} \right)$ is the physical dielectric thickness, t_{ox} is the effective oxide

thickness (EOT), and ε_{Sio_2} is the permittivity of silicon dioxide [Kumar *et al.* (2006)].



Fig. 4.2: Potential on source and drain due to uniformly charged plate

Now, the potential $V(x, \sigma_i)$ at a distance 'x' from uniformly charged plate as shown in Fig. (4.2) [Kumar *et al.* (2006)]

$$V(x,\sigma_i) = \frac{1}{2} \int_{0}^{t_{sp}} \int_{\frac{W}{2}}^{\frac{W}{2}} \frac{\sigma_i dx_1 dx_2}{4\pi\varepsilon_{si} \sqrt{(x+x_1)^2 + (x_2)^2}}$$
(4.35)

By calculating the integral in Eq. (4.35) the expression for $V(x, \sigma_i)$ is obtained as

$$V(x,\sigma_i) = \frac{\sigma_i}{8\pi\varepsilon_{si}} [U(x)]$$
(4.36)

where,

Chapter 4

$$U(x) = (x + t_{sp}) ln \left\{ \frac{\sqrt{(x + t_{sp})^{2} + (\frac{W^{2}}{4})} + (\frac{W}{2})}{\sqrt{(x + t_{sp})^{2} + (\frac{W^{2}}{4})} - (\frac{W}{2})} - x.ln \left\{ \frac{\sqrt{x^{2} + (\frac{W^{2}}{4})} + (\frac{W}{2})}{\sqrt{x^{2} + (\frac{W^{2}}{4})} - (\frac{W}{2})} \right\} + W.ln \left\{ \frac{\sqrt{x^{2} + (\frac{W^{2}}{4})} + t_{sp} + \sqrt{x^{2} + t_{sp}^{2} + 2t_{sp}\sqrt{x^{2} + (\frac{W^{2}}{4})}}}{\sqrt{x^{2} + (\frac{W^{2}}{4})} + x} \right\}$$
(4.37)

4.3 Threshold Voltage Formulation

The threshold voltage associated with front surface and back surface of the channel could be obtained using [Chen *et al.* (2003)]

$$\phi_{f\min,b\min}'\Big|_{V_{G}=V_{thf,b,}} = 2\phi^{*}_{f,Si} = \begin{cases} 2\phi_{f,Si} & \text{for}, N_{a} > n_{T} \\ \phi_{f,Si} + \frac{kT}{q}\ln\left(\frac{n_{T}}{n_{i}}\right) & \text{for}, N_{a} < n_{T} \end{cases}$$
(4.38)

where, q is the electron charge, N_a is the channel doping concentration, n_i is the intrinsic carrier concentration, k is the Boltzmann's constant, and T is the temperature, $\phi_{f,Si}$ is the Fermi potential, n_T is a critical concentration of electrons in channel to turn on the device [Chen *et al.* (2003)]. However, the critical concentration, n_T depends on device parameters like channel thickness and channel length and it has been obtained by following the method of Chen [Chen *et al.* (2003)].

Now, after using Eq. (4.38) in Eq. (4.28), we get the following equations for front and back channel-surface threshold voltages $V_{th f,b}$

$$V_{ihf,b} = \frac{-b_{f,b} + \sqrt{b_{f,b}^{2} - 4a_{f,b}c_{f,b}}}{2a_{f,b}}$$
(4.39)

where,

$$a_{f,b} = u_{f1,b1} u_{f2,b2} - \left(\frac{m'_{f,b}}{2\alpha_{f,b}}\right)^2$$
(4.40)

$$c_{f,b} = v_{f1,b1} v_{f2,b2} - \phi_{f,Si}^{2} - \left(\frac{n'_{f,b}}{2\alpha_{f,b}}\right)^{2} - \frac{n'_{f,b}}{\alpha_{f,b}} \phi_{f,Si}$$
(4.41)

$$b_{f,b} = u_{f1,b1}v_{f2,b2} + u_{f2,b2}v_{f1,b1} - \frac{m'_{f,b}n'_{f,b}}{2\alpha_{f,b}^2} - \frac{m'_{f,b}}{\alpha_{f,b}}\phi_{f,Si}$$
(4.42)

$$v_{f1,b1} = \frac{n_{f,b}' \left(e^{\sqrt{\alpha_{f,b}}L} - 1 \right) + \alpha_{f,b} \left[V_{bi} \left(e^{\sqrt{\alpha_{f,b}}L} - 1 \right) + V_{DS} e^{\sqrt{\alpha_{f,b}}L} \right]}{\alpha_{f,b} \left(e^{2\sqrt{\alpha_{f,b}}L} - 1 \right)}$$
(4.43)

$$v_{f2,b2} = \frac{e^{\sqrt{\alpha_{f,b}}L} \left\{ n_{f,b}' \left(e^{\sqrt{\alpha_{f,b}}L} - 1 \right) - \alpha_{f,b} \left[V_{bi} \left(1 - e^{\sqrt{\alpha_{f,b}}L} \right) + V_{DS} \right] \right\}}{\alpha_{f,b} \left(e^{2\sqrt{\alpha_{f,b}}L} - 1 \right)}$$
(4.44)

$$u_{f1,b1} = \frac{m'_{f,b} \left(e^{\sqrt{\alpha_{f,b}}L} - 1 \right)}{\alpha_{f,b} \left(e^{2\sqrt{\alpha_{f,b}}L} - 1 \right)}$$
(4.45)

$$u_{f2,b2} = \frac{m'_{f,b} e^{\sqrt{\alpha_{f,b}}L} \left(e^{\sqrt{\alpha_{f,b}}L} - 1 \right)}{\alpha_{f,b} \left(e^{2\sqrt{\alpha_{f,b}}L} - 1 \right)}$$
(4.46)

$$m'_{f,b} = m_{f,b} + P_1 \alpha_{f,b} \tag{4.47}$$

$$n'_{f,b} = n_{f,b} + P_2 \alpha_{f,b} \tag{4.48}$$

$$\beta_{f,b} = m_{f,b} V_{GS} + n_{f,b} \tag{4.49}$$

$$U(x)_{x=x_{min}} + U(x)_{x=L-x_{min}} = P_1 V_{GS} + P_2$$
(4.50)

where, P_1 and P_2 are the constants which can be obtained by rearranging the expression $U(x)_{x=x_{min}} + U(x)_{x=L-x_{min}}$

Here, it should be noted that in Re-S/D SOI MOSFETs, the position of inversion layer depends on the device parameters such as channel doping, recessed-source/drain

thickness and channel length as discussed in Chapter 3. Thus, the threshold voltage V_{th} of Re-S/D SOI MOSFETs with high-*k* gate-dielectric material could be given as follows [Kumar *et al.* (2014)]

$$V_{th} = \begin{cases} V_{thf}, & \text{for } \phi_{f,\min} > \phi_{b,\min} \\ V_{thb}, & \text{for } \phi_{f,\min} < \phi_{b,\min} \end{cases}$$
(4.51)

Further, the following quantum effects induced correction term ΔV_{th} is added in the threshold voltage model of Eq. (4.51) [Dort *et al.* (1992)].

$$\Delta V_{th} = \Delta E_0 \left(1 + \frac{\varepsilon_{Si} t'_{ox}}{\varepsilon_k t_{Si} + \varepsilon_{Si} t_{box}} \right)$$
(4.52)

where, *h* is the planks constant, m^* is effective mass of the electron, ΔE_0 can be found out using Eq. (3.98) of Chapter 3.

4.4 Subthreshold Current Formulation

By assuming that the diffusion is the dominant current flow mechanism in the weak inversion regime [Dey *et al.* (2008)], the subthreshold current can be modeled by using the concept of effective channel length. Thus, the expression of subthreshold current density can be written as [Dey *et al.* (2008)]:

$$J_n(y) = \frac{qD_n n_{\min}(y)}{L_e} \left(1 - \exp\left(-\frac{V_{DS}}{V_T}\right) \right)$$
(4.53)

 D_n is the coefficient of diffusion, L_e is the effective channel length [Dey at al. (2008)], V_T is the thermal voltage. The electron density at the virtual cathode, $n_{\min}(y)$ can be expressed as [Yeh *et al.* (1995), Dey at al. (2008)]

$$n_{min}(y) = \frac{n_i^2}{N_a} exp\left(\frac{\phi_{vc}(y_m)}{V_T}\right)$$
(4.54)

where, $\phi_{vc}(y_m) = \phi_{vc}(y)$ is the value of the minimum surface potential function along the channel thickness at $y = y_m$, where y_m can be obtained by solving $\frac{\partial \phi_{vc}(y)}{\partial y} = 0$ and $\phi_{vc}(y_m) = \phi_{vc}(y)|_{y=y}$

The subthreshold current, I_{sub} can be obtained by integrating Eqn. (4.53) along the silicon film, i.e., from 0 to t_{si} , as done in Chapter 3.

$$I_{sub} = \int_{0}^{t_{Si}} J_{n}(y) dy = \int_{0}^{t_{Si}} \frac{qD_{n}n_{i}^{2}}{L_{e}N_{a}} \exp\left(\frac{\phi_{vc}(y)}{V_{T}}\right) \left(1 - \exp\left(-\frac{V_{DS}}{V_{T}}\right)\right) dy$$
(4.53)

$$I_{sub} = K \int_{0}^{t_{si}} \exp\left(\frac{\phi_{vc}(y)}{V_T}\right) dy$$
(4.54)

where,
$$K = \frac{qD_n n_i^2}{L_e N_a} \left(1 - \exp\left(-\frac{V_{DS}}{V_T}\right) \right)$$
 (4.55)

By following the method which is given in section 3.4 of Chapter 3, the above equation has been solved and the obtained final subthreshold current expression is

$$I_{sub} = KV_{T} \left[\frac{\exp\left(\frac{\phi_{vc}(y=0)}{V_{T}}\right) - \exp\left(\frac{\phi_{vc}(y=y_{m})}{V_{T}}\right)}{(\phi_{vc}(y=0) - \phi_{vc}(y=y_{m}))/y_{m}} + \frac{\exp\left(\frac{\phi_{vc}(y=y_{m})}{V_{T}}\right) - \exp\left(\frac{\phi_{vc}(y=t_{Si})}{V_{T}}\right)}{(\phi_{vc}(y=y_{m}) - \phi_{vc}(y=t_{Si}))/(t_{Si} - y_{m})} \right]$$
(4.56)

4.5 Subthreshold Swing Formulation

The subthreshold slope (S) of a Re-S/D SOI MOSFET is defined as the reciprocal of the slope of the subthreshold characteristic curve representing the variation of $log(I_D)$ as a

function of the gate voltage, V_G , where, I_D is the drain current in the subthreshold regime of operation of the device. Thus, the subthreshold swing, S of the Re-S/D SOI MOSFET can be written as [Dey *et al.* (2008), Svilicic *et al.* (2010)]

$$S = \begin{cases} V_T (\ln 10) \times \left(\frac{d\phi'_{f\min}}{dV_{GS}}\right)^{-1} for \phi'_{f\min} > \phi'_{b\min} \\ V_T (\ln 10) \times \left(\frac{d\phi'_{b\min}}{dV_{GS}}\right)^{-1} for \phi'_{b\min} > \phi'_{f\min} \end{cases}$$
(4.57)

By substituting the Eq. (4.28) in Eq. (4.57) the following closed form expression of the subthreshold swing of the Re-S/D UTB SOI MOSFET with high-*k* gate-dielectric material is obtained as

$$S = \begin{cases} V_{T}(ln10) \times \left[\left(1 + \frac{C_{Si}(2C_{rsd} + C_{box})}{C_{K}(C_{Si} + 2C_{rsd} + C_{box})} \right) \left(\frac{1 + e^{\sqrt{\alpha_{f}L}}}{1 + e^{\sqrt{\alpha_{f}L}} - e^{\sqrt{\alpha_{f}}x_{\min,f}} - e^{\sqrt{\alpha_{f}}(L - x_{\min,f})}} \right) \right]^{-1} \\ - \frac{C_{bot}}{Wt_{sp}8\pi\varepsilon_{Si}} (U_{S} + U_{D}) \\ for \phi'_{f\min} > \phi'_{b\min} \\ V_{T}(ln10) \times \left(\left(1 + \frac{(C_{K} + C_{Si})(2C_{rsd} + C_{box})}{C_{K}C_{Si}} \right) \left(\frac{1 + e^{\sqrt{\alpha_{b}L}}}{1 + e^{\sqrt{\alpha_{b}L}} - e^{\sqrt{\alpha_{b}}x_{\min,b}} - e^{\sqrt{\alpha_{b}}(L - x_{\min,b})}} \right) \right]^{-1} \\ - \frac{C_{bot}}{Wt_{sp}8\pi\varepsilon_{Si}} (U_{S} + U_{D}) \\ for \phi'_{b\min} > \phi'_{f\min} \end{cases}$$
(4.58)

where, $U_{S} = U(x)|_{x=x_{min}}, U_{D} = U(x)|_{x=L-x_{min}}$

4.6 **Results and Discussion**

This section presents some theoretical and ATLASTM simulation results of the surface potential, threshold voltage, subthreshold current and subthreshold swing of the Re-S/D SOI MOSFETs with high-*k* gate-dielectric material. The high-*k* gate-dielectric material considered for modeling and simulations is HfO₂ with a relative permittivity $\varepsilon_k = 22$,

unless stated otherwise. All the results have been plotted by considering the same effective oxide thickness (EOT), however, the physical dielectric thickness varies according to this relation: physical dielectric thickness, $t_k = t_{ox} \left(\frac{\varepsilon_k}{\varepsilon_{sio_x}} \right)$; t_{ox} is the

effective oxide thickness (EOT), and ε_{Sio_2} is the permittivity of silicon dioxide. Further, three types of parasitic capacitances might affect the device performance: the bottom capacitance, C_{bot} , between bottom edge of the gate electrode to the source and drain regions (through high-*k*); C_{pp} , the parallel plate capacitance between gate and source/drain electrodes; and C_{top} , the top fringing capacitance between top surface of the gate and source/drain electrodes through the first layer of passivation dielectrics [Mohapatra *et al.* (2003)]. However, a thick spacer between source and drain is considered in our proposed structure. Thus, the effect of C_{pp} and C_{top} is negligible on the device characteristics. The simulation methodology has been discussed in section 3.6 of Chapter 3.

4.6.1 Surface Potential

The front-channel and back-channel surface potentials of a Re-S/D SOI MOSFET along the channel length direction are displayed in Fig. 4.3 considering HfO₂ ($\varepsilon_k = 22$) as a gate- dielectric material. Other device parameters are $\phi_M = 4.8$ eV, L = 45nm, $N_a = 10^{16}$ cm⁻³, $N_d = 10^{20}$ cm⁻³, $t_{box} = 200$ nm, $t_{Si} = 8$ nm, $t_{rsd} = 30$ nm, $t_{ox} = 1.5$ nm, $V_{DS} = 0.1$ V, $V_{GS} = 0.1$ V. It is noticed from the figure that back-channel minimum surface potential is higher than the front-channel minimum surface potential by 30mV for the selected device dimensions and the channel-doping level. Therefore, the device threshold voltage is mainly under the control of the back-channel rather than the frontchannel in contrast to the conventional SOI MOSFETs.

Figure 4.4 demonstrates the back surface potential variation against the channel length for different gate-dielectric constants (ε_k). A material with high dielectric constant as a gate-dielectric raises the minimum value of back surface potential. Figure 4.4

demonstrates the back surface potential variation against the channel length for different gate-dielectric constants (ε_k). A gate-dielectric with high dielectric constant (more fringing fields) raises the minimum value of back surface potential. Thus, results in lowering the source-channel barrier height, which subsequently results in the threshold voltage change. Therefore, the fringing fields due to the bottom capacitance, C_{bot} between bottom edge of the gate electrode to the source and drain regions could considerably modulate the threshold voltage of a Re-S/D SOI MOSFET. In Fig. 4.4, the highest level of minimum surface potential is observed for $\varepsilon_k = 27$ (more fringing fields).



Fig. 4.3: Surface potential along the channel length at front-channel and back-channel



Fig. 4.4: Back surface potential along the channel length for various gate-dielectric constants (ε_k)

4.6.2 Threshold Voltage

Chapter 4

Figure 4.5 presents the variation of threshold voltage with respect to channel length for different effective oxide thicknesses, t_{ox} =1.5nm and 3nm. It is observed that for a fixed value of channel thickness, the capacitive coupling between the gate electrode and channel region increases for a reduced gate-dielectric thickness. Thus, the gate electrode has more control over the channel region, resulting in decreased threshold voltage roll-off.

Figure 4.6 exhibits the variation of threshold voltage with respect to channel length while t_{rsd} as a variable parameter with HfO₂ as a gate-dielectric material. The selected

thickness for source and drain penetration in the BOX is $t_{rsd} = 0$ nm, 30nm, and 100nm. It should be noted that for $t_{rsd} = 0$ the device acts like conventional SOI MOSFET and as a result the threshold voltage remains higher compared to the cases of $t_{rsd} = 30$ nm and 100nm. It may be attributed to the fact that at lower channel doping, back surface gets inverted at first if the channel length is too short. And, since the back channel is being partially controlled by source and drain extension, subthreshold characteristics of the device get deteriorated.

Figure 4.7 displays the threshold voltage variation against the gate-dielectric constant, where the device parameters are $\phi_M = 4.6 \text{eV}$, $N_a = 10^{16} \text{cm}^{-3}$, $N_d = 10^{20} \text{cm}^{-3}$, $t_{Si} = 8 \text{nm}$, $t'_{ox} = 1.5 \text{nm}$, $t_{box} = 200 \text{nm}$, $t_{rsd} = 30 \text{nm}$, $V_{DS} = 0.1 \text{V}$, L=45 nm. The threshold voltage of the device is found to be decreasing with increase in gate-dielectric constant. This observation is in line with Fig. 4.4, where minimum of surface potential was found to be increasing with increase in the gate-dielectric constant, ε_k . It should be noted that the source channel barrier height gets lower with increase in the minimum of the surface potential and as a result charge carrier from the source may be easily injected into the channel region which is an indication of lowering of the threshold voltage.



Fig. 4.5: Threshold voltage versus channel length for different gate oxide thicknesses



Fig. 4.6: Threshold voltage versus channel length for different recessed oxide thicknesses



Fig. 4.7: Threshold voltage versus gate-dielectric constant (ε_k)

4.6.3 Subthreshold Current

Chapter 4

Figure 4.8 displays the subthreshold current against the gate-to-source voltage for different channel lengths keeping other device parameters constant. It is observed that the subthreshold leakage current is found to be higher at shorter channel length for a fixed gate voltage. This may be attributed to the increased dominance of source/drain over channel electric charges with the decrease in channel length. It should be noted that the model is developed for subthreshold regime of device operation only, and therefore the model results deviate from the numerical simulation results at higher V_{GS} (above threshold).

The effect of gate oxide thicknesses on the subthreshold current is examined in Fig. 4.9. The subthreshold current characteristics are plotted with gate-to-source voltages for two different effective oxide thicknesses (t_{ox} =1 and 1.5nm). It is observed that a thinner gate oxide renders lower subthreshold current with better gate electrostatics in the channel

region. In Fig. 4.10, the variations of subthreshold current against gate to source voltage have been shown for different gate-dielectric constants. It can be observed from the figure that for the same amount of equivalent oxide thickness, the high-*k* gate-dielectric materials cause slightly higher subthreshold leakage current compared to silicon dioxide. The reason of the same may be as follows. For the same effective oxide thickness a high-*k* dielectric material will be thicker by a factor of $\varepsilon_k / \varepsilon_{siO_2}$ compared to the SiO₂. And, a thicker dielectric layer results in an increase in the fringing electric field lines from the bottom of the gate electrode to the source and drain regions.



Fig. 4.8: Subthreshold current versus gate to source voltage for different channel lengths.



Fig. 4.9: Subthreshold current versus gate to source voltage for different oxide thicknesses.



Fig. 4.10: Subthreshold current versus gate to source voltage for varying gate-dielectric constants.

4.6.4 Subthreshold Swing (*S*)

Chapter 4

Figures 4.11 and 4.12 show the dependence of the subthreshold swing on the channel length (*L*) for different channel thicknesses (t_{Si}) and gate oxide thickness (t_{ox}). The switching characteristics of the device are found to be deteriorated with decreasing channel length. The results indicate that for the Re-S/D SOI MOSFET with high-*k* gate-dielectric material for a fixed channel length (*L*), the subthreshold swing characteristics can be improved by using the thinner channel thickness (t_{Si}) and effective gate oxide thickness (t_{ox}) of the device. This is due to the fact that a thin layer of front-gate oxide and channel thickness improves the gate control over the channel by strengthening the vertical electric field across the channel.

Fig. 4.13 demonstrates the subthreshold swing variation for different recessed source/drain thicknesses keeping all other parameters constant. The subthreshold swing is found to be increased with the increasing recessed-source/drain thicknesses. This increase in the subthreshold swing may be attributed to the coupling of the back-surface of the channel region to the extended source and drain through the buried-oxide.

The influence of gate-dielectric permittivity on subthreshold swing is shown in Fig. 4.14 by keeping all other device parameters constant. A device with high gate-dielectric permittivity is found to possess high subthreshold swing parameter value. However, this slight increase in subthreshold swing may not be a severe problem for the devices with high-*k* gate-dielectric materials which significantly controls the gate-leakage currents.



Fig. 4.11: Subthreshold swing variation with device channel length for different Si film thicknesses



Fig. 4.12: Subthreshold swing variation with device channel length for different oxide thicknesses



Fig. 4.13: Subthreshold swing variation with device channel length for different recessed source/drain thicknesses



Fig. 4. 14: Subthreshold swing variation with device channel length for different gate-dielectric constants.

4.7 Summary and Conclusion

Chapter 4

In this chapter, a detailed study of the subthreshold characteristics is presented for fullydepleted recessed-source/drain (Re-S/D) SOI MOSFETs with high-k gate-dielectric material. The 2D surface potential is formulated by solving two-dimensional Poisson's equation in the channel region using appropriate boundary conditions. Subsequently, the modeling of other subthreshold characteristics such as threshold voltage, subthreshold current and subthreshold swings are presented for the device. The present models include the effect of high-k gate-dielectric induced fringing field lines in terms of a capacitance C_{bot} , which is originated due to the electrostatic interaction between the bottom of the gate-electrode and source/drain. However, due to the thick spacer between source and drain the other parasitic capacitances C_{pp} and C_{top} have been neglected in the present models. It is observed that decrease in the gate oxide thickness and channel thickness reduces the threshold voltage roll-off, subthreshold current and subthreshold swing with better gate electrostatics in the channel region. A slight degradation of subthreshold characteristics with high-k gate-dielectric materials is observed due to the induced fringing field lines from gate electrode to source/drain. It is found that the present models predict the threshold voltage, subthreshold current and subthreshold swing of Re-S/D SOI MOSFETs correctly in a wide range of gatedielectric materials and other device parameters. Furthermore, the obtained models don't need iterative calculations; hence these models are convenient for fast evaluation of the short-channel device design criteria. Thus, this model provides an efficient tool for the design of high-k gate-dielectric Re-S/D SOI MOSFETs including the effects of parasitic internal fringe capacitance. A reasonably good matching between the model results and ATLASTM simulation data confirms the validity of the proposed subthreshold characteristic model of the Re-S/D SOI MOSFET with high-k gatedielectric material over a wide variation in device parameters.

Chapter 5

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFET with Back-Gate Control

5.1 Introduction

In Chapters 3 and 4, the modeling and simulation of subthreshold characteristics of short-channel DMG Re-S/D SOI MOSFET and Re-S/D SOI MOSFET with high-*k* gatedielectric material were presented respectively. In the present chapter, an attempt has been made to analytically analyze the subthreshold characteristics of short-channel Re-S/D SOI MOSFET with back-gate control by considering the 2D effects in both channel region and buried oxide (BOX) layer. It has been discussed in Chapter 2 that the short-channel effects are well suppressed in a FD SOI MOSFET, when the silicon channel thickness (t_{si}) is less than or equal to one-fourth of the channel length (L). However, in sub-30nm channel length regime, t_{si} needs to be scaled below 5nm; therefore, the large parasitic series resistance and threshold voltage sensitivity to t_{si} variation become serious issues [Noguchi *et al.* (2001), Chan *et al.* (1994)]. The extended source and drain offer low source and drain resistance, respectively, and facilitate better source and drain contacts and consequently decrease the total series resistance [Thean *et al.* (2006)]. On the other hand, it is found from the literature survey that short- channel devices severely suffer with threshold voltage fluctuation. Hence, to suppress the threshold voltage fluctuation in short-channel length devices, the application of back-gate bias is necessary [Yang *et al.* (1995), Numata *et al.* (2004), Majumdar *et al.* (2009)]. Thus, the presence of a back-gate in FD Re-S/D SOI MOSFETs could be a potential option to improve the device performance in terms of large threshold voltage (V_{th}) controllability, less parasitic series resistance, on-current (I_{on}) improvement, and off-current (I_{off}) suppression as well [Majumdar *et al.* (2009)]. Keeping the above facts in view, an attempt has been made in this chapter to present a theoretical and simulation-based study of the potential distribution, threshold voltage, subthreshold swing and subthreshold current of the FD Re-S/D SOI MOSFETs with back-gate control. The layout of the present chapter is outlined as follows.

In section 5.2, the 2D surface potential function of the short-channel FD Re-S/D SOI MOSFET device with back-gate control is derived by solving the 2D Poisson's equation using parabolic approximation method. Section 5.3 presents the analytical threshold voltage modeling of the device. The analytical subthreshold current and subthreshold swing models are developed in sections 5.4 and 5.5 respectively. Results and discussion are presented in section 5.6. Finally, the summary and conclusion of the present chapter are presented in section 5.7.

5.2 Modeling of Two-Dimensional (2D) Surface Potential

The schematic structure of a Re-S/D UTB SOI MOSFET with back-gate control is shown in Fig. 5.1, where, *L* is the gate length, t_{Si} is the Si film thickness, t_{ox} is the gate oxide thickness and t_{box} is the buried oxide thickness. N_a , N_d and N_{sub} represent channel, source/drain, and substrate doping concentrations respectively. V_{Gs} , V_{DS} and V_{BG} are the gate to source, drain to source, and back-gate voltages, respectively. The work functions of front-gate and back-gate materials are considered to be ϕ_{FM} and ϕ_{BM} respectively. The x- and y-axes of the 2D structure are considered to be along the channel-gate oxide interface and the source-channel interface respectively, as shown in Fig. 5.1.

Chapter 5

It is important to mention that the Re-S/D SOI structures of Chapters 3 and 4 were shown with the p^+ substrate. This was deliberately done to simplify the analytical modeling, although, the Si wafers, which are available in the market, are generally lightly doped. A lightly doped p substrate may not provide an ideal back-gate contact and builds the substrate induced surface potential (SISP) effects [Imam et al. (1999), Kumar et al. (2014)]. The SISP may alter the threshold characteristics of the device. Therefore, the back-gate control or biasing may be avoided if the device is fabricated on p substrate. However, a highly doped p^+ substrate may cut down the SISP effects significantly. In the present case, since we have considered a separate back-gate on the substrate as done in Numata et al. (2004), and Fasarakis et al. (2014) the substrate doping really does not matter much and the substrate only provide mechanical support to the structure. Furthermore, it has been discussed in Chapter 2 that 2D effects in the BOX layer could not be neglected for short-channel devices with back-gate control. Thus, in the present chapter, we have considered the 2D effects in both channel region and buried oxide (BOX) layer to model a short-channel Re-S/D UTB SOI MOSFET with back-gate control. The potentials in the channel region and BOX are named as $\phi_1(x, y)$ and $\phi_2(x, y)$ respectively. The other parameters used in modeling and simulation are detailed in Table 5.1.


Fig. 5.1: The schematic structure of the Re-S/D FD SOI MOSFET with back-gate control

Parameters	Symbol	values
Front-gate work-function	$\phi_{_{FM}}$	4.71eV, 4.8eV
Back-gate work-function	$\phi_{\scriptscriptstyle BM}$	4.71eV, 4.8eV
Channel Doping	N _a	10^{15}cm^{-3}
Source/Drain Doping	N_d	10^{20}cm^{-3}
Substrate Doping	N _{sub}	10^{15}cm^{-3}
Silicon Thickness	t _{si}	6-10nm
Gate Oxide Thickness	t _{ox}	1.5-3nm
Buried Oxide Thickness	t _{box}	40-80nm
The depth of S/D in the buried oxide	t _{rsd}	0- 25nm
Recessed Length	d_{box}	3nm
Channel Length	L	20-120nm
Front-gate Voltage	V_{GS}	-0.2-0.6V
Drain Voltage	V_{DS}	0.05V
Back-gate Voltage	V_{BG}	-5 to 2V

Table 5. 1: Device parameter values used for modeling and simulation of a Re-S/D FD SOIMOSFET with back-gate control

The 2D Poisson's equations for potential in the channel region and BOX are

$$\frac{\partial^2 \phi_1(x, y)}{\partial x^2} + \frac{\partial^2 \phi_1(x, y)}{\partial y^2} = \frac{qN_a}{\varepsilon_{si}} \qquad \text{for } 0 \le y \le t_{si}$$
(5.1)

$$\frac{\partial^2 \phi_2(x, y)}{\partial x^2} + \frac{\partial^2 \phi_2(x, y)}{\partial y^2} = 0 \qquad \text{for } t_{Si} \le y \le t_{Si} + t_{box} \tag{5.2}$$

The potential distribution in the channel region and BOX are approximated by the following polynomials [Young *et al.* (1989)]

$$\phi_1(x, y) = \phi_f(x) + C_{11}(x)y + C_{12}(x)y^2 \qquad \text{for } 0 \le y \le t_{Si}$$
(5.3)

$$\phi_2(x, y) = C_{20}(x) + C_{21}(x)y + C_{22}(x)y^2 \qquad \text{for } t_{Si} \le y \le t_{Si} + t_{box} \qquad (5.4)$$

where, $\phi_f(x) = \phi_1(x,0)$ is the surface potential at SiO₂/Si interface and the arbitrary coefficients $C_{11}(x)$, $C_{12}(x)$, $C_{20}(x)$, $C_{21}(x)$ and $C_{22}(x)$ are the functions of x only. These coefficients are to be determined using the following boundary conditions:

The continuity of electric flux at SiO₂/Si interface is formulated as:

$$\left[\frac{\partial\phi_1(x,y)}{\partial y}\right]_{y=0} = \frac{\varepsilon_{ox}}{\varepsilon_{Si}} \frac{\phi_f(x) - V'_{GS}}{t_{ox}} = \frac{\phi_f(x) - V'_{GS}}{t_{ox}.\gamma}$$
(5.5)

The electric flux at Si/BOX interface is also continuous as:

$$\left[\frac{\partial\phi_1(x,y)}{\partial y}\right]_{y=t_{Si}} = -\frac{E_b}{\gamma} + \frac{C_{rsd}}{\varepsilon_{Si}L} \left[V'_{DS} - 2\phi_b(x)\right]$$
(5.6)

The electric field at the bottom of the channel region is:

$$\left[\frac{\partial \phi_2(x, y)}{\partial y}\right]_{y=t_{Si}} = -E_b + \frac{V'_{DS} - 2\phi_b(x)}{t^*_{rsd}}$$
(5.7)

The surface potential at the source end is

$$\phi_1(0, y) = V_{bi} \tag{5.8}$$

The surface potential at the drain end is

$$\phi_1(L, y) = V_{bi} + V_{DS}$$
(5.9)

The surface potential at the back-gate is

$$\phi_2(x, y = t_{Si} + t_{ox}) = V'_{BG}$$
(5.10)

where,
$$\gamma = \frac{\varepsilon_{Si}}{\varepsilon_{ox}}$$
, $V'_{GS} = V_{GS} - V_{FB1}$, $V'_{DS} = V_{DS} - 2V_{FB2}$, and $V'_{BG} = V_{BG} - V_{FB3}$; V_{FB1} , V_{FB2}

and V_{FB3} represent the flat-band voltages which are given as $V_{FB1} = \phi_{FM} - \phi_{Si}$, $V_{FB2} = V_T ln \left(\frac{N_a N_d}{n_i^2} \right)$ and $V_{BM} = \phi_{BM} - \phi_{Si}$; The symbol ϕ_{Si} represents the silicon body

work function and given by $\phi_{Si} = \chi_{Si} + \frac{E_g}{2} + V_T \ln\left(\frac{N_a}{n}\right)$, χ_{Si} and E_g are the electron affinity and energy band gap of the silicon, respectively. N_d is the doping concentration, n_i is the intrinsic carrier concentration in the silicon, V_{bi} is the built-in voltage at the source/drain and Si interface given by $V_{bi} = V_T ln \left(\frac{N_a N_d}{n_c^2}\right), V_T$ is the thermal voltage, V_{DS} , V_{GS} and V_{BG} are the drain to source, gate to source and back-gate bias voltages, respectively. E_b represents the BOX electric field component at Si/BOX interface. The surface back potential at Si/BOX interface is $\phi_b(x) = \phi_1(x, y = t_{Si}) = \phi_2(x, y = t_{Si})$. C_{rsd} is the recessed source/drain buried oxide capacitance per unit length which is given as [Svilicic et al. (2009)]

$$C_{rsd} = \begin{cases} \frac{\varepsilon_{box}}{\left(\frac{\pi}{2} + \frac{\pi}{2} \cdot \sec h \frac{t_{rsd}}{d_{box}}\right)} ln \left(1 + \frac{L}{2d_{box}}\right) & for \frac{L}{2} < t_{rsd} \\ \frac{\varepsilon_{box}}{\left(\frac{\pi}{2} + \frac{\pi}{2} \cdot \sec h \frac{t_{rsd}}{d_{box}}\right)} ln \left(1 + \frac{t_{rsd}}{2d_{box}}\right) & for \frac{L}{2} > t_{rsd} , t_{rsd} = 0 \end{cases}$$

$$(5.11)$$

where, d_{box} is the length of source/drain overlap over buried oxide, $t_{rsd}^* = \frac{\varepsilon_{box}L}{C_{rsd}}$ is the

effective thickness of source/drain extensions in buried oxide.

By using the boundary conditions of Eqns. (5.5) and (5.6) in Eq. (5.3), the coefficients $C_{11}(x)$ and $C_{12}(x)$ can be written as

$$C_{11}(x) = \frac{\phi_f(x) - V'_{GS}}{t_{ox} \cdot \gamma}$$
(5.12)

$$C_{12}(x) = \frac{V_{GS}' - \phi_f(x)}{2t_{Si}t_{ox}.\gamma} - \frac{E_b}{2t_{Si}\gamma} + \frac{C_{rsd}}{2t_{Si}\varepsilon_{Si}L} [V_{DS}' - \phi_b(x)]$$
(5.13)

Now, the expression for $\phi_1(x, y)$ is obtained by substituting the values of $C_{11}(x)$ and $C_{12}(x)$ into Eq. (5.3). Thus, the potential $\phi_1(x, y)$ can be written as $\phi_1(x, y) = \phi_f(x) + \frac{\phi_f(x) - V'_{GS}}{t_{ox} \cdot \gamma} y + \left\{ \frac{V'_{GS} - \phi_f(x)}{2t_{Si} t_{ox} \cdot \gamma} - \frac{E_b}{2t_{Si} \gamma} + \frac{C_{rsd}}{2t_{Si} \varepsilon_{Si} L} \left[V'_{DS} - \phi_b(x) \right] \right\} y^2$ (5.14)

Substituting $y = t_{Si}$ in Eq. (5.14) gives

$$\phi_{b}(x) = \frac{C_{Si}}{C_{Si} + C_{rsd}} \left\{ \phi_{f}(x) \left[1 + \frac{t_{Si}}{2t_{ox}} \cdot \gamma \right] - \frac{t_{Si}}{2t_{ox}} V_{GS}' - \frac{t_{Si}}{2\gamma} E_{b} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right\}$$
(5.15)

where, $C_{Si} = \frac{\varepsilon_{Si}L}{t_{Si}}$ is the silicon-body capacitance per channel width [Svilicic *et al.* (2009)]

With the help of Eq. (5.15), Eq (5.14) can be expressed in terms of $\phi_b(x)$ as

$$\phi_{1}(x, y) = V_{GS}' + \frac{2\gamma_{ox}}{t_{Si} + 2\gamma_{ox}} \left[\phi_{b}(x) \left(\frac{C_{Si} + C_{rsd}}{C_{Si}} \right) - V_{GS}' + \frac{E_{b}t_{Si}}{2\gamma} - \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right]$$

$$+ \frac{2}{t_{Si} + 2\gamma_{ox}} \left[\phi_{b}(x) \left(\frac{C_{Si} + C_{rsd}}{C_{Si}} \right) - V_{GS}' + \frac{E_{b}t_{Si}}{2\gamma} - \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right] y$$

$$+ \left\{ \frac{1}{t_{Si}(t_{Si} + 2\gamma_{ox})} \left[-\phi_{b}(x) \left(\frac{C_{Si} + C_{rsd}}{C_{Si}} \right) + V_{GS}' - \frac{E_{b}(t_{Si} + \gamma_{ox})}{\gamma} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right] \right\} y^{2}$$

$$+ \left\{ \frac{1}{t_{Si}(t_{Si} - 2\gamma_{ox})} \left[-\phi_{b}(x) \left(\frac{C_{Si} + C_{rsd}}{C_{Si}} \right) + V_{GS}' - \frac{E_{b}(t_{Si} + \gamma_{ox})}{\gamma} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right] \right\} y^{2}$$

$$+ \left\{ \frac{1}{t_{Si}(t_{Si} - 2\gamma_{ox})} \left[-\phi_{b}(x) \left(\frac{C_{Si} + C_{rsd}}{C_{Si}} \right) + V_{GS}' - \frac{E_{b}(t_{Si} - \gamma_{ox})}{\gamma} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right] \right\} y^{2}$$

Chapter 5

Furthermore, the arbitrary constants $C_{20}(x)$, $C_{21}(x)$ and $C_{22}(x)$ of Eq. (5.4) can be obtained by using the boundary conditions Eqns. (5.7) and the surface potentials at $y = t_{Si}$ and $y = t_{Si} + t_{box}$ in Eq. (5.4):

$$C_{20}(x) = \frac{t_{box}^{2} - t_{Si}^{2}}{t_{box}^{2}} \left(-V_{BG}' + \phi_{b}(x)\right) + V_{BG}' + \frac{t_{Si}(t_{Si} + t_{box})}{t_{box}} E_{b} - \frac{t_{Si}(t_{Si} + t_{box})}{t_{rsd}^{*} t_{box}} \left(V_{DS}' - 2\phi_{b}(x)\right)$$
(5.17)

$$C_{21}(x) = \frac{2t_{Si}}{t_{box}^{2}} \left(-V_{BG}' + \phi_{b}(x) \right) - \frac{\left(2t_{Si} + t_{box}\right)}{t_{box}} E_{b} + \frac{\left(2t_{Si} + t_{box}\right)}{t_{box}^{*}} \left(V_{DS}' - 2\phi_{b}(x)\right)$$
(5.18)

$$C_{22}(x) = \frac{-1}{t_{box}^{2}} \left(-V_{BG}' + \phi_{b}(x) \right) + \frac{1}{t_{box}} E_{b} - \frac{1}{t_{rsd}^{*} t_{box}} \left(V_{DS}' - 2\phi_{b}(x) \right)$$
(5.19)

Now, the expression $\phi_2(x, y)$ can be obtained by substituting the values of $C_{20}(x)$, $C_{21}(x)$ and $C_{22}(x)$ into Eq. (5.4).

$$\phi_{2}(x, y) = \frac{t_{box}^{2} - t_{Si}^{2}}{t_{box}^{2}} \left(-V_{BG}' + \phi_{b}(x) \right) + V_{BG}' + \frac{t_{Si}(t_{Si} + t_{box})}{t_{box}} E_{b} - \frac{(t_{Si} + t_{box})t_{Si}}{t_{rsd}^{*}t_{box}} \left(V_{DS}' - 2\phi_{b}(x) \right)$$

$$+ \left[\frac{2t_{Si}}{t_{box}^{2}} \left(-V_{BG}' + \phi_{b}(x) \right) - \frac{(2t_{Si} + t_{box})}{t_{box}} E_{b} + \frac{(2t_{Si} + t_{box})}{t_{box}} \left(V_{DS}' - 2\phi_{b}(x) \right) \right] y$$

$$+ \left[\frac{-1}{t_{box}^{2}} \left(-V_{BG}' + \phi_{b}(x) \right) + \frac{1}{t_{box}} E_{b} - \frac{1}{t_{rsd}^{*}t_{box}} \left(V_{DS}' - 2\phi_{b}(x) \right) \right] y^{2}$$

$$(5.20)$$

Using the Eq. (5.20) into Eq. (5.2), the BOX electric field component at Si/BOX interface can be obtained as

$$E_{b} = \frac{1}{t_{box}} \left(-V_{BG}' + \phi_{b}(x) \right) + \frac{\left(V_{DS}' - 2\phi_{b}(x) \right)}{t_{rsd}^{*}} - \frac{t_{box}}{2} \frac{d^{2}\phi_{b}(x)}{dx^{2}}$$
(5.21)

Now, the second-order differential equation of back-surface potential, $\phi_b(x)$ can be obtained by solving the Eq. (5.1) using Eqns. (5.16) and (5.21):

$$\frac{d^2\phi_b(x)}{dx^2} + \alpha_b\phi_b(x) = \beta_b$$
(5.22)

where, α_b and β_b are

$$\alpha_{b} = \frac{\frac{-2}{t_{Si}(t_{Si} + 2\gamma_{ox})} \left[\frac{C_{Si} + C_{rsd}}{C_{Si}} + \frac{(t_{Si} + \gamma_{ox})}{t_{box}\gamma} - 2\frac{(t_{Si} + \gamma_{ox})}{\gamma_{rsd}^{*}} + \frac{C_{rsd}}{C_{Si}t_{Si}}(t_{Si} + 2\gamma_{ox}) \right]}{\left[1 + \frac{t_{box}(t_{Si} + \gamma_{ox})}{t_{Si}(t_{Si} + 2\gamma_{ox})\gamma} \right]}.$$
(5.23)

$$\beta_{b} = \frac{\left[\frac{qN_{a}}{\varepsilon_{si}} - \left[\frac{2}{(t_{si} + 2\gamma_{ox})t_{si}}\right]V_{GS}' - \frac{2}{t_{si}\gamma}\left(\frac{t_{si} + \gamma_{ox}}{t_{si} + 2\gamma_{ox}}\right)\left(\frac{V_{BG}'}{t_{box}} - \frac{V_{DS}'}{t_{rsd}'}\right)\right] + \frac{C_{rsd}}{C_{si}t_{si}}\left(\frac{2}{t_{si} + 2\gamma_{ox}} + \frac{1}{t_{si}}\right)V_{DS}'\right]}{\left[1 + \frac{t_{box}(t_{si} + \gamma_{ox})}{t_{si}(t_{si} + 2\gamma_{ox})\gamma}\right]}$$
(5.24)

By solving the second-order differential equation (5.22), the following expression for back surface potential distribution along the channel can be obtained as

$$\phi_b(x) = A_b \cdot e^{\sqrt{\alpha_b}x} + B_b \cdot e^{-\sqrt{\alpha_b}x} - \frac{\beta_b}{\alpha_b}$$
(5.25)

Now, the coefficients A_b and B_b of Eq. (5.22) are determined using the boundary conditions of Eqns. (5.8) and (5.9)

$$A_{b} = \frac{\beta_{b} \left(e^{\sqrt{\alpha_{b}L}} - 1 \right) + \alpha_{b} \left[V_{bi} \left(e^{\sqrt{\alpha_{b}L}} - 1 \right) + V_{DS} e^{\sqrt{\alpha_{b}L}} \right]}{\alpha_{b} \left(e^{2\sqrt{\alpha_{b}L}} - 1 \right)}$$
(5.26)

$$B_{b} = \frac{e^{\sqrt{\alpha_{b}L}} \left\{ \beta_{b} \left(e^{\sqrt{\alpha_{b}L}} - 1 \right) - \alpha_{b} \left[V_{bi} \left(1 - e^{\sqrt{\alpha_{b}L}} \right) + V_{DS} \right] \right\}}{\alpha_{b} \left(e^{2\sqrt{\alpha_{b}L}} - 1 \right)}$$
(5.27)

In order to obtain the front-surface potential, $\phi_f(x)$ the Eq. (5.15) has been differentiated twice with respect to x

$$\frac{d^{2}\phi_{f}(x)}{dx^{2}} = \frac{2\gamma_{ox}}{t_{Si} + 2\gamma_{ox}} \left(\frac{C_{Si} + C_{rsd}}{C_{Si}}\right) \frac{d\phi_{b}(x)}{dx^{2}} + \frac{t_{ox}t_{Si}}{(t_{Si} + 2\gamma_{ox})} \frac{d^{2}E_{b}}{dx^{2}}$$
(5.28)

The Eq. (5.28) is solved using Eq. (5.21), and the obtained expression is written as

$$\frac{d^{2}\phi_{f}(x)}{dx^{2}} = \frac{d^{2}\phi_{b}(x)}{dx^{2}} \frac{2\eta_{ox}}{t_{Si} + 2\eta_{ox}} \left[\left(\frac{C_{Si} + C_{rsd}}{C_{Si}} \right) + \frac{t_{Si}}{2\eta_{box}} - \frac{t_{Si}}{2\eta_{rsd}^{*}} \right] - \frac{\eta_{box}t_{ox}t_{Si}}{2\gamma(t_{Si} + 2\eta_{ox})} \frac{d^{4}\phi_{b}}{dx^{4}} \quad (5.29)$$

By differentiating the Eq. (5.22) twice with respect to x, we can obtain

$$\frac{d^4\phi_b(x)}{dx^4} = -\alpha_b \frac{d^2\phi_b(x)}{dx^2}$$
(5.30)

Using Eqns. (5.29) and (5.30) into Eq. (5.21), E_b can be obtained in terms of frontsurface potential, $\phi_f(x)$

$$E_{b} = \frac{1}{C_{1}} \begin{bmatrix} -\frac{V_{sub}'}{t_{box}} + \frac{V_{DS}'}{t_{rsd}} + \left(1 + \frac{t_{Si}}{2\eta_{box}}\right) \left(\frac{C_{Si}}{C_{Si} + C_{rsd}}\right) \left(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*}t_{box}}\right) \left(\phi_{f}(x) - \frac{V_{GS}'t_{Si}}{2\eta_{box}} + \frac{C_{rsd}}{2C_{Si}}V_{D}'\right) \\ -\frac{t_{box}}{2} \left(\frac{1}{C_{2}}\right) \frac{d^{2}\phi_{f}(x)}{dx^{2}} \end{bmatrix} (5.31)$$

$$C_{1} = 1 + \frac{t_{Si}}{2\gamma t_{box}} \left(\frac{C_{Si}}{C_{Si} + C_{rsd}} \right) \left(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*} t_{box}} \right)$$

$$C_{2} = \frac{2\gamma t_{ox}}{t_{Si} + 2\gamma t_{ox}} \left[\left(\frac{C_{Si} + C_{rsd}}{C_{Si}} \right) + \frac{t_{Si}}{2\gamma t_{box}} - \frac{t_{Si}}{2\gamma t_{rsd}^{*}} - \frac{t_{box} t_{Si} \alpha_{b}}{4\gamma} \right]$$
(5.32)

Using the Eqns. (5.14), (5.15) and (5.31), Eq. (5.1) can be solved to obtain the secondorder differential equation of front-surface potential, $\phi_f(x)$ which is given as

$$\frac{d^2\phi_f(x)}{dx^2} + \alpha_f\phi_f(x) = \beta_f$$
(5.33)

where, α_f and β_f are

$$\alpha_{f} = \frac{\frac{-1}{t_{Si} \mathcal{M}_{ox}} + \left(1 + \frac{t_{Si}}{2 \mathcal{M}_{ox}}\right) \left(\frac{C_{Si}}{C_{Si} + C_{rsd}}\right) \left[\frac{C_{3}}{C \mathcal{M}_{ox}} \left(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*} t_{box}}\right) - 2\right]}{1 + \frac{t_{box}}{2} \left(\frac{C_{3}}{C_{2}C_{1}}\right)}$$
(5.34)

$$C_{3} = 1 + \frac{C_{rsd}}{\varepsilon_{si}L} \left(\frac{C_{si}}{C_{si} + C_{rsd}}\right) \left(\frac{t_{si}}{2}\right)$$
(5.35)

$$\frac{qN_{a}}{\varepsilon_{Si}} - \frac{C_{3}}{C_{1}\gamma} \left(1 + \frac{t_{Si}}{2\gamma_{box}} \right) \left(\frac{C_{Si}}{C_{Si} + C_{rsd}} \right) \left(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*}t_{box}} \right) \left[-\frac{V_{GS}'t_{Si}}{2\gamma_{box}} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right] \\
+ \frac{C_{rsd}}{\varepsilon_{Si}L} V_{DS}' - 2\frac{C_{rsd}}{\varepsilon_{Si}L} \left[-\frac{V_{GS}'t_{Si}}{2\gamma_{box}} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right] - \frac{V_{GS}'}{\gamma_{Si}t_{box}} - \frac{C_{3}}{C_{1}\gamma} (C_{4}) \\
1 + \frac{t_{box}}{2} \left(\frac{C_{3}}{C_{2}C_{1}} \right)$$
(5.36)

$$C_{4} = \left(\frac{C_{Si} + C_{rsd}}{C_{Si}}\right) \left[-\frac{V_{sub}'}{t_{box}} - \frac{V_{GS}' t_{Si}}{2\gamma t_{ox}} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' \right]$$
(5.37)

By solving the second-order differential equation (5.33), the following expression for front-surface potential distribution along the channel can be obtained as

$$\phi_f = A_f \cdot e^{\sqrt{\alpha_f}x} + B_f \cdot e^{-\sqrt{\alpha_f}x} - \frac{\beta_f}{\alpha_f}$$
(5.38)

Now, the coefficients A_f and B_f of Eq. (5.38) are determined using the boundary conditions of Eqs. (5.8) and (5.9).

$$A_{f} = \frac{\beta_{f} \left(e^{\sqrt{\alpha_{f}}L} - 1 \right) + \alpha_{f} \left| V_{bi} \left(e^{\sqrt{\alpha_{f}}L} - 1 \right) + V_{DS} e^{\sqrt{\alpha_{f}}L} \right|}{\alpha_{f} \left(e^{2\sqrt{\alpha_{f}}L} - 1 \right)}$$
(5.39)

Chapter 5

$$B_{f} = \frac{e^{\sqrt{\alpha_{f}L}} \left\{ \beta_{f} \left(e^{\sqrt{\alpha_{f}L}} - 1 \right) - \alpha_{f} \left[V_{bi} \left(1 - e^{\sqrt{\alpha_{f}L}} \right) + V_{DS} \right] \right\}}{\alpha_{f} \left(e^{2\sqrt{\alpha_{f}L}} - 1 \right)}$$
(5.40)

The minimum of the front- and back-surface potential $\phi_{f\min,b\min}$ could be derived by following the same method used in Chapters 3 and 4, and the obtained expression is

$$\phi_{f\min,b\min} = 2\sqrt{A_{f,b}B_{f,b}} - \frac{\beta_{f,b}}{\alpha_{f,b}}$$
(5.41)

Now, the virtual cathode potential $\phi_{vc}(y)$ in terms of minimum front- and back- surface potentials similar to Chapters 3 and 4 [Chen *et al.* (2003)]

$$\begin{split} \phi_{vc}(y) &= \phi_{bmin} \begin{bmatrix} \frac{C_{Si} + C_{rsd}}{C_{Si}} \frac{1}{t_{Si} + 2\gamma_{ox}} \left(2\gamma_{ox} + 2y - \frac{1}{t_{Si}} y^{2} \right) \\ &+ \left(\frac{1}{t_{box}} - \frac{2}{t_{rsd}^{*}} + \alpha_{b} \frac{t_{box}}{2} \right) \left(t_{Si} t_{ox} + \frac{t_{Si}}{\gamma} y - \frac{t_{Si} + 2\gamma_{ox}}{t_{Si}} y^{2} \right) - \frac{C_{rsd}}{C_{Si} t_{Si}^{2}} y^{2} \end{bmatrix} \\ &+ \left(V_{GS}' \left[1 - \frac{1}{t_{Si} + 2\gamma_{ox}} \left(-2\gamma_{ox} - 2y + \frac{1}{t_{Si}} y^{2} \right) \right] \right] \\ &+ \left(V_{DS}' \right) \left\{ \frac{C_{rsd}}{C_{Si}} \left[\frac{-\gamma_{ox}}{t_{Si} + 2\gamma_{ox}} - \frac{1}{t_{Si} + 2\gamma_{ox}} y + \frac{1}{t_{Si}} \left(\frac{1}{t_{Si} + 2\gamma_{ox}} \right) y^{2} + \frac{1}{2t_{Si}^{2}} y^{2} \right] \right\} \\ &+ \left(-\frac{V_{DS}'}{t_{box}} - \beta_{b} \frac{t_{box}}{2} \right) \left(t_{Si} t_{ox} + \frac{t_{Si}}{\gamma} y - \frac{t_{Si} + 2\gamma_{ox}}{t_{Si}} y^{2} \right) \end{split}$$
(5.42)

$$\begin{split} \phi_{vc}(y) &= \phi_{f\min} \Biggl[\Biggl[\Biggl[1 + \frac{1}{\gamma_{ox}} y - \Biggl(\frac{1}{2t_{si}\gamma_{ox}} \Biggr] y^{2} + \frac{C_{5}}{C_{1}} \Biggl[1 + \frac{t_{box}}{2C_{2}} \alpha_{f} \Biggr] \Biggr] + \Biggr] \\ &- \frac{V_{GS}'}{\gamma_{ox}} y + \Biggl(\frac{1}{2t_{Si}} \Biggr) \Biggl[\frac{V_{GS}'}{\gamma_{ox}} + \frac{C_{rsd}}{\varepsilon_{si}L} V_{DS}' \Biggr] y^{2} + \frac{1}{C_{1}} \Biggl[\frac{-V_{sub}'}{t_{box}} + \frac{-V_{DS}'}{t_{rsd}^{*}} \Biggr] \\ &+ \frac{C_{5}}{C_{1}} \Biggl(\frac{-V_{GS}'t_{Si}}{2\gamma_{box}} + \frac{C_{rsd}}{2C_{Si}} V_{DS}' - \frac{t_{box}}{2} \Biggl(\frac{1}{C_{2}} \Biggr) \beta_{f} \Biggr] \\ C_{5} &= \Biggl(1 + \frac{t_{Si}}{2\gamma_{box}} \Biggr) \Biggl(\frac{C_{Si}}{C_{Si}} + C_{rsd}} \Biggr) \Biggl(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*} t_{box}} \Biggr) \end{split}$$
(5.43)

5.3 Threshold Voltage Formulation

The threshold voltage associated with front- and back-surfaces of the channel could be obtained using Kumar *et al.* (2014), similar to our consideration in Chapters 3 and 4.

$$\phi_{f\min,\min}\Big|_{V_G = V_{thf,b}} = 2\phi^*_{f,Si} = \begin{cases} 2\phi_{f,Si} & \text{for}, N_a > n_T \\ \phi_{f,Si} + \frac{kT}{q} \ln\left(\frac{n_T}{n_i}\right) & \text{for}, N_a < n_T \end{cases}$$
(5.44)

where, $\phi_{f,Si}$ is the Fermi potential, n_T is a critical concentration of electron in the channel to turn on the device [Chen *et al.* (2003), Svilicic *et al.* (2010) and Kumar *et al.* (2014)].

Now, after using Eq. (5.44) in Eq. (5.41), we get the following equations for backchannel threshold voltages V_{hb}

$$V_{thb} = \frac{-b_b + \sqrt{b_b^2 - 4a_b c_b}}{2a_b}$$
(5.45)

where,

$$a_b = u_{b1}u_{b2} - \left(\frac{m_b}{2\alpha_b}\right)^2 \tag{5.46}$$

$$c_{b} = v_{b1}v_{b2} - \phi_{f,Si}^{2} - \left(\frac{m_{b}}{2\alpha_{b}}\right)^{2} - \frac{n_{b}}{\alpha_{b}}\phi_{f,Si}$$
(5.47)

$$b_{b} = u_{b1}v_{b2} + u_{b2}v_{b1} - \frac{m_{b}n_{f,b}}{2\alpha_{b}^{2}} - \frac{m_{b}}{\alpha_{b}}\phi_{f,Si}$$
(5.48)

$$v_{b1} = \frac{n_b \left(e^{\sqrt{\alpha_b L}} - 1 \right) + \alpha_b \left[V_{bi} \left(e^{\sqrt{\alpha_b L}} - 1 \right) + V_{DS} e^{\sqrt{\alpha_b L}} \right]}{\alpha_b \left(e^{2\sqrt{\alpha_b L}} - 1 \right)}$$
(5.49)

$$v_{b2} = \frac{e^{\sqrt{\alpha_b L}} \left\{ n_b \left(e^{\sqrt{\alpha_b L}} - 1 \right) - \alpha_b \left[V_{bi} \left(1 - e^{\sqrt{\alpha_b L}} \right) + V_{DS} \right] \right\}}{\alpha_b \left(e^{2\sqrt{\alpha_b L}} - 1 \right)}$$
(5.50)

$$u_{b1} = \frac{m_b \left(e^{\sqrt{\alpha_b L}} - 1 \right)}{\alpha_b \left(e^{2\sqrt{\alpha_b L}} - 1 \right)}$$
(5.51)

$$u_{b2} = \frac{m_b e^{\sqrt{\alpha_b L}} \left(e^{\sqrt{\alpha_b L}} - 1 \right)}{\alpha_b \left(e^{2\sqrt{\alpha_{f,b} L}} - 1 \right)}$$
(5.52)

$$m_{b} = \frac{\left[\frac{-2}{(t_{Si} + 2\eta_{ox})t_{Si}}\right]V_{GS}'}{\left[1 + \frac{t_{box}(t_{Si} + \eta_{ox})}{t_{Si}(t_{Si} + 2\eta_{ox})\gamma}\right]}$$
(5.53)

$$n_{b} = \frac{\left\{\frac{qN_{a}}{\varepsilon_{Si}} - \frac{2}{t_{Si}\gamma} \left(\frac{t_{Si} + \gamma t_{ox}}{t_{Si} + 2\gamma t_{ox}}\right) \left(\frac{V_{BG}'}{t_{box}} - \frac{V_{DS}'}{t_{sd}^{*}}\right) + \frac{C_{rsd}}{C_{Si}t_{Si}} \left(\frac{2}{t_{Si} + 2\gamma t_{ox}} + \frac{1}{t_{Si}}\right) V_{DS}'\right\}}{\left[1 + \frac{t_{box}(t_{Si} + \gamma t_{ox})}{t_{Si}(t_{Si} + 2\gamma t_{ox})\gamma}\right]}$$
(5.54)

Similarly, we get the following equations for front-channel threshold voltages V_{thf}

$$V_{thf} = \frac{-b_f + \sqrt{b_f^2 - 4a_f c_f}}{2a_f}$$
(5.55)

where,

 $a_f = u_{f1}u_{f2} - \left(\frac{m_f}{2\alpha_f}\right)^2 \tag{5.56}$

$$c_{f} = v_{f1}v_{f2} - \phi_{f,Si}^{2} - \left(\frac{m_{f}}{2\alpha_{f}}\right)^{2} - \frac{n_{f}}{\alpha_{f}}\phi_{f,Si}$$
(5.57)

$$b_f = u_{f1}v_{f2} + u_{f2}v_{f1} - \frac{m_f n_f}{2\alpha_f^2} - \frac{m_f}{\alpha_f}\phi_{f,Si}$$
(5.58)

$$v_{f1} = \frac{n_f \left(e^{\sqrt{\alpha_f L}} - 1 \right) + \alpha_f \left[V_{bi} \left(e^{\sqrt{\alpha_f L}} - 1 \right) + V_{DS} e^{\sqrt{\alpha_f L}} \right]}{\alpha_f \left(e^{2\sqrt{\alpha_f L}} - 1 \right)}$$
(5.59)

$$v_{f2,b2} = \frac{e^{\sqrt{\alpha_f}L} \left\{ n_f \left(e^{\sqrt{\alpha_f}L} - 1 \right) - \alpha_f \left[V_{bi} \left(1 - e^{\sqrt{\alpha_f}L} \right) + V_{DS} \right] \right\}}{\alpha_f \left(e^{2\sqrt{\alpha_f}L} - 1 \right)}$$
(5.60)

$$u_{f1} = \frac{m_f \left(e^{\sqrt{\alpha_f L}} - 1 \right)}{\alpha_f \left(e^{2\sqrt{\alpha_f L}} - 1 \right)}$$
(5.61)

$$u_{f2} = \frac{m_f e^{\sqrt{\alpha_f L}} \left(e^{\sqrt{\alpha_f L}} - 1 \right)}{\alpha_f \left(e^{2\sqrt{\alpha_f L}} - 1 \right)}$$
(5.62)

$$m_{f} = \frac{\left\{\frac{C_{3}}{C_{1}\gamma}\left(1 + \frac{t_{Si}}{2\gamma_{box}}\right)\left(\frac{C_{Si}}{C_{Si} + C_{rsd}}\right)\left(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*}t_{box}}\right)\left[-\frac{t_{Si}}{2\gamma_{box}}\right] - 2\frac{C_{rsd}}{\varepsilon_{Si}L}\left[-\frac{t_{Si}}{2\gamma_{box}}\right] - \frac{1}{\gamma_{Si}t_{box}}\right\}}{1 + \frac{t_{box}}{2}\left(\frac{C_{3}}{C_{2}C_{1}}\right)}V_{GS}'(5.63)$$

$$\frac{qN_{a}}{\varepsilon_{Si}} - \frac{C_{3}}{C_{1}\gamma} \left(1 + \frac{t_{Si}}{2\gamma_{box}}\right) \left(\frac{C_{Si}}{C_{Si} + C_{rsd}}\right) \left(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*}t_{box}}\right) \left[\frac{C_{rsd}}{2C_{Si}}V_{DS}'\right] + \frac{C_{rsd}}{\varepsilon_{Si}L}V_{DS}' - 2\frac{C_{rsd}}{\varepsilon_{Si}L}\left[\frac{C_{rsd}}{2C_{Si}}V_{DS}'\right] - \frac{C_{3}}{C_{1}\gamma}(C_{4}) + \frac{t_{box}}{2}\left(\frac{C_{3}}{C_{2}C_{1}}\right)$$

$$(5.64)$$

The threshold voltage of the device depends on that surface (front or back) of the device which will have the higher value of minimum surface potential (lesser barrier height). The device threshold voltage can be given as

$$V_{th} = \begin{cases} V_{thf}, & for \quad \phi_{f,min} > \phi_{b,min} \\ V_{thb}, & for \quad \phi_{f,min} < \phi_{b,min} \end{cases}$$
(5.65)

Further, the quantum effects induced correction term ΔV_{th} is added in the threshold voltage model of Eq. (5.65) models like our previous consideration in Chapters 3 and 4 [Dort *et al.* (1992)].

5.4 Subthreshold Current Formulation

The subthreshold current is mainly dominated by the diffusion phenomenon and is proportional to the carrier concentration at the minimum surface potential position (virtual cathode) as calculated in section 5.3. Therefore, by employing the 2D surface potential function derived in section 5.2, the expression of subthreshold current can be written as follows:

$$I_{sub} = \frac{qD_n n_i^2}{L_e N_a} \left(1 - exp\left(-\frac{V_{DS}}{V_T}\right) \right) \int_0^{t_{Si}} exp\left(\frac{\phi_{vc}(y)}{V_T}\right) dy$$
(5.66)

where, $\phi_{vc}(y)$ is given in (5.42) and (5.43); the other parameters are given in section 3.4 of Chapter 3.

Now, Eq. (5.66) has been solved by following the method which is given in section 3.4 of Chapter 3, and the obtained final subthreshold current expression, I_{sub} is

$$I_{sub} = \frac{qD_n n_i^2}{L_e N_a} \left(1 - exp\left(-\frac{V_{DS}}{V_T}\right) \right) V_T \left[\frac{I_f}{E_f} + \frac{I_b}{E_b} \right]$$
(5.67)

where,

$$I_{f} = exp\left(\frac{\phi_{vc}(y=0)}{V_{T}}\right) - exp\left(\frac{\phi_{vc}(y=y_{m})}{V_{T}}\right)$$
(5.68)

$$I_{b} = exp\left(\frac{\phi_{vc}(y = y_{m})}{V_{T}}\right) - exp\left(\frac{\phi_{vc}(y = t_{Si})}{V_{T}}\right)$$
(5.69)

$$E_{f} = (\phi_{vc}(y=0) - \phi_{vc}(y=y_{m}))/y_{m}$$
(5.70)

$$E_b = (\phi_{vc}(y = y_m) - \phi_{vc}(y = t_{Si})) / (t_{Si} - y_m)$$
(5.71)

5.5 Subthreshold Swing Formulation

This section presents the modeling of subthreshold swing of short-channel Re-S/D SOI MOSFETs with back-gate control. Subthreshold swing (S) of Re-S/D SOI MOSFETs with back-gate control can be written as [Dey *et al.* (2008), Svilicic *et al.* (2010)]

$$S = \begin{cases} ln 10.V_T \cdot \left(\frac{d\phi_{f\min}}{dV_{GS}}\right)^{-1} for \phi_{f\min} > \phi_{b\min} \\ ln 10.V_T \cdot \left(\frac{d\phi_{b\min}}{dV_{GS}}\right)^{-1} for \phi_{b\min} > \phi_{f\min} \end{cases}$$
(5.72)

By substituting Eq. 5.41 into Eq. 5.72, the following subthreshold slope expression for Re-S/D SOI MOSFETs with back-gate control is obtained

$$S = \begin{cases} \frac{\ln 10 V_{T}}{\alpha_{f}} \left[\frac{A_{f} e^{\sqrt{\alpha_{f}L}} + B_{f}}{\sqrt{A_{f}B_{f}} \left(e^{\sqrt{\alpha_{f}L}} + 1 \right)^{-1}} \right] \left[\frac{K_{f}}{1 + \frac{t_{box}}{2} \left(\frac{C_{3}}{C_{2}C_{1}} \right)} \right] & for \phi_{f\min} > \phi_{b\min} \\ \frac{\ln 10 V_{T}}{\alpha_{b}} \left[\frac{A_{b} e^{\sqrt{\alpha_{b}L}} + B_{b}}{\sqrt{A_{b}B_{b}} \left(e^{\sqrt{\alpha_{b}L}} + 1 \right)^{-1}} \right] K_{b} & for \phi_{b\min} > \phi_{f\min} \end{cases}$$
(5.73)

where,

$$K_{f} = -\frac{C_{3}}{C_{1}\gamma} \left(\frac{C_{Si}}{C_{Si} + C_{rsd}} \right) \left(-\frac{t_{Si}}{2\gamma_{box}} \right) \left[1 - \left(1 + \frac{t_{Si}}{2\gamma_{box}} \right) \left(\frac{t_{rsd}^{*} - 2t_{box}}{t_{rsd}^{*} t_{box}} \right) \right] - 2\frac{C_{rsd}}{\varepsilon_{Si}L} \left(-\frac{t_{Si}}{2\gamma_{box}} \right) - \frac{1}{\gamma_{Si}t_{box}}$$

$$(5.74)$$

$$K_{b} = \frac{-2}{\gamma} \left[\frac{\gamma + 2\gamma_{box} + t_{Si}}{t_{Si}^{2} + (2\gamma + 1)t_{box}t_{Si} + \gamma t_{box}^{2}} \right]$$
(5.75)

5.6 **Results and Discussion**

In this section, we have compared the analytical results obtained from the proposed models with the numerical simulation data obtained by simulating the device structure under consideration with a commercially available 2D device simulator ATLASTM.

5.6.1 Surface Potential

Chapter 5

Figure 5.2 shows the variation of front-channel and back-channel surface potentials along the channel length direction for two different back-gate bias voltages (V_{BG} =0V, -2V). For $V_{BG}=0V$, it could be observed that the minimum of back-channel surface potential is higher than that of front-channel minimum surface potential. This is attributed to the strong coupling between extended-source/drain and back-channel which in turn causes an early inversion at the back-channel compared to the frontchannel. Since the inversion takes place at Si-BOX interface rather than Si-SiO₂ interface, the front-gate does not control the channel charge effectively. Subsequently, the device becomes prone towards the short-channel effects, which is one of the drawbacks of Re-S/D SOI MOSFETs. The above mentioned behavior of the Re-S/D SOI MOSFET has been reported in Chapters 3 and 4. However, a negative voltage at the back-gate lifts the minimum of front-surface potential higher than that of backsurface potential. And subsequently, the front-surface of the channel gets inverted before the back-surface, like the FD SOI MOSFETs. Thus, the introduction of back-gate in Re-S/D SOI MOSFETs is expected to improve the short-channel effects immunity of the device, which is investigated hereafter in this section.



Fig. 5.2: Surface potential along the channel length at front-channel and back-channel

5.6.2 Threshold Voltage

Chapter 5

Figure 5.3 displays the front-channel and back-channel threshold voltage along the channel length for different back-gate voltages. It also includes the corresponding numerical simulation data of threshold voltage obtained from ATLAS simulation. It could be observed that the back-channel threshold voltage results obtained from the developed model are in close agreement with the ATLAS simulation results for $V_{BG}=0$, and 1V. But in case of $V_{BG}=-1V$, the numerical simulation results match with the model results which are associated with the front-channel. This is in complete agreement with the observations of Fig. 5.2 where, for $V_{BG}=-1V$, the minimum of front-surface potential is higher than the same of back-surface potential resulting in the threshold voltage being associated with the front-channel. Further, the threshold voltage roll-off at a shorter channel length (L<60nm) is found to be lesser in case of negative back-gate bias voltage.

Figure 5.4 shows the variation in the front-channel threshold voltage due to variation in the Si-body thickness. It is observed that the roll-off increases at higher channel thickness due to the reduction in the controllability of the gate electrodes. Thus, thinner channel provides better control on short-channel effects.

In Fig. 5.5, front-channel threshold voltage is plotted against channel length at t_{rsd} = 0nm, 10nm, and 25nm to show the threshold voltage sensitivity against the source/drain depth in the BOX. The device with t_{rsd} = 0nm is nothing but an asymmetrical DG MOSFET, which shows better SCE immunity than recessed-S/D structure. From the figure, it can be observed that the threshold voltage is found to be decreased with the increase of source/drain penetration in the BOX. The decrease in the threshold voltage is attributed to the loss of the gate control and growing influence of the source and drain.

Figure 5.6 shows the dependence of the front-gate and back-gate threshold voltage on the back-gate bias voltage (V_{BG}) . It is observed from the figure that the threshold voltage V_{th} increases almost linearly with decreasing V_{BG} . Figure 5.6 also reflects the results of Fig. 5.3 i.e. the back-gate bias voltage effectively control the threshold voltage. For positive back-gate voltage, back-channel gets inverted before the frontchannel. However, a negative back-gate voltage causes peak of inversion charge density to shift back to the Si-SiO₂ interface.



Fig. 5.3: Threshold voltage along the channel length for various back-gate bias voltages



Fig. 5.4: Front-channel threshold voltage along the channel length for various silicon channel thickness



Fig. 5.5: Front-channel threshold voltage versus gate length for varying recessed oxide thickness



Fig. 5.6: Front- and back-threshold voltage versus back-gate potential

5.6.3 Subthreshold Current

Chapter 5

The subthreshold current (I_{Sub}) with the gate voltage for three different buried oxide thicknesses (t_{box}) is plotted in Fig. 5.7. It should be noted that the present subthreshold current model of Eq. (5.79) is based on only the diffusion phenomenon of current transport, and therefore, the subthreshold current can be observed in agreement with the simulation results below the threshold voltage of devices. It is observed from Fig. 5.7 that for the fixed values of V_{GS} the subthreshold current is decreased with the decrease in BOX thickness. This decrease of I_{Sub} using a thinner BOX is due to a higher impact of back-gate reverse bias on it. In Fig. 5.8, subthreshold current has been plotted against the applied gate voltage for different values of the channel lengths. The subthreshold current is observed to be increased with the reduction in the channel length due to the increased short-channel effects. It is also observed that decreasing the channel length of the device from 40nm to 30nm increases the subthreshold leakage current by two orders of magnitude. The subthreshold current (I_{Sub}) variation with the gate to source voltage V_{GS} for different recessed-source/drain thickness (t_{rsd}) is considered in Fig. 5.9. It is found that at $t_{rsd} = 0$ nm (conventional SOI with back-gate), I_{Sub} is ~ 2*10⁻¹⁰ A/µm and as t_{rsd} increases to 25nm, I_{Sub} becomes ~ 5*10⁻¹⁰ A/µm. It is due to the fact that a deeper recessed source/drain offers higher short-channel effects due to the stronger coupling between recessed-source/drain and channel region. Figure 5.10 displays the impact of different back-gate voltages over the subthreshold current variation. The subthreshold current is found to be decreased with decreasing back-gate voltage. This is because the reverse back-gate voltage decreases the front-surface barrier height, which causes the inversion layer at the front-surface of the channel like conventional SOI MOSFETs.



Fig. 5.7: Subthreshold current versus gate to source voltage for different buried oxide thickness.



Fig. 5.8: Subthreshold current versus gate to source voltage for different channel Lengths.



Fig. 5.9: Subthreshold current versus gate to source voltage for different recessed source/drain thicknesses

Analytical Modeling and Simulation of Subthreshold Characteristics of Short-Channel Fully-Depleted Recessed-Source/Drain SOI MOSFET with Back-Gate Control



Fig. 5. 10: Subthreshold current versus gate to source voltage for different back-gate voltage

5.6.4 Subthreshold Swing(*S*)

The variation of the subthreshold swing (S) against the channel length is shown in Fig. 5.11 for different channel thicknesses. It is observed that at L = 20nm and $t_{Si} = 8$ nm the subthreshold swing, S is found to be ~ 162mV/Dec. and as t_{Si} decreases to 6nm, S becomes ~ 128mV/Dev. i.e., the subthreshold swing is increasing rapidly with the increase in the silicon channel thickness.

Figure 5.12 shows the variation of subthreshold swing (S) against the device channel lengths for different BOX thicknesses. The parameter S is found to be increased with the shrinkage of channel length < 70nm resulting in the poor switching characteristics of the device. However, for a fixed gate-length, the switching characteristics are observed to be improved with decreasing values of gate oxide thickness. Since, the gate will be in better position to control the channel for smaller gate oxide thickness, the above results seem to be well justified. The impact of back-gate voltage on the subthreshold swing (S) is presented in Fig. 5.13. It is observed that the reverse back-gate bias voltage can improve switching characteristics immensely. As the reverse back-gate voltage decreases from -1V to -3V the switching characteristics are found to be improved by ~ 30mV/Dec by keeping the other parameters constant.



Fig. 5.11: Subthreshold swing variation with channel length for different silicon thicknesses.



Fig. 5. 12: Subthreshold swing variation with channel length for different buried oxide thickness



Fig. 5. 13: Subthreshold swing variation with channel length for different back-gate voltage.

5.7 Summary and Conclusion

Chapter 5

In this chapter, threshold voltage, subthreshold swing and subthreshold current models are formulated for short-channel back-gated Re-S/D SOI MOSFETs by considering the 2D effects in both channel region and buried oxide (BOX) layer. The variations in the threshold voltage, subthreshold current and subthreshold swing have been analyzed against the back-gate bias voltage, channel thickness, BOX thickness, gate oxide thickness and recessed source/drain thickness. It has been found that a negative backgate bias voltage shift the peak of inversion charge density from back-channel to the front-channel resulting in an excellent short-channel-effects immunity in the device. The V_{th} roll-off and subthreshold swing roll-up of the Re-S/D SOI MOSFET are found to be significantly decreased with reverse back-gate bias voltage compared to the positive or no potential at the back-gate. Hence, the use of negative bias voltage is a potential solution to reduce the short-channel effects in the Re-S/D SOI MOSFET. The excess short-channel effects owing to the recessed-source/drain thickness may also be

compensated by carefully applying the reverse back-gate bias voltage. Moreover, the model results are found to be in very good agreement with numerical simulation data obtained from ATLASTM.

Chapter 6

Conclusion and Future Scopes of Work

6.1 Introduction

The present dissertation reports the analytical modeling and simulation of subthreshold characteristics of short-channel fully-depleted (FD) recessed-source/drain (Re-S/D) SOI MOSFETs. In Re-S/D SOI MOSFETs, the source and drain are extended deeper into the buried oxide (BOX) region in order to reduce total series resistance and facilitate better source and drain contacts. Recessing source and drain deeper into the BOX improves the 'on' current of SOI MOSFETs. Further, CMOS performance boosters such as high-*k* gate-dielectric, multi-gate and back-gate etc. help to improve the on-current of the device. However, the subthreshold characteristics of Re-S/D SOI MOSFETs need to be analyzed rigorously as the extended source and drain may control the channel and, hence, diminish the subthreshold performance. Further, it is also interesting to observe how the presence of a performance booster affects the subthreshold behavior of the device. The present chapter has been devoted to summarize the works carried out in this dissertation and conclude the major observations which are already presented in the previous chapters of this dissertation.

6.2 Chapter-wise Summary and Conclusion

Chapter 1 presents a brief history of semiconductor devices. MOSFET scaling techniques and its effect on the performance of the CMOS devices have also been discussed. Various CMOS technology boosters including channel engineering and gate material engineering have been introduced. The problems associated with the multi-gate

field-effect transistors (MuGFETs) are briefly discussed. The salient features, challenges and remedies of FD ultra-thin SOI MOSFETs are also presented in Chapter 1. Finally, suggestions for further work have been outlined at the end of this chapter.

Chapter 2 presents a brief review of SOI MOSFETs. Some state-of-the-art research works on SOI MOSFETs have been briefly discussed. Some major findings of the literature review are given below:

- The SOI-MOSFET is a highly scalable structure. The device could be used to fabricate a highly dense integrated circuit as the device offers higher drive current, higher scalability etc. as compared to the conventional CMOS devices.
- A number of studies have shown that the SOI MOS structures with more than one material for gate contacts can provide better immunity to short-channel effects (SCEs) and hot-carrier effects (HCEs). Thus, the theoretical modeling of the subthreshold characteristics could be very important to study the switching characteristics of these devices.
- Analytical modeling of the subthreshold characteristics of Re-S/D SOI MOSFET with high-k gate-dielectric could be very useful for studying the fringing field effects on the threshold voltage, subthreshold swing and subthreshold current of the device.
- ➢ Further, there is an ample scope of developing analytical models for the subthreshold characteristics of Re-S/D SOI MOSFETs with back-gate control.

Chapter 3 presents the analytical modeling and simulation of the subthreshold characteristics such as threshold voltage, subthreshold current and swing for a short-channel dual-metal-gate (DMG) recessed-source/drain (Re-S/D) SOI MOSFET. The proposed models are validated by comparing the theoretical results with the simulation data obtained by using the commercially available ATLASTM device simulation software from Silvaco International.

The major observations are given as follows:

- ➤ It is observed that the back-channel minimum surface potential is greater than the same of front-channel surface potential by ~70 mV for ϕ_{M1} =4.8eV, ϕ_{M2} =4.6eV, L=60nm, L₁: L₂ = 1:1, t_{Si}=10nm, t_{ox}=2nm, t_{rsd}=30nm, t_{box} =100nm, d_{box} =3nm, N_a =10¹⁶cm⁻³, V_{DS} =0.1V, V_{GS} =0.1V. Therefore, the back-surface of the channel gets inverted before the front-surface. This trend is observed for undoped short-channel length devices.
- The reduction in the gate oxide, buried oxide and recessed S/D thicknesses effectively control the SCEs.
- It is found that the increase in the recessed source/drain thickness increases the SCEs due to a stronger coupling between extended source/drain and back-surface of the channel region. However, reducing the gate oxide and channel thicknesses compensate the increased SCEs effectively
- > The control to screen gate length ratio $(L_1 : L_2)$ may act as an additional controlling parameter which can modulate the subthreshold characteristics of DMG Re-S/D SOI MOSFETs.
- The increased threshold voltage roll-off, subthreshold swing and subthreshold current because of recessing source and drain deeper into the BOX can be suppressed by choosing the device parameters judiciously
- The developed models include the effect of various device parameters such as the thickness of gate oxide, channel, buried oxide and recessed source/drain, control to screen gate length ratio, and channel length.
- The model results are observed to be in good agreement with the simulation data, which confirms the validity of the proposed model.

Chapter 4 presents the modeling and simulation of the subthreshold characteristics of the Re-S/D SOI MOSFET with high-k gate-dielectric material. The developed models estimate the effect of fringing fields, due to thicker physical thickness of high-k

dielectrics, on the subthreshold characteristics of Re-S/D SOI MOSFETs. The effects of the gate-dielectric permittivity, spacer oxide permittivity, spacer width, gate length, and the width of device are incorporated in the model. The major observations of Chapter 4 are given as follows.

- Different high-k dielectrics such as HfSiO₄, HfO₂ and La₂O₃ were used as the insulators to examine the subthreshold characteristics of Re-S/D SOI MOSFET with high-k gate-dielectric material.
- The surface potential, which was obtained from 2D electrostatic potential, was modified by adding a potential V(x) in order to capture the effects of high-k dielectric induced fringing fields.
- A drop of ~30mV is observed in the threshold voltage when HfO₂ replaces SiO₂ as a gate dielectric with $\phi_M = 4.8 \text{eV}$, $N_a = 10^{16} \text{cm}^{-3}$, $N_d = 10^{20} \text{cm}^{-3}$, $t_{box} = 200 \text{nm}$, $t_{Si} = 8 \text{nm}$, $t_{rsd} = 30 \text{nm}$, $t_{ox} = 1.5 \text{nm}$, $V_{DS} = 0.1 \text{V}$, $V_{GS} = 0.1 \text{V}$
- It is also observed that decrease in the gate oxide thickness and channel thickness reduces the threshold voltage roll-off, subthreshold current and subthreshold swing with better gate electrostatics in the channel region.
- A slight degradation in subthreshold characteristics is observed with high-k gatedielectric materials due to the induced fringing field lines from gate electrode to source/drain.
- The present models predict the subthreshold characteristics such as threshold voltage, subthreshold current and subthreshold swing of Re-S/D SOI MOSFETs correctly for a wide range of gate dielectric materials and other device parameters.
- The model results are observed to be in good agreement with the simulation data, which confirms the validity of the proposed models.

Chapter 5 presents the modeling and simulation of the effects of back-gate engineering on the subthreshold characteristics of Re-S/D SOI MOS structures. The 2D effects of

back-gate bias voltage are considered in both channel region and buried oxide (BOX) layer. The developed models are tested for wide variations in parameters like back-gate voltage, the channel length, the silicon thickness, the front-gate oxide thickness, the buried oxide thickness and the recessed source/drain thickness. The main results of this chapter can be written as follows:

- As the BOX layer thickness is reduced, the control of the back-gate bias placed underneath the SOI layer over the channel is increased.
- It has been found that a negative back-gate bias voltage (> -0.5V) shifts the peak of inversion charge density from back-surface of the channel to the front-surface resulting in an excellent short-channel-effects immunity in the device.
- Moreover, the threshold voltage can be changed by changing reverse back-gate bias voltage. This dynamic threshold voltage control can be used to optimize the power verses delay trade-off.
- > Our investigations reveal that the threshold voltage roll-off is improved by ~55mV as the channel thickness t_{Si} reduces from 10nm to 6nm for $\phi_{FM} = 4.71 \text{eV}$, $\phi_{BM} = 4.71 \text{eV}$, $N_a = 10^{15} \text{cm}^{-3}$, $N_d = 10^{20} \text{cm}^{-3}$, $t_{box} = 40 \text{nm}$, $t_{rsd} = 25 \text{nm}$, $t_{ox} = 2 \text{nm}$, $V_{DS} = 0.05 \text{V}$, $V_{BG} = -1 \text{V}$ due to the improvement in the gate electrostatics on the channel.
- By varying the back-gate bias voltage from -1V to -3V the subthreshold current and subthreshold swing can be reduced by ~one order of magnitude and ~35mV/Dec, respectively by keeping the other parameters constant.
- The degradation of the threshold voltage, subthreshold current and subthreshold swing owing to the increased Re-S/D thickness can be compensated by applying a suitable negative bias voltage at the back-gate.
- ➤ The model results are in good agreement with the simulation data obtained by using the commercially available 2D device simulator ATLASTM.

6.3 Suggestions for Further Work

We know that research is a continuous process and no dissertation is complete in every respect.

The work carried out in this dissertation may be further in the following areas.

- It has been observed in the literature that the mobility of the charge carriers improves by a strain in the channel, hence modeling and simulation of the subthreshold characteristics of the short-channel Re-S/D strained-Si SOI MOSFETs could be done.
- Since the analog and radio frequency (RF) performance evaluation is very important for a device to check its suitability for RF circuit design and systemon- chip applications, the analog and RF analysis of Re-S/D SOI MOSFETs could be done in detail.
- A continuous drain current model valid in all regions of the device operation is required for short-channel Re-S/D SOI MOSFETs.
- The analytical models for various device capacitors could be presented in the subthreshold regime of the device operation
- Unified 2D models can be developed for studying the subthreshold characteristics of the Re-S/D SOI MOSFETs including the effects of double/triple materials for the gate electrode, high-k dielectric material, and back-gate biasing
- Experimental results can provide further confirmation of the efficacy of developed models.

References

- J. E. Lilienfeld, "Method and apparatus for controlling electric currents," January 1930, US Patent US 1745175.
- W. Shockley, "A unipolar field-effect transistor", Proc. IRE, vol. 40, pp. 1365-1376, 1952.
- D. Kahng and M. M. Atalla, "Silicon-silicon dioxide field induced surface devices," in *IRE-AIEE Solid-state Device Res. Conf., Carnegie Inst. of Technol., Pittsburgh, PA*, 1960.
- F. M. Wanlass and C. T. Sah, "Nanowatt logic using field-effect metal-oxide semiconductor triodes", *Tech. Dig. IEEE 1963 Int. Solid State Circuit Conf.*, pp. 32-33, 1963.
- J. S. Kilby, "Miniaturized electronic circuits," June 1964, US Patent US3138743A.
- G. E. Moore, "Cramming More Components onto Integrated Circuits," *Electronics*, pp. 114–117, April 19, 1965.
- R. H. Dennard, F. H. Gaensslen, H.-N. Yu, V. L. Rideout, E. Bassous, and A. R. LeBlanc, "Design of ion-implanted MOSFET',s with very small physical dimensions", *IEEE J. Solid-State Circuits*, vol. 9, pp. 256-268, 1974.
- S. Wong, D. Bradbury, and K. Chiu, "Elevated source/drain MOSFET", *IEDM Tech. Dig.*, pp. 634-637, 1984.
- R. H. Dennard and M. R. Wordeman, "MOSFET miniaturization-From one micron to the limits", *Physica*, vol. 129B, pp. 3-15, 1985.
- F. Balestra, S. Cristoloveanu, M. Benachir, J. Brini, and T. Elewa, "Double -gate silicon-on insulator transistor with volume inversion: A new device with greatly enhanced performance," *IEEE Electron Device Lett.*, vol. 8, pp. 410-12, 1987.
- C. T. Lee, and K. K. Young, "Submicrometer near-intrinsic thin-film SOI complementary MOSFETs," *IEEE Trans. Electron Devices*, vol. 36, pp. 2537-2547, 1989.
- M. Matloubian, "Smart body contact for SOI MOSFETs," *Proc. IEEE SOI Tech. Conf.*, pp.128-129, 1989.
- M. Yoshimi , H. Hazama , M. Takahashi , S. Kambayashi , T. Wada , K. Kato and H. Tango, "Two-Dimensional Simulation and Measurement of High-Performance MOSFET\'s Made on a Very Thin SOI Film", *IEEE Trans. Electron Devices*, vol. ED-36, pp. 493-503, 1989.
- K. K. Young, "Short-channel effect in fully depleted SOI MOSFETs", *IEEE Trans. Electron Devices*, vol. 36, pp. 399-402, 1989.
- R. H. Yan, A. Ourmazd, K. F. Lee, D. Y. Jeon, C. S. Rafferty, and M. R. Pinto, "Scaling the Si metal-oxide-semiconductor field-effect transistor into the 0.1 μm regime using vertical doping engineering", *Appl. Phys. Lett.*, vol. 59, pp. 3315-3317, 1991.
- M. J. Van Dort, P. H. Woerlee, A. J. Walker, C. A. Juffermans, and H. Lifka, "Influence of high substrate doping levels on the threshold voltage and the mobility of deep-
submicrometer MOSFET',s", IEEE Trans. Electron Devices, vol. 39, pp. 932-938, 1992.

- R. Yan, A. Ourmazd, and K. F. Lee, "Scaling the Si MOSFET: From bulk to SOI to bulk", *IEEE Trans. Electron Devices*, vol. 39, pp. 1704-1710, 1992.
- H. O. Joachim, Y. Yamaguchi, K. Ishikawa, Y. Inoue, and T. Nishimura, "Simulation and twodimensional analytical modeling of subthreshold slope in ultrathin-film SOI MOSFETs down to 0.1µm gate length", *IEEE Trans. Electron Devices*, vol. 40, no. 10, pp. 1812-1817, 1993.
- J. H. Sim and B. Kuo, "An analytical back-gate effect model for ultra-thin SOI CMOS devices", *IEEE Trans. Electron Devices*, vol. 40, pp. 755-765, 1993.
- M. Chan, F. Assaderaghi, S. A. Parke, and C. Hu, "Recessed-channel structure for fabricating ultrathin SOI MOSFET with low series resistance", *IEEE Electron Device Lett.*, vol. 15, pp. 22-24, 1994.
- S. R. Banna, P. C. H. Chan, P. K. Ko, C. T. Nguyen, and M. Chan, "Threshold voltage model for deep-submicrometer fully depleted SOI MOSFETs", *IEEE Trans. Electron Devices*, vol. 42, pp. 1949-55, 1995.
- M. Bruel, B. Aspar, B. Charlet, C. Maleville, T. Poumeyrol, A. Souble, A.J.A.- Herve, J.M. Lamure, T. Barge, F. Metral, and S. Trucchi, "Smart Cut: A promising new SOI material technology", Proc. IEEE Int. SOI Conf., pp. 178-179, 1995
- I. Yang, C. Vieri, A. Chandraksan, and D. Antoniadis, "Back gated CMOS on SOIAS for dynamic threshold control", *Proc. IEDM*, pp. 877-880, 1995.
- P. C. Yeh and J. G. Fossum, "Physical subthreshold MOSFET modeling applied to viable design of deep-submicrometer fully-depleted SOI low-voltage CMOS technology", *IEEE Trans. Electron Devices*, vol. 42, 9, pp. 1605-1613, 1995.
- M. Bruel, Bernard Aspar and Andre-Jacques Auberton-Hervé1 Smart-Cut: A New Silicon On Insulator Material Technology Based on Hydrogen Implantation and Wafer Bonding, Japanese Journal of Applied Physics, vol. 36, pp. 1636, 1997
- F. Deng, R. A. Johnson, P. M. Asbeck, and S. S. Lau, "Salicidation process using NiSi and its device application", *J. Appl. Phys.*, vol. 81, pp. 8047-8051, 1997.
- W. F. Brinkman , D. E. Haggan and W. W. Troutman, "A history of the invention of the transistor and where it will lead us", *IEEE Journal of Solid-State Circuits*, vol. 32, pp. 1858-1865, 1997
- M. Ieong, P. M. Solomon, S. E. Laux, W.-S. P. Wong, and D. Chidambarrao, "Comparison of raised and Schottky S/D MOSFETs using a novel tunneling contact model", *IEDM Tech. Dig.*, pp. 733-736, 1998.
- K. Izumi, "History of SIMOX material", MRS Bull., vol. 23, pp. 20-24, 1998.
- Y. Taur and T. H. Ning, Fundamentals of Modern VLSI Devices, Cambridge Univ. Press, 1998.

- H.-S. Wong, D. Frank, and P. Solomon, "Device design considerations for double-gate, ground-plane, and single-gated ultra-thin SOI MOSFETs at the 25 nm channel length generation", *Proc. Int. Electron Devices Meeting*, pp. 407, 1998.
- M. A. Imam, M. A. Osman, and A. A. Osman, "Threshold voltage model for deep-submicron fully depleted SOI MOSFETs with back gate substrate induced surface potential effects," *Microelectronics Reliability*, vol. 39, pp. 487-495, 1999.
- W. Long, H. Ou, J. M. Kuo, and K. K. Chin, "Dual material gate (DMG) field effect transistor ", *IEEE Trans. Electron Devices*, vol. 46, pp. 865-870, 1999.
- S. Tiwari, J. J. Wesler, A. Kumar, and S. Cohen, "Straddle gate transistors: High Ion/Ioff transistors at short gate lengths", *DRC Dig.*, pp. 26-27, 1999.
- E. Suzuki, K. Ishii, S. Kanemaru, T. Tsutsumi, T. Sekigawa, K. Nagai, and H. Hiroshima, "Highly suppressed short channel effects in ultrathin SOI n-MOSFETs," IEEE Trans. On Electron Device, Vol. 47, No. 2, pp. 453-357, Feb. 2000.
- L. Chang, et al., "Reduction of direct-tunneling gate leakage current in double-gate and ultrathin body MOSFETs", *IEDM Tech. Dig.*, pp. 99-102, 2001.
- S. Han, S. Chang, J. Lee, and H. Shin, "50 nm MOSFET with electrically induced source/drain (S/D) extensions", *IEEE Trans. Electron Devices*, vol. 48, 9, pp. 2058-2064, 2001.
- M. Koh, W. Mizubayashi, K. Ivamoto, H. Murakami, T. Ono, M. Tsuno, T. Mihara, K. Shibahara, S. Miyazaki, and M. Hirose, "Limit of gate oxide thickness scalling in MOSFETs due to apparent threshold voltage fluctuation introduced by tunneling leakage current", *IEEE Trans. Electron Devices*, vol. 48, pp. 259-264, 2001.
- M. Noguchi, T. Numata, Y. Mitani, T. Shino, S. Kawanaka, Y. Oowaki, and A. Toriumi, "Back gate effects on threshold voltage sensitivity to SOI thickness in fully depleted SOI MOSFETs", *IEEE Electron Device Lett.*, vol. 22, pp. 32-34, 2001.
- J. T. Park, J. P. Colinge, and C. H. Diaz, "Pi-Gate SOI MOSFET", *IEEE Electron Device Lett.*, vol. 22, pp. 405-406, 2001.
- G. D. Wilk, R. M. Wallace and J. M. Anthony, "High-*k* gate dielectrics: Current status and materials properties considerations ", *J. Appl. Phys.*, vol. 89, pp. 5243-5275, 2001.
- A. O. Conde, F. J. G. Sanchez, J. J. Liou, A. Cerdeira, M. Estrada and Y. Yue, "A review of recent MOSFET threshold voltage extraction methods", *Microelectron. Reliab.*, vol. 42, pp. 583-596, 2002.
- H. S. P. Wong, "Beyond the conventional transistor ", *IBM J. Res. Develop.*, vol. 46, pp. 133-168, 2002.
- G. K. Celler and S. Cristoloveanu, "Frontiers of silicon-on-insulator", J. Appl. Phys., vol. 93, no. 9, pp. 4955-4978, 2003.
- L. Chang, Y. K. Choi, S. Ha, P. Ranade, S. Ziong, J. Bokor, C. Hu and T. J. King, "Extremely scaled silicon nano-CMOS devices", *Proc. IEEE*, vol. 91, pp. 1860-1873, 2003.

- M. Fischetti, "Scaling MOSFETs to the limit: A physicist', s perspective", *J. Comput. Electron.*, vol. 2, pp. 73-79, 2003.
- N. R. Mohapatra , M. P. Desai , S. G. Narendra and V. R. Rao, "Modeling of parasitic capacitances in deep submicrometer conventional and high-*k* dielectric MOS transistors", *IEEE Trans. Electron Devices*, vol. 50, pp. 959-966, 2003.
- K. Suzuki and S. Pidin, "Short-channel single-gate SOI MOSFET model ", *IEEE Trans. Electron Devices*, vol. 50, pp. 1297-1305, 2003.
- V. P. Trivedi and J. G. Fossum, "Scaling fully depleted SOI CMOS", *IEEE Trans. Electron Devices*, vol. 50, pp. 2095-2103, 2003.
- M. Yang, M. Ieong, L. Shi, K. Chan, V. Chan, A. Chou, E. Gusev, K. Jenkins, D. Boyd, Y. Ninomiya, D. Pendleton, Y. Surpris, D. Heenan, J. Ott, K. Guarini, C. D'Emic, M. Cobb, P. Mooney, B. To, N. Rovedo, J. Benedict, R. Mo and H. Ng, "High performance CMOS fabricated on hybrid substrate with different crystal orientations ", *IEDM Tech. Dig.*, pp. 453-456, 2003.
- H. W. Zan, T. C. Chang, P.-S. Shih, D.-Z. Peng, P.-Y. Kuo, T.-Y. Huang, C.-Y. Chang and P.-T. Liu, "A study of parasitic resistance effects in thin-channel polycrystalline silicon TFTs with tungsten-clad source/drain", *IEEE Electron Device Lett.*, vol. 24, pp. 509-511, 2003.
- C. G. Ahn, W. J. Cho, K. J. Im, J. H. Yang, I. B. Baek, S. K. Baek, and S. J. Lee, "Recessed source-drain(S/D) SOI MOSFETs with low S/D extension (SDE) external resistance," IEEE International SOI Conference, pp. 207-208, 2004.
- A. Chaudhry and M. J. Kumar, "Controlling short-channel effects in deep submicron SOI MOSFET',s for improved reliability: A review", *IEEE Trans. Device Mater. Rel.*, vol. 4, pp. 99-109, 2004.
- J.-P. Colinge, "Multiple gate SOI MOSFETs", Solid State Electron., vol. 48, pp. 897-905, 2004.
- J. P. Colinge, "Silicon-on-insulator technology: materials to VLSI," 3rd edition, *Kluwer* Academic Publishers. 2004.
- M. J. Kumar and A. Chaudhry, "Two-dimensional analytical modeling of fully depleted dualmaterial gate (DMG) SOI MOSFET and evidence for diminished short-channel effects", *IEEE Trans. Electron Devices*, vol. 51, pp. 569-574, 2004.
- D. Jimenez, J. J. Saenz, B. Iníguez, J. Sune, L. F. Marsal, and J. Pallares, "Modeling of nanoscale gate-all-around MOSFETs", *IEEE Electron Device Lett.*, vol. 25, pp. 314-316, 2004.
- T. Numata, M. Noguchi, Y. Oowaki and S. Takagi, "Voltage fluctuation in fully depleted SOI MOSFETs with back-gate control", *Solid-State Electron.*, vol. 48, pp. 979-984, 2004.
- Z. Zhang, S. Zhang, and M. Chan, "Self-Align Recessed Source Drain Ultrathin Body SOI MOSFET", *IEEE Devices Lett.*, vol. 25, pp. 740-742, Nov. 2004.

- S. Barraud, L. Clavelier and T. Ernst, "Electron transport in thin SOI, strained-SOI and GeOI MOSFET by Monte-Carlo simulation", *Solid State Electron.*, vol. 49, no. 7, pp. 1090-1097, 2005.
- P. Clarke, "EE Times: Intel Enters Billion-Transistor Processor Era," 14 November 2005.
- H. I. Hanafi, D. C. Boyd, K. K. Chan, W. Natzle, and L. Shi, "Fully-depleted SOI MOSFETs with low source and drain resistance and minimal overlap capacitance using a recessed channel damascene gate process," Jan 2005, U.S. Patent-6841831 B2.
- R. W. Keyes, "Physical limits of silicon transistors and circuits", *Rep. Progr. Phys.*, vol. 68, 2005.
- C. D. Sheraw, M. Yang, D. M. Fried, G. Costrini, T. Kanarsky, W.-H. Lee, V. Chan, M. V. Fischetti, J. Holt, L. Black, M. Naeem, S. Panda, L. Economikos, J. Groschopf, A. Kapur, Y. Li, R. T. Mo, A. Bonnoit, D. Degraw, S. Luning, D. Chidambarrao, X. Wang, A. Bryant, D. Brown, P. Agnello, M. Ieong, S.-F. Huang, X. Chen, and M. Khare, "Dual stress liner enhancement in hybrid orientation technology", *VLSI Symp. Tech. Dig.*, pp. 12, 2005.
- V. Trivedi and J. Fossum, "Nanoscale FD/SOI CMOS: Thick or thin BOX?", *IEEE Electron Device Lett.*, vol. 26, pp. 26-28, 2005.
- G. Tsutsui, M. Saitoh, T. Nagumo and T. Hiramoto, "Impact of SOI thickness fluctuation on threshold voltage variation in ultra-thin body SOI MOSFETs", *IEEE Trans. Nanotechnol.*, vol. 4, pp. 369-373, 2005.
- K. N. Tu, U. Gosele, "Hollow nanostructures based on the Kirkendall effect: Design and stability considerations, *Appl. Phys. Lett.*, vol. 86, pp. 093111(1-3), 2005.
- C. Ahn, W. Cho, K. Im, J. Yang, I. Baek, S. Lee and S. Baek, "Ultra-thin film SOI MOSFET having recessed source/drain structure and method of fabricating the same" June 2006, U.S. Patent U.S-20060131648 A1.
- K. Goel, M. Saxena, M. Gupta, and R. S. Gupta, "Modeling and simulation of a nanoscale three region tri-material gate stack (TRIMGAS) MOSFET for improved carrier transport efficiency and reduced hot-electron effects," *IEEE Trans. Electron Devices*, vol. 53, pp.1623-1633, 2006.
- M. Ieong, V. Narayanan, D. Singh, A. Topol, V. Chan, and Z. Ren, "Transistor scaling with novel materials", *Materials Today*, vol. 9, pp. 26-31, 2006.
- W. Ke, X. Han, D. Li, X. Wang, R. Han, T. Zhang, R. Han, and S. Zhang "Recessed source/drain for scaling SOI MOSFET to the limit," *International Conf. on Solid-State and Integrated Circuit Technology(ICSICT)*, pp. 84 86, 2006.
- J. M. Larson and J. P. Snyder, "Overview and status of metal S/D Schottky-barrier MOSFET technology", *IEEE Trans. Electron Devices*, vol. 53, pp. 1048-1058, 2006.
- C. Pacha, K. von Arnim, T. Schulz, W. Xiong, M. Gostkowski, G. Knoblinger, A. Marshall, T. Nirschl, J. Berthold, C. Russ, H. Gossner, C. Duvvury, P. Patruno, R. Cleavelin and K. Schruefer, "Circuit design issues in multi-gate FET CMOS technologies", *Proc. ISSCC*, pp. 420-421, 2006.

- J. C. Ranuarez, M. J. Deen and C.-H. Chen, "A review of gate tunneling current in MOS devices", *Microelectron. Reliab.*, vol. 46, 2006.
- G. Ribes , S. Bruyere , D. Roy , C. Parthasarthy , M. Muller , M. Denais , V. Huard , T. Skotnicki and G. Ghibaudo, "Origin of V_t instabilities in high-*k* dielectrics: Jahn-Teller effect or oxygen vacancies", *IEEE Trans. Device Mater. Rel.*, vol. 6, pp. 132-135, 2006.
- V. Y. Thean, B. J. Goolsby, B. Y. Nguyen, T. T. Nguyen, and T. A. Stephens, "Semiconductor fabrication process including recessed source/drain regions in an SOI wafer," August 2006, U. S. Patent US-7091071, 2006.
- S. E. Thompson and S. Parthasarthy, "Moore's law: The future of Si microelectronics", *Mater. Today*, vol. 9, pp. 20-25, 2006.
- M. Yang, V. W. C. Chan, K. K. Chan, L. Shi, D. M. Fried, J. H. Stathis, A. I. Chou, E. Gusev, J. A. Ott, L. E. Burns, M. V. Fischetti and M. Ieong, "Hybrid-orientation technology (HOT): Opportunities and challenges", *IEEE Trans. Electron Devices*, vol. 53, pp. 965-978, 2006.
- H. Zhao, N. Agrawal, R. Javier, S. C. Rustagi, M. Jurczak, Y. C. Yeo and G. S. Samudra, "Simulation of multiple gate FinFET device gate capacitance and performance with gate length and pitch scaling", *Proc. SISPAD 2006*, pp. 251-255.
- Y. Chen, L. M. Walthour, W. M., Moller, "Method for forming metal silicide regions in an integrated circuit," July 2007, US Patent US-7250356 B2.
- D. L. Critchlow, "Recollections on MOSFET Scaling," IEEE Solid-State Circuits Society Newsletter, vol. 12 pp. 19-22, 2007.

International Technology Roadmap for Semiconductors, 2007.

- H. X. Yu, Z. F. Long, H. R , and Z. Xing, "Corner effects in doublegate/ gate-allaround MOSFETs," Chin. Phys., vol. 16, pp. 812–816, 2007.
- W. Ke, X. Han, D. Li, X. Wang, R. Han, T. Zhang, R. Han and S. Zhang, "Recessed source/drain for sub-50nm UTB SOI MOSFET," *Semiconductor Sci. and Technol.*, vol. 22, pp.577-583, 2007.
- S. Takagi, Strained-Si CMOS Technology, Springer, 2007.
- V. Venkataraman, S. Nawal and M. J. Kumar, "Compact analytical threshold-voltage model of nanoscale fully depleted strained-Si on Silicon-Germanium-on-Insulator (SGOI) MOSFETs", *IEEE Trans. Electron Devices*, vol. 54, pp. 554-562, 2007
- J. P. Colinge, "The new generation of SOI MOSFETs", *Romanian J. Inf. Sci. Technol.*, vol. 11, pp. 3-15, 2008.
- J. P. Colinge, FinFETs and Other Multi-Gate Transistors, Springer, 2008.
- A. Dey , A. Chakravorty , N. DasGupta and A. DasGupta, "Analytical model of subthreshold current and slope for asymmetric 4-T and 3-T double-gate MOSFETs ", *IEEE Trans. Electron Devices*, vol. 55, pp. 3442-3449, 2008.

- M. J. Kumar and M. Silva, "The ground plane in buried oxide for controlling short-channel effects in nanoscale SOI MOSFETs ", *IEEE Trans. Electron Devices*, vol. 55, pp. 1554-1557, 2008.
- C. R. Manoj, M. Nagpal, D. Varghese and V. R. Rao, "Device design and optimization considerations for bulk FinFETs", *IEEE Trans. Electron Devices*, vol. 55, pp. 609-615, 2008.
- G. Zhang , Z. Shao and K. Zhou, "Threshold voltage model of short channel FD-SOI MOSFETs with vertical Gaussian profile", *IEEE Trans. Electron Devices*, vol. 55, pp. 803-809, 2008.
- S. Bangsaruntip, G. M. Cohen, A. Majumdar, Y. Zhang, S. U. Engelmann, N. C. M. Fuller, L. M. Gignac, S. Mittal, J. S. Newbury, M. Guillorn, T. Barwicz, L. Sekaric, M. M. Frank and J. W. Sleight, "High performance and highly uniform gate-all-around silicon nanowire MOSFETs with wire size dependent scaling", *IEDM Tech. Dig.*, pp. 297-300, 2009.
- M. Chu, Y. Sun, U. Aghoram, and S. E. Thompson," Strain: A Solution for Higher Carrier Mobility in Nanoscale MOSFETs" Annual Review of Mat. Res., vol. 39, pp. 203-229, 2009.
- H. G. Hanatian, M. Fathipour, H. Talebi, "Nanoscale Ultra-Thin Body -Silicon-On-Insulator Field Effect Transistor with Step BOX: Self heating and Short Channel Effects", *Ultimate Integration of Silicon*. pp. 325-328, 2009.
- H. Iwai, "Technology roadmap for 22nm and beyond", Electron Devices and Semiconductor Technology, 2009. *IEDST'09. 2nd International workshop on Electron Devices and Semiconductor Technology*, pp. 1-4.
- H. Li, C. Xu, N. Srivastava and K. Banerjee, "Carbon nanomaterials for next-generation interconnects and passives: Physics, status, and prospects", *IEEE Trans. Electron Devices*, vol. 56, pp. 1799-1821, 2009.
- S. Luan, H. Liu, R. Jia, and J. Wang, "Two-dimensional subthreshold current model for dualmaterial gate SOI nMOSFETs with single halo, Front. *Electr. Electron. Eng. China*, vol.4, pp.98-103, 2009.
- A. Majumdar , Z. Ren , S. J. Koester and W. Haensch, "Undoped-body, extremely thin SOI MOSFETs with back gates", *IEEE Trans. Electron Devices*, vol. 56, pp. 2270-2276, 2009.
- M. Masahara, Y. Liu, K. Endo, T. Matsukawa and E. Suzuki, "Vertical Ultrathin-channel Multi-gate MOSFETs (MuGFETs): Technological Challenges and Future Developments", *IEEJ Transactions on Electrical and Electronic Engineering*, vol. 4, pp. 386-391, 2009.
- B. Svilicic, V. Jovanovic and T. Suligoj, "Analytical models of front- and back-gate potential distribution and threshold voltage for recessed source/drain UTB SOI MOSFETs ", *Solid State Electron*, vol. 53, pp. 540-547, 2009.

- A. Chaudhry, J. N. Roy, J. N and G. Joshi, "Nanoscale strained-Si MOSFET physics and modeling approaches: a review, Nanoscale strained-Si MOSFET physics and modeling approaches: a review," *Journal of semiconductors*, vol, 31, pp. 104001 (1-5), 2010.
- L. Jin, L. Hongxia, L. Bin, C. Lei., and Y. Bo, "Two-dimensional threshold voltage analytical model of DMG strained-silicon-on-insulator MOSFETs," *Journal of Semiconductors*, vol. 31, pp. 084008, 2010.
- P.-C. Huang, L.-A. Chen, and J.-T. Sheu, "Electric-Field enhancement of a gate-all-around nanowire thin-film transistor memory," IEEE Electron. Dev. Lett., vol. 31, pp. 216– 218, 2010.
- Y. B. Kim, "Review paper: Challenges for nanoscale MOSFETs and emerging nanoelectronics", *Trans. Electr. Electron. Mater.*, vol. 11, pp. 93-105, 2010.
- A. Kranti, S. B. Rashmi, J. P. Raskin and G. A. Armstrong, "Analog/RF performance of sub-100 nm SOI MOSFETs with non-classical gate-source/drain underlap channel design", *Proc. Tropical Meeting Silicon Monolithic Integr. Circuits RF Syst. (SiRF)*, pp. 45-48, 2010.
- B. Svilicic, V. Jovanovic, T. Suligoj, "Analysis of subthreshold conduction in short channel recessed source/drain UTB SOI MOSFETs," *Solid-State Electronics*, vol. 54, pp. 545-551, 2010.
- H. Takeda , K. Takeuchi and Y. Hayashi, "Scalability study of ultra-thin-body SOI MOSFET using full-band and quantum mechanical based device simulations", *Proc. Symp. VLSI Technol.*, pp. 63-64, 2010.
- K. Chena, G. Zhanga, and X. Zhenga, "A Subthreshold Swing Model for FD SOI MOSFET with Vertical Gaussian Profile," *ISDRS, College Park, MD, USA*, pp. 7-9, 2011.
- A. K. Kushwaha, On the modeling of dual material double gate fully depleted silicon on insulator MOSFETs, *Phd thesis, NIT Kuruksetra*, India, 2011.
- I. Ferain, C. A. Colinge and J.-P. Colinge, "Multigate transistors as the future of classical metal-oxide-semiconductor field-effect transistors", *Nature*, vol. 479, pp. 310-316, 2011
- Lu, D. D., Dunga, M. V., Lin, C.-H., Niknejad, A. M., and Hu, C., A computationally efficient compact model for fully-depleted SOI MOSFETs with independent controlled front and back-gates, *Solid State Electron.*, vol. 62, pp. 31-39, 2011
- ATLAS user's manual: Device simulation software, Santa Clara, CA: Silvaco International (2012)
- R. Cavin, P. Lugli, and V. Zhirnov, "Science and engineering beyond Moore's law", Proc. *IEEE, Special Centennial Issue*, vol. 100, pp. 1720-1749, 2012.
- T. K. Chiang, "A new quasi-2-D threshold voltage modelfor short-channel junctionless cylindrical surrounding gate (JLCSG) MOSFETs", *IEEE Trans.Electron Devices*, vol. 59, pp. 3127-3129, 2012.
- A. Khakifirooz, K. Cheng, A. Reznicek, T. Adam, N. Loubet, H. He, J. Kuss, J. Li, P. Kulkarni, S. Ponoth, R. Sreenivasan, Q. Liu, B. Doris and G. Shahidi, "Scalability

of extremely thin SOI (ETSOI) MOSFETs to sub-20-nm gate length", IEEE Electron Device Lett., vol. 33, pp. 149-151, 2012

- K. J. Kuhn, "Considerations for ultimate CMOS scaling", *IEEE Trans. Electron Devices*, vol. 59, pp. 1813-1828, 2012.
- S. Markov, B. Cheng and, A. Asenov, "Statistical variability in fully depleted SOI MOSFETs due to random dopant fluctuations in the source and drain extensions", *IEEE Electron Device Lett.*, vol. 3, pp. 315-317, 2012.
- S. L. Tripathi, R. Mishra, and R. A. Mishra, "dielectric materials", *J. Electron Devices*, vol. 16, pp. 1388-1394, 2012.
- S. Das and S. Kundu, "Simulation to study the effect of oxide thickness and high-k dielectric on drain induced barrier lowering in n-type MOSFET," IEEE Trans. on Nanotechnology, vol. 12, pp. 945-947, 2013
- M. Kumar, S. Dubey, S., P. K. Tiwari, and S. Jit, "Analytical modeling and simulation of subthreshold characteristics of back-gated SSGOI and SSOI MOSFETs: A comparative study," *Current Applied Physics*, vol. 13, pp. 1778-1786, 2013.
- M. Kumar, S. Dubey, P. K. Tiwari and S. Jit, "An analytical model of threshold voltage for short-channel double-material-gate (DMG) strained-Si (s-Si) on silicon-germanium-on-insulator (SGOI) MOSFETs", J. Comput. Electron., vol. 12, pp. 20-28, 2013.
- M. R. Narayanan, H. A. Nashash, D. Pal, and M. Chandra, "Thermal model of MOSFET with SELBOX structure," *J Comput Electron*, vol. 12, pp. 803-811, 2013.
- G. C. Patil, and S. Qureshi, "Asymmetric drain underlap dopant-segregated Schottky barrier ultrathin-body SOI MOSFET for low-power mixed-signal circuits", *Semicond. Sci. Technol.*, vol. 28, pp. 045002-1-045002-9, 2013.
- A. Kumar, and P. K. Tiwari, "A threshold voltage model of short-channel fully-depleted recessed-source/drain (Re-S/D) UTB SOI MOSFETs including substrate induced surface potential effects," *Solid-State Electron*. vol. 95, pp. 52-60, 2014.
- P. Arobinda, and S Angsuman, "Analytical study of Dual Material Surrounding Gate MOSFET to suppress short-channel effects (SCEs)," *Eng. Sci. and Technol., an Int.*, vol. 17, pp. 205-212, 2014.
- R. Basak, B. Maiti, and A. Mallik, "Analytical model of gate leakage current through bilayer oxide stack in advanced MOSFET", *Superlattices and Microstruct.*, vol. 80, pp. 20– 31, 2015
- J. H. Park, G. S. Jang, H. Y. Kim, S. K. Lee and S. K. Joo, "High-performance poly-Si thinfilm transistor with high- k ZrTiO 4 gate dielectric", IEEE Electron Device Lett., vol. 36, no. 9, pp. 920-922, 2015
- K. R. A. Sasaki, M. Aoulaiche, E. Simoen, C. Claeys and J. A. Martino, "Extension less UTBB FDSOI Devices in Enhanced Dynamic Threshold Mode under Low Power Point of View,' J. of Low Power Electron. Appl. Vol. 5, pp 69-80 2015.

- H. Wu, M. Si, L. Dong, J. Gu, J. Zhang and P. D. Ye, "Germanium nMOSFETs with recessed channel and S/D: Contact, scalability, interface, and drain current exceeding 1 A/mm", IEEE Trans. Electron Devices, vol. 62, no. 5, pp. 1419-1426, 2015
- C. Y. Chena, J. T. Lina and M. H. Chiang, Fabrication, "Variability in Multiple Gate MOSFETs: A Bulk FinFET Study," ECS J. Solid State Sci. Technol. vol. 5, pp. 3096-3100, 2016.

List of Publications

List of Papers in Journals

- [1]. G.K. Saramekala, A. Santra, S. Dubey, S. Jit, and P.K. Tiwari, "An analytical threshold voltage model for a short-channel dual-metal-gate (DMG) fully depleted recessed-source/drain (Re-S/D) SOI MOSFET", *Superlattices and Microstructures* vol. 60, pp. 580-595, 2013.
- [2]. G.K. Saramekala, A. Santra, M. Kumar, S. Dubey, S. Jit, and P.K. Tiwari, "Analytical subthreshold current and subthreshold swing models of short-channel dual-metal-gate (DMG) fully depleted recessed- source/drain (Re-S/D) SOI MOSFETs", *Journal of Computational Electronics*, vol. 13 pp. 467–476, 2014.
- [3]. G.K. Saramekala, S. Dubey, and P.K. Tiwari, "Analog and radio-frequency (RF) performance evaluation of fully-depleted (FD) recessed-source/drain (Re-S/D) SOI MOSFETs", *Superlattices and Microstructures*, vol. 76, pp.77–89, 2014.
- [4]. G.K. Saramekala and P.K. Tiwari, "An Analytical threshold voltage model for recessed source drain SOI MOSFETs with high-*k* dielectric", *Chinese Physics B*, vol. 24, pp. 108505, 2015.
- [5]. G.K. Saramekala and P.K. Tiwari, "A ballistic subthreshold current model for ultrashort channel recessed-source/ drain (Re-S/D) SOI MOSFETs", *Journal of Nanoengineering and Nanomanufacturing*, vol. 5, pp. 1–5, 2015.
- [6]. G.K. Saramekala and P.K. Tiwari, "An Analytical threshold voltage model for a fullydepleted (FD) recessed-source/drain (Re-S/D) SOI MOSFET with back-gate control", *Journal of Electronic Materials*, vol. 45 pp. 1-8, 2016.
- [7]. G.K. Saramekala and P.K. Tiwari, "Analytical Subthreshold Current and Subthreshold Swing Models for a Fully Depleted (FD) Recessed-Source/Drain (Re-S/D) SOI MOSFET with Back-Gate Control", *Journal of Electronic Materials*, vol. 46 pp. 1-11, 2017.

List of Papers in Conferences

- [1]. G.K. Saramekala, S. Jit and P.K. Tiwari, "An analytical subthreshold surface potential model for a short-channel dual-metal-gate (DMG) recessed-source/ drain (Re-S/D) SOI MOSFET", ICATET jaipur, 2013.
- [2]. G.K. Saramekala, S. Jit and P.K. Tiwari, "AtlasTM based simulation study of the electrical characteristics of dual-metal-gate (DMG) fully depleted recessed-source/drain (Re-S/D) SOI MOSFETs", in: Proc. IEEE Int. Advances in Electrical Engineering, Vellore, India, 2014, pp. 1–4
- [3]. G.K. Saramekala and P.K. Tiwari, "Analytical subthreshold current modeling of Re-S/D SOI MOSFETs with high K dielectric", IEEE Asia Modeling Symposium 2014, Taipei, Taiwan, ISBN 978-1-4799-6487-1, pp.258-264
- [4]. G.K. Saramekala, M. Kushwaha and P.K. Tiwari, "Threshold Voltage Modeling of Recessed- Source/drain SOI Mosfets with Vertical Gaussian Doping Profile" International Conference on Materials for Advance Material Research Society of Singapore, 2015